

Explanatory note

This explanatory note relates to this Bill as introduced into Parliament.

Overview of Bill

The objects of this Bill are:

- (a) to provide for the protection, conservation and ecologically sustainable development of the water sources of the State, and
- (b) to repeal certain Acts, to amend certain other Acts and to enact savings and transitional provisions consequent on the enactment of the proposed Act, and
- (c) to make other provisions of a minor, consequential or ancillary nature.

Outline of provisions

Chapter 1 Preliminary

Chapter 1 contains provisions dealing with the name of the proposed Act, its commencement, its objects and its interpretation.

Chapter 2 Water resource planning

Part 1 Water sharing principles

Part 1 contains provisions establishing water sharing principles and environmental water principles, and provides for the classification of water sources.

Part 2 Management committees

Part 2 contains provisions with respect to the constitution of water management areas, the establishment of management committees for those areas, the membership of those committees and the functions that those committees may exercise.

Part 3 Management plans

Division 1 contains provisions with respect to how a management plan comes to be prepared and what, in general terms, a management plan is to deal with.

Division 2 contains provisions specific to management plans that deal with water sharing.

Division 3 contains provisions specific to management plans that deal with water source protection.

Division 4 contains provisions specific to management plans that deal with drainage management.

Division 5 contains provisions specific to management plans that deal with floodplain management.

Division 6 contains provisions with respect to the procedures to be followed in making a management plan, including provisions for public exhibition, notification of major stakeholders and consultation with the Minister for the Environment.

Division 7 contains a provision that enables the Minister to vary the bulk access regime of a management plan when of the opinion that it is necessary to do so in the public interest.

Division 8 contains provisions with respect to the validity and effect of a management plan.

Part 4 Minister's plans

Part 4 contains a single provision that empowers the Minister to make plans for areas that are not within a water management area, for water management areas that are not yet the subject of a management plan and for matters not dealt with by management plans. Generally speaking, a Minister's plan is to have the same effect as a management plan.

Part 5 Implementation programs

Part 5 contains a single provision that empowers the Minister to develop programs to implement management plans and Minister's plans. Implementation programs relating to management plans will be developed in conjunction with the management committee by which the management plan was prepared.

Chapter 3 Water resource management

Part 1 Basic landholder rights

Division 1 contains a single provision conferring rights on landholders to take and use water for domestic and stock purposes without the need for any access licence or approval.

Division 2 contains provisions conferring rights on landholders within a harvestable rights area to collect and store rainwater run-off.

Part 2 Access licences

Division 1 defines the rights conferred by an access licence, and provides for the various categories of access licences that may be granted and the relative priorities between them. The Division also provides for the making of "available water determinations" which, together with an access licence, determine when and how the holder of an access licence may take water under the licence.

Division 2 contains provisions with respect to the procedures to be followed for the grant and renewal of access licences.

Division 3 contains provisions with respect to the conditions that may be imposed on access licences (including mandatory conditions arising from management plans and Minister's plans) and the period for which access licences are to be in force (20 years for public utilities, 15 years generally and 2 years for access licences for "opportunistic water").

Division 4 contains provisions with respect to the transfer of access licences and the water allocations under them in accordance with specified transfer rules to be established by management plans and Minister's plans.

Division 5 contains provisions enabling water allocations to be recredited to access licences in accordance with specified water return flow rules to be established by the Minister.

Division 6 contains provisions with respect to the surrender, suspension, cancellation and compulsory acquisition of access licences. Compensation for access licences compulsorily acquired will be determined by the Valuer-General whose decisions will be appealable to the Land and Environment Court.

Division 7 contains provisions enabling the Minister to declare embargoes on applications for access licences.

Division 8 contains provisions with respect to records and registration.

Division 9 contains a provision that provides for the payment of compensation to the holders of access licences who are adversely affected as a consequence of a variation to a bulk access regime that is made by the Minister under proposed section 38.

Division 10 contains a provision with respect to the making of regulations.

Part 3 Approvals

Division 1 contains provisions describing the various kinds of approval that may be granted (water use approvals, water management work approvals and activity approvals) and empowers the Minister to grant temporary exemptions from the requirements for an approval.

Division 2 contains provisions with respect to the procedures to be followed for the grant of an approval. In some cases, applications will have to be advertised and third party objections may be made. The approval process is linked into the integrated development procedures of the *Environmental Planning and Assessment Act 1979*.

Division 3 contains provisions with respect to the conditions that may be imposed on approvals (including mandatory conditions arising from management plans and Minister's plans) and the period for which approvals are to be in force (20 years for water management work approvals, 10 years for water use approvals and aquifer interference approvals and 3 years for controlled activity approvals).

Division 4 contains provisions with respect to the amendment, surrender, suspension and cancellation of approvals.

Division 5 contains provisions enabling the Minister to declare embargoes on applications for approvals.

Division 6 contains provisions with respect to the keeping of registers in relation to approvals.

Part 4 Finance

Part 4 contains provisions with respect to the recovery of administrative costs, the establishment of guidelines for charges and other matters relating to costs and charges.

Chapter 4 Joint private works

Part 1 Irrigation corporations

Division 1 applies Part 1 to existing irrigation corporations and defines certain expressions used in the Part.

Division 2 contains provisions with respect to the functions of irrigation corporations.

Division 3 contains provisions with respect to the operating licences under which irrigation corporations conduct their affairs.

Division 4 contains administrative provisions with respect to irrigation corporations.

The provisions of this Part replicate such of the provisions of the *Irrigation Corporations Act 1994* as are still relevant to irrigation corporations.

Part 2 Private irrigation districts

Division 1 applies Part 2 to both new and existing private irrigation districts and defines certain expressions used in the Part.

Division 2 contains provisions with respect to the constitution of lands as a private irrigation district on the petition of landholders within the proposed district.

Division 3 contains provisions with respect to the establishment of a private irrigation board for a private irrigation district so constituted.

Division 4 empowers a private irrigation board to construct new water supply works and take over existing water supply works.

Division 5 contains provisions with respect to the payment of compensation by a private irrigation board.

Division 6 contains provisions with respect to the fixing, assessment and levying of rates and charges by a private irrigation board.

Division 7 contains provisions with respect to the supply of water by a private irrigation board to landholders within its district.

Division 8 contains provisions with respect to the effects of subdividing land within a private irrigation district in relation to water allocations.

Division 9 contains provisions with respect to meetings of landholders within a private irrigation district.

Division 10 contains provisions with respect to certain financial matters.

Division 11 contains administrative provisions with respect to private irrigation boards.

The provisions of this Part replicate provisions of the *Private Irrigation Districts Act* 1973.

Part 3 Private drainage boards

Division 1 applies Part 3 to existing drainage boards and defines certain expressions used in the Part.

Division 2 contains provisions with respect to the constitution and functions of private drainage boards.

Division 3 contains provisions with respect to the fixing, assessment and levying of rates by a private drainage board.

Division 4 contains provisions with respect to the valuation for rating purposes of land within a private drainage board's district.

Division 5 contains provisions with respect to the effects of subdividing land within a drainage district in relation to the use of drainage works.

Division 6 contains administrative provisions with respect to private drainage boards.

The provisions of this Part replicate such of the provisions of the *Drainage Act* 1939 as are still relevant to private drainage boards.

Part 4 Private water trusts

Division 1 applies Part 4 to existing water supply trusts and defines certain expressions used in the Part.

Division 2 contains provisions with respect to the constitution and functions of private water trusts.

Division 3 contains provisions with respect to the functions exercisable by private water trusts in connection with the supply of water to land within the trust district.

Division 4 contains provisions with respect to the fixing, assessment and levying of rates by a private water trust.

Division 5 contains provisions with respect to the effects of subdividing land within a water supply district in relation to water allocations.

Division 6 contains administrative provisions with respect to private water trusts.

The provisions of this Part replicate such of the provisions of Part 3 of the *Water Act 1912* as are still relevant to private water trusts.

Chapter 5 Public works

Part 1 Fish River water supply works

Division 1 defines certain expressions used in Part 1, vests the control of the Fish River water supply works in the Minister and empowers the Minister to repair those works.

Division 2 contains provisions with respect to the supply of water from the Fish River water supply works.

Division 3 contains provisions with respect to the establishment of special areas within which public and private development are to be controlled under the proposed Act.

Division 4 contains provisions with respect to the accounting procedures to be observed in relation to the Fish River water supply works.

Division 5 contains provisions empowering regulations to be made with respect to specified matters.

The provisions of this Part replicate such of the provisions of the *Fish River Water Supply Administration Act 1945* as are still relevant.

Part 2 Hunter Valley flood mitigation works

Division 1 applies Part 2 to the Hunter Valley, and defines certain expressions used in the Part.

Division 2 confers certain powers on the Minister with respect to the construction of works and the restriction of certain activities that might adversely affect flood mitigation works.

Division 3 contains provisions with respect to the contributions to be made by local councils, public authorities and the Hunter Catchment Management Trust towards the costs of constructing and maintaining flood mitigation works within the Hunter Valley.

Division 4 contains administrative provisions with respect to the Hunter Valley flood mitigation works.

The provisions of this Part replicate such of the provisions of the *Hunter Valley Flood Mitigation Act 1956* as are still relevant.

Part 3 Lowbidgee flood control and irrigation works

Part 3 empowers the Minister to supply water by way of flood irrigation to landholders within the Lowbidgee flood control and irrigation district, and to levy rates in connection with that supply.

The provisions of this Part replicate such of the provisions of Part 7 of the *Water Act 1912* as are still relevant.

Chapter 6 Public utilities

Part 1 Major utilities

Division 1 contains provisions with respect to the declaration of certain public utilities as major utilities and certain other administrative provisions.

Division 2 contains provisions with respect to the granting of water authorisations to major utilities. A water authorisation is a substitute for a water management work approval under Part 3 of Chapter 3.

Division 3 contains provisions enabling the Minister to give, and enforce, directions for the carrying out of remedial work.

Division 4 contains provisions with respect to rights of appeal to the Land and Environment Court.

The provisions of this Part replicate such of the provisions of Part 9 of the *Water Act 1912* as are still relevant.

Part 2 Water supply authorities

Division 1 defines certain expressions used in Part 2.

Division 2 contains provisions with respect to the constitution of water supply authorities.

Division 3 contains provisions with respect to the functions of water supply authorities.

Division 4 contains provisions with respect to the establishment of special areas within which public and private development are to be controlled under the proposed Act.

Division 5 contains provisions with respect to compliance certificates issued by water supply authorities (for use in connection with the subdivision and development of land) and for developer contributions to a water supply authority's costs of augmenting water, sewerage and drainage works.

Division 6 contains provisions with respect to the fixing, assessment and levying of service charges and other charges.

Division 7 contains administrative provisions with respect to water supply authorities.

The provisions of this Part replicate such of the provisions of the *Water Supply Authorities Act 1987* as are still relevant.

Chapter 7 Enforcement

Part 1 Directions to landholders and other persons

Part 1 enables the Minister to give directions to landholders and other persons for the purpose of enforcing compliance with the requirements of the proposed Act. The Part also allows the Minister to do whatever is necessary to give effect to a direction that is not complied with, and to recover the costs of so doing from the person to whom the direction was given. The Part also empowers the Minister to seek injunctions from the Land and Environment Court to restrain a breach of the proposed Act.

Part 2 Powers of entry

Part 2 gives certain powers of entry onto private land, and regulates the manner in which such powers are to be exercised.

Part 3 Offences

Division 1 contains provisions creating a number of major offences (that is, offences that carry maximum penalties of \$275,000 if committed by corporations and \$132,000 if committed by individuals).

Division 2 contains provisions creating a number of minor offences (that is, offences that carry lesser maximum penalties).

Part 4 Recovery of rates and charges

Part 4 contains provisions with respect to the recovery of rates and charges, whether by private bodies under Chapter 4 or public authorities under Chapters 5 and 6. Unpaid rates and charges are to be a charge on land, enforceable (in the case of rates and charges owed to the Minister or a water supply authority) by compulsory sale of land. Interest on unpaid rates and charges will be payable at the current rate for unpaid judgments of the Supreme Court. There will be provision for the issuing of certificates as to the amount of unpaid rates or charges in relation to specified land, such a certificate to be conclusive proof of its contents in favour of the person to whom it is issued or to a purchaser of the land.

Part 5 Legal proceedings and appeals

Division 1 contains a number of miscellaneous provisions with respect to legal proceedings. In particular, it allows proceedings for offences to be taken before the Land and Environment Court or a Local Court, and limits a Local Court in relation to the maximum penalties that it may impose for an offence. The Division also provides for the issuing of penalty notices.

Division 2 provides a right of appeal to the Land and Environment Court against certain decisions under the proposed Act.

Chapter 8 Administration

Part 1 Water Advisory Council

Part 1 contains provisions with respect to the establishment and functions of the Water Advisory Council.

Part 2 Water Administration Ministerial Corporation

Part 2 contains provisions with respect to the constitution and functions of the Water Administration Ministerial Corporation. In particular, the Ministerial Corporation is to be empowered to compulsorily acquire land and is to be a Constructing Authority for the purposes of the *Public Works Act 1912*.

Part 3 General administration

Part 3 contains general provisions with respect to administration, including the appointment of committees, the delegation of functions the appointment of authorised officers, the resolution of disputes between public authorities and the establishment and maintenance of registers.

Chapter 9 Miscellaneous

Chapter 9 contains provisions:

- (a) asserting the State's rights to the control, use and flow of water, and
- (b) abolishing certain common law riparian rights, and
- (c) providing for the service of documents, and
- (d) binding the Crown to the provisions of the proposed Act, and
- (e) excluding personal liability for persons involved in the administration of the proposed Act, and
- (f) excluding Crown liability for failures in water supply, and
- (g) providing for annual reports, and
- (h) providing for the making of regulations, and
- (i) providing for the repeal of certain Acts, the amendment of other Acts, and the enactment of savings and transitional provisions, and
- (j) requiring the proposed Act to be reviewed after 5 years.
- **Schedule 1** lists the irrigation corporations to which the proposed Act applies.
- **Schedule 2** lists the major utilities to which the proposed Act applies.
- **Schedule 3** lists the water supply authorities to which the proposed Act applies.
- **Schedule 4** lists the lands that are exempt from service charges imposed by water supply authorities.
- **Schedule 5** deals with the constitution and procedure of water supply authorities.
- **Schedule 6** deals with the constitution and procedure of the Water Advisory Council and management committees.

Schedule 7 lists the Acts to be repealed by the proposed Act.

Schedule 8 contains amendments to other Acts as a consequence of the enactment of the proposed Act.

Schedule 9 contains savings and transitional provisions consequent on the enactment of the proposed Act.

The Dictionary defines certain words and expressions used in the proposed Act.



		Page
Chapte	r 1 Preliminary	
	1 Name of Act2 Commencement3 Objects4 Interpretation	2 2 2 2 3
Chapte	r 2 Water management planning	
Part 1	Water sharing principles	
	5 Water sharing principles6 Environmental water principles7 Classification of water sources	4 4 5

				Page
Part 2	Mana	ageme	ent committees	
	9	Establi: Membe	tution of water management areas shment of management committees ership of committee ons of management committees	6 6 7
Part 3	Mana	ageme	ent plans	
	Divisi	ion 1	Preliminary	
	13 14	Manag Provisi	ation of draft management plan ement plan to be consistent with other instruments ons applicable to all management plans economic impact	9 9 10 10
	Divisi	ion 2	Water sharing	
	17	Core p	ation of Division rovisions nal provisions	11 11 11
	Divisi	ion 3	Water source protection	
			ation of Division source protection zones	12 12
	Divisi	ion 4	Drainage management	
	22 23	Draina Core p Manag	ation of Division ge management principles rovisions lement plan to be consistent with drainage lement principles	13 13 14 14
	Divisi	ion 5	Floodplain management	
		Floodp Core p Manag	ation of Division lain management principles rovisions ement plan to be consistent with floodplain ement principles	14 14 15
	Divisi	ion 6	Procedures for making management plans	
	29 30 31 32 33	Notifica Refere Public	t of management plan ation of certain persons and bodies nce of draft management plan to Minister exhibition of draft management plan ssions on draft management plan	16 16 17 17 18

				Page	
	35	Making Amend	nission of draft management plan to Minister of management plan ment and repeal of management plans n of management plans	18 18 19 19	
	Divis	ion 7	Amendment of management plans by Minister		
	38		ment of management plans in relation to bulk regimes	19	
	Divis	ion 8	Miscellaneous		
		Effect o	of management plans of management plans on exercise of Minister's	20	
	41	function Conside	eration of management plans by public authorities	20 20	
Part 4	Mini	ster's	olans		
	42	Ministe	r's plans	21	
Part 5	Impl	lementa	ation programs		
	43	Implem	entation programs	22	
Chapter	· 3 \	Nater	management implementation		
Part 1	Basic landholder rights				
	Divis	ion 1	Domestic and stock rights		
	44	Domes	tic and stock rights	23	
	Divis	ion 2	Harvestable rights		
	45 46	Harvest Harvest	table rights table rights orders	24 24	
Part 2	Acc	ess lice	ences		
	Divis	ion 1	Preliminary		
	47 48 49	Catego	licences ries of licence s between different categories of licence	26 26 27	

			Page
50	Available	e water determinations	27
Divis	ion 2	Granting and renewal of access licences	
53 54	Objection Determine Notice of	and renewal of access licences ns to granting of access licences nation of applications f decision ed allocation of access licences	28 28 29 30 30
Divis	ion 3	Conditions and duration of access licences	
57 58	Imposition Revocat	ns of access licence on of conditions after access licence is granted ion of conditions of access licences	30 31 31 31
Divis	ion 4	Access licence transfers	
61 62	Transfer Water al	principles of access licences locations may be transferred e transfers	32 32 33 33
Divis	ion 5	Water return flows	
		eturn flow rules locations may be credited	33 33
Divis	ion 6	Surrender, suspension, cancellation and compulsory acquisition of access licences	
66 67 68	Suspens	er of access licences sion and cancellation of access licences sory acquisition of access licences	34 34 34
Divis	ion 7	Embargoes on applications for access licences	
	Permane	ary embargo ent embargo n of embargo	35 36 36
Divis	ion 8	Registers and registration	
72 73 74 75	Register Water al	of access licences of available water determinations location accounts n access licence to be created by instrument in	36 37 37
76 77	writing Registra	tion of certain interests on of rights of holder of access licence	37 37 38

-				Page
	Division 9		Compensation when management plan bulk access regime varied	
	78		nsation payable for reductions in water allocations from Minister's amendment of management plan	38
	Divis	ion 10	Miscellaneous	
	79	Regula	tions	39
Part 3	Арр	rovals		
	Divis	ion 1	Preliminary	
	80		use approvals	40
	81 92		management work approvals	40 40
	83	•	approvals tion from requirement for approval	41
	Divis	ion 2	Applications for approvals	
		Objection Determine	tions for approvals ons to applications for water use approvals ination of applications affected by public inquiries he Environmental Planning and Assessment Act	41 42 43
		Determ Matters Ground Notice	ination of applications affecting consideration of applications s of refusal of certain applications of decision ures in relation to integrated development	43 44 44 45 45
	Divis	ion 3	Conditions and duration of approvals	
	93 94	Condition Imposit Revoca Duration Land be	ons of approval generally ons of approval for joint water supply schemes ion of conditions after approval is granted approval of conditions of approval enefited by water use approval or water ement work approval	46 46 46 47 47
	Divis	ion 4	Amendment, surrender, suspension and cancellation of approvals	
	98	Amend	ment of water management work approval on	
		applicat	tion by holder of approval	48
	99 100		der of approvals asion and cancellation of approvals	49 49

				Page
	Divisio	n 5	Embargoes on applications for approvals	
	102 F	ermar	rary embargo nent embargo on of embargo	50 50 50
	Divisio	n 6	Registers	
	104 F	Registe	er of approvals	51
Part 4	Finan	се		
	106 N 107 C	/liniste Charge	nination of recoverable costs r's guidelines for charges es payable for access licences and approvals eate as to charges outstanding	52 52 53 53
Chapte	r4 Jo	int p	orivate works	
Part 1	Irrigat	ion c	orporations	
	Divisio	n 1	Preliminary	
	110 E	efiniti	ition of Part ons ements for access licences and approvals	54 54 54
	Divisio	n 2	Irrigation corporations	
	113 E	ntry o	of irrigation corporations n to land g to be an irrigation corporation	55 55 56
	Divisio	n 3	Operating licences	
	116 T 117 T 118 C 119 C 120 Ir	erms erm o Contrav Cancel	ty conferred by operating licence and conditions of operating licence f operating licence vention of operating licence lation of operating licence on corporation may make arrangements with aries	57 57 58 58 59
	Divisio	n 4	Miscellaneous	
	121 S	Succes	sor in title liable for unpaid contract charges	60

				Page
	122 123	Indemr Registe	nities er of licences, audits and management programs	60 61
Part 2	Priv	ate irri	gation districts	
	Divis	ion 1	Preliminary	
	124 125 126	Definiti	ation of Part ons ements for access licences and approvals	62 62 63
	Divis	ion 2	Private irrigation districts	
	130	Additio Alterati Excisio	ution of private irrigation districts n of lands to private irrigation districts on of private irrigation district n of lands from private irrigation districts mentary petitions and objections	63 64 65 66 67 68
	Divis	ion 3	Private irrigation boards	
		Election Remov	irrigation boards n of members of private irrigation boards ral of members of private irrigation board from office	69 70
	136 137 138	Abolitic	pointment of administrator on of private irrigation districts g-up of private irrigation boards //ees	70 71 71 72
	Divis	ion 4	Construction and taking over of works	
	139 140		sed sites uction, maintenance and operation of water supply	72 73
	141	Private	irrigation board may apply to take over water	
	142 143	Govern	works ons to applications or may authorise construction or taking over of supply work	73 74 74
	144 145 146	Taking Transfe	over works er of lands of entry	74 75 75
	Divis	ion 5	Compensation	
	147		npensation for surveys on land within private on district	75
	148	Compe	ensation for surveys on lands outside private on district	76

			Page
	takes over Determin	sation where private irrigation board constructs or er works ation of amount of compensation of compensation	76 76 77
Divis	ion 6	Rates and charges for water	
154 155 156 157 158 159	Special rands Assessman Assessman Provision Liability for Charge of	ment of rates	77 78 78 78 79 79 80 80
Divis	ion 7	Supply of water	
161 162 163 164	Supply of Discontin	nation of allocation f water nuance or reduction of supply of water ances in which private irrigation board not obliged	81 81 82
165	to supply Landhold		82
166	works. Sale of s	urplus water	82 83
Divis	ion 8	Effect of new subdivisions	
167 168	holdings Supply o	f water for domestic and stock purposes to new resulting from subdivisions f water for irrigation purposes to a new holding from a subdivision	83 84
169	Additiona	al works required as a result of subdivision	85
Divis	ion 9	Meetings of landholders	
170 171	General Voting rig		86 87
Divis	ion 10	Finance	
	Banking	to be rendered	87 87 87 88
Divis	ion 11	Miscellaneous	
176	Delegation	on	88

				Page
			ment book admissible as evidence of notice of proceedings on a private irrigation	89
	179	board Recover	ry of rates ment of proclamations	89 89 89
	181	By-laws		89
Part 3	Priva	ate drai	nage boards	
	Divisi	ion 1	Preliminary	
		Definitio	ion of Part ons ments for approvals	91 91 91
	Divisi	ion 2	Private drainage boards	
	186 187 188 189	Appoints Function Entry or Extension	drainage boards ment and election of directors ns of private drainage boards n lands on of drainage works ment of boundaries	91 92 92 93 93
	Divisi	ion 3	Rating	
	192 193 194 195 196 197 198	Annual of Rates, r Irregular Rate bo Due dat Liability Liability	ok	94 94 95 96 96 96 96
	Divisi	ion 4	Valuations	
	201 202 203 204 205 206 207 208	Inclusive Separat Land ex Valuatio Alteratio Notice of Objection Appeals	ns in valuation book of valuation	97 98 98 98 99 99 100 101

				Page
	Divisi	on 5	Effect of new subdivisions	
		Supply subdivi	of water to new holdings resulting from sions	102
			nal works required as a result of subdivision	102
	Divisi	on 6	Miscellaneous	
	213 214	Dissolu Debts Accour Regula	nts	103 103 104 104
Part 4	Priva	ite wa	ter trusts	
	Divisi	on 1	Preliminary	
	217	Definition	ntion of Part ons ements for access licences and approvals	105 105 105
	Divisi	on 2	Private water trusts	
	220 221	Alteration Remov	tment and election of members on of water supply districts al of land from water supply district of trusts	105 106 106 107
	Divisi	on 3	Functions of members	
	224 225 226	Powers Supply Power	of members s and duties of members of water of members to enter and inspect npensation	107 107 108 110 110
	Divisi	on 4	Rating	
	229 230			111 112 113 113
	Divisi	on 5	Effect of new subdivisions	
		subdivi	of water to new holdings resulting from sions	113 114

			Page
	Division 6	Miscellaneous	
	disso	ster may remove members and Governor may olve trust ulations	115 115
Chaptei	5 Publ	ic works	
Part 1	Fish Rive	er water supply works	
	Division 1	Preliminary	
		nitions River water supply works controlled by the Minister ster may repair works	117 117 117
	Division 2	Supply of water	
	240 Cour	ster may supply water ncil not to supply outside area without consent ompensation for failure of supply	118 118 118
	Division 3	Special areas	
		cial areas n land in special areas cise of certain functions in special areas	118 119 119
	Division 4	Finance	
	246 Alloc	ounts to be kept in Special Deposits Account eation of money nents by Treasurer	119 120 120
	Division 5	Miscellaneous	
	248 Regu	ulations	120
Part 2	Hunter V	alley flood mitigation works	
	Division 1	Preliminary	
		ication of Part nitions	122 122

				Page
	Divisi	ion 2	Powers of Minister	
	251 252 253 254	Restricti	construction powers on on use of lands adjacent to levee banks ction of fences, structures and flood works of trees	123 123 124 125
	Divisi	ion 3	Finance	
	256 257	Allocation Cost of Cos	s to be kept in Special Deposits Account on of money works generally works to protect towns and villages protective works contribution to works program by Hunter ent Management Trust ation with Hunter Catchment Management Trust orks program ation with local council as to works program at of contribution towards maintenance by a local eation with local council as to maintenance program maintenance works ats to be duly made	125 125 125 126 126 127 127 128 129 129 130
	Divisi	ion 4	Miscellaneous	
	267 268 269 270 271	approve Major flo councils Mainten Applicat	o protect public and local government works to be d by Minister podgates to be maintained and operated by local ance of minor floodgates ion of Public Works Act 1912 e of cost of works	130 131 131 132 132
Part 3	Low	bidgee	flood control and irrigation works	
		Fixing of Rates	ns of water f rates to be taken into account	133 133 133 134 134 135 135

			Page
Chapte	er 6 Pub	lic utilities	
Part 1	Major ut	ilities	
	Division 1	Preliminary	
	279 Maj 280 Red	or utilities juirements for access licences and approvals	136 136
	Division 2	Water authorisations	
	282 Cor 283 Imp	ter authorisation Iditions of water authorisation osition of conditions after water authorisation is	136 137
	284 Řev 285 Dur 286 Effe 287 Rev	nted rocation of conditions rocation of water authorisation ct of water authorisation riew of water authorisation riew of water authorisation retrieved the management charges	137 138 138 138 138 139
	Division 3	Directions to major utilities	
	290 Rer 291 Lan 292 Cor 293 Offe	ections for remedial work nedial measures may be taken by Minister d and Environment Court may grant injunctions npensation for loss of water ences with respect to water authorisations I penalties	139 140 140 140 140 141
	Division 4	Appeals	
	295 App	eals to Land and Environment Court	141
Part 2	Water s	upply authorities	
	Division 1	Preliminary	
		nitions juirements for access licences and approvals	143 144
	Division 2	Water supply authorities	
	299 Cor	ter supply authorities stitution of water supply authorities cutory body may be water supply authority	144 144 145

		Page
301	Provisions relating to constitution and procedure of water	4.45
302	supply authorities Area of operations	145 146
303	Employees	147
304	Delegation	147
Divis	ion 3 Functions of water supply authorities	
	Functions of water supply authority	147
	Exercise of functions by a water supply authority	147
	Commercial operations Assistance to statutory body	148 148
	Entry on land to read meters or carry out works	148
310	Entry on land for inspections	149
311	Power to break up roads	151
	Altering position of conduits	152 152
	Obstruction of works Finding source of pollution of water supply	153
		100
DIVIS	ion 4 Special areas	
	Special areas	153
	Crown land in special areas Exercise of certain functions in special areas	153 154
	·	134
DIVIS	ion 5 Developer contributions to the construction of works	
318	Application for certificate of compliance	154
319	Authority may impose certain requirements before	
000	granting certificate of compliance	154
	Granting of certificates of compliance	155
Divis	ion 6 Finance	
	Orders for purpose of service charges	156
	Connections	156
323	Authority may levy service charges and impose fees and other charges	158
324	Land in respect of which a water supply authority may levy	
<u></u>	service charges	158
325	Certain land exempt from service charges	159
326	Classification of lands	159
327 328	Basis of levying service charges Determinations by water supply authority	160 160
329	Assessment of service charges	162
330	Re-assessment of service charges	163
Divis	ion 7 Miscellaneous	
331	Acquisition and divesting of land	163

		Page
	 Compensation for interference or damage Efficiency review Appointment of administrator Regulations 	164 164 164 165
Chapter	r 7 Enforcement	
Part 1	Directions to landholders and other persons	
	Directions concerning the production of information Directions concerning waste of water Directions to protect water sources Directions to stop work where unlawful activity occurring Directions to holders of basic landholder rights Temporary stop work order Directions concerning unusable water management wo Directions concerning damage caused by straying stock Measures that may be specified in direction Prior notice of direction not required Remedial measures may be taken by Minister Land and Environment Court may grant injunctions Restraint of breaches of this Act	169 169 orks 169
Part 2	Powers of entry	
	 349 Powers of entry generally 350 Search warrants 351 Powers exercisable following lawful entry to land 352 Duty of care 	173 173 174 174
Part 3	Offences	
	Division 1 Major offences	
	 Unlawful taking of water Using water without a water use approval Constructing or using water management work without water management work approval Unlawful carrying out of certain activities Contravention of certain directions Destruction, damage and interference with certain work Taking water from public or private works Maximum penalty for offences under this Division 	175 176 176

			Page
	Divis	sion 2 Other offences	
	361	Exposure of underground pipes	178
		Work done by unqualified person	178
		Obstruction	178
	364	False or misleading information	179
Part 4	Rec	overy of rates and charges	
	365	Definition	180
	366	Certain rates and charges to be a charge on land	180
	367		180
	368	, ,	180
	369		180
	370		181
	371		181
		Liability where an estate or interest is transferred	182
	373	Liability where a person becomes entitled to an estate or	400
		interest	182
Part 5	Lega	al proceedings and appeals	
	Divis	sion 1 Legal proceedings	
	374	Offences by corporations	183
	375		183
	376	•	184
	377	•	
		other action under this Act	185
	378	Evidentiary certificates	185
	Divis	sion 2 Appeals	
	379	Appeals to Land and Environment Court	186
	_		
Chapter	8 /	Administration	
Part 1	Wat	er Advisory Council	
		·	4.5.5
	380	Establishment of Water Advisory Council	189
	381	Functions of Water Advisory Council	190

			Page
Part 2	Wat	er Administration Ministerial Corporation	
	382	Constitution of Water Administration Ministerial	
		Corporation	191
	383	Functions of Ministerial Corporation	191
		Commercial operations	191
		Application of Public Works Act 1912	192
		Acquisition of land	192
		Staff of Ministerial Corporation Delegation of functions	192 192
Part 3	Gen	eral administration	
	389	Committees	193
		Delegation	193
		Authorised officers	193
		Resolution of disputes between public authorities	193
Chapte	r 9 I	Miscellaneous	
•			
		State's water rights	194
		Abolition of common law riparian rights	194
	395		195
		Unpaid fees and charges	195
		Act binds Crown	195 195
		Exclusion of personal liability Exclusion of Crown liability	195
		Annual report of Department	197
	401		197
		Repeals	198
	403	·	198
	404		198
	405	Review of Act	198
Schedule	es		
	1	Irrigation corporations	199
	2	Major utilities	200
	3 4	Water supply authorities Land exempt from water supply authority service charges	201 202

		Page
	 Constitution and procedure of water supply authorities Water Advisory Council and management committees Repeals Amendment of other Acts Savings, transitional and other provisions 	205 218 224 225 241
Dictionary		254



No , 2000

A Bill for

An Act to provide for the protection, conservation and ecologically sustainable development of the water sources of the State, and for other purposes.

The Legislature of New South Wales enacts:

Cha	pte	r 1	Preliminary	2
1	Nan	ne of A	Act	3
		This	Act is the Water Management Act 2000.	4
2	Con	nmend	cement	5
	(1)		Act commences on a day or days to be appointed by amation.	6 7
	(2)	provi	rent days may be appointed for the commencement of a single sion of Schedule 7 or 8 for the purpose of commencing the ls or amendments effected by the provision on different days.	8 9 10
3	Obj	ects		11
	(1)	mana	objects of this Act are to provide for the sustainable and integrated gement of the water sources of the State for the benefit of both nt and future generations and, in particular:	12 13 14
		(a)	to apply the principles of ecologically sustainable development, and	15 16
		(b)	to protect, enhance and restore water sources, their associated ecosystems, ecological processes and biological diversity and their water quality, and	17 18 19
		(c)	to recognise and foster the significant social and economic benefits to the State that result from the sustainable and efficient use of water, including: (i) benefits to the environment, and (ii) benefits to urban communities, agriculture, fisheries, industry and recreation, and (iii) benefits to culture and heritage, and (iv) benefits to the Aboriginal people in relation to their spiritual, social and customary use of land,	20 21 22 23 24 25 26 27 28
		(d)	to recognise role of the community, as a partner with government, in resolving issues relating to the management of water sources,	29 30 31

		(e)	to provide for the orderly, efficient and equitable sharing of water from water sources,	1 2
		(f)	to integrate the management of water sources with the management of other aspects of the environment, including the land, its soil, its native vegetation and its native fauna,	3 4 5
		(g)	to encourage the sharing of responsibility for the sustainable and efficient use of water between the Government and water users,	6 7 8
		(h)	to encourage best practice in the management and use of water.	9
4	Inte	rpreta	ition	10
	(1)		ds and expressions that are defined in the Dictionary at the end of Act have the meanings set out in that Dictionary.	11 12
	(2)	Note	s in the text of this Act do not form part of this Act.	13

Chapter 2 Water management planning

Part	1 \	Vate	er sharing principles	2		
5	Water sharing principles					
	(1)		following water sharing principles are recognised for the purposes is Act:	4 5		
		(a)	firstly, that management of a water source should seek to protect its dependent ecosystems,	6 7		
		(b)	secondly, that management of a water source should ensure that landholders are able to exercise their basic landholder rights,	8		
		(c)	thirdly, that the extraction of water pursuant to any other authority to extract water from a water source must not be permitted to prejudice the principles set out in paragraphs (a) and (b).	10 11 12 13		
	(2)	unde	the intention of Parliament that all persons exercising functions r this Act should take all reasonable steps to do so in accordance and so as to promote, the principles set out in subsection (1).	14 15 16		
6	Env	ironm	nental water principles	17		
	(1)		following classes of environmental water are recognised for the oses of this Act:	18 19		
		(a)	environmental health water (that is, water that must be provided for fundamental ecosystem health at all times),	20 21		
		(b)	supplementary environmental water (that is, water that must be provided for specific environmental purposes at specific times or in specific circumstances but may otherwise be used for other purposes),	22 23 24 25		
		(c)	adaptive environmental water (that is, water that is subject to an access licence but is committed for use for environmental purposes).	26 27 28		

1

	(2)	establ	he intention of Parliament that principles for the identification, lishment and maintenance of each class of environmental water	1 2
			ronmental water principles) be established for all of the water	3
			es in the State, either by means of management plans or a	4
			ster's plan, as soon as practicable after the commencement of this	5
		Act.		6
7	Clas	ssificat	tion of water sources	7
	(1)	The N	Minister may, by order published in the Gazette, classify water	8
	` ′		es for the purposes of this Act.	9
	(2)	Water	r sources are to be classified as follows:	10
		(a)	as to the extent to which they are at risk (that is, the extent to	11
			which harm to the water source or its dependent ecosystems is	12
			likely to occur),	13
		(b)	as to the extent to which they are subject to stress (that is, the	14
			extent to which harm to the water source or its dependent	15
			ecosystems has occurred or is occurring),	16
		(c)	as to the extent to which they are high conservation value water	17
			sources (that is, the extent to which they have intrinsic value	18
			that merits a higher degree of protection from risk and stress).	19
	(3)	It is th	he intention of Parliament that:	20
		(a)	the water sources of the State be classified in accordance with	21
			this section as soon as practicable after the commencement of	22
			this Act, and	23
		(b)	in establishing management plans and a Minister's plan under	24
			this Act, plans are to be prepared first for high risk, high stress	25
			and high conservation value water sources.	26

Part 2 Management committees

8	Cor	stitution of water management areas	2
	(1)	The Minister may, by order published in the Gazette, constitute any land as a water management area.	3 4
	(2)	An order under this section must name the area and fix its boundaries.	5
9	Esta	ablishment of management committees	6
	(1)	The Minister may, by order published in the Gazette:	7
		(a) establish a management committee to carry out a specific task in relation to water management in a water management area, and	8 9 10
		(b) set terms of reference in accordance with which the committee is to carry out that task.	11 12
	(2)	The task for which a management committee is appointed may relate to any aspect of water management, including (without limitation) water sharing, water source protection, floodplain management and drainage management.	13 14 15 16
	(3)	At any time the Minister may, by order published in the Gazette, abolish a management committee, and may do so whether or not it has completed the task for which it was established.	17 18 19
	(4)	Nothing in this section prevents the establishment of two or more committees for the same water management area so long as they do not have overlapping functions.	20 21 22
10	Mer	nbership of committee	23
	(1)	A management committee consists of at least 10, but not more than 20, members appointed by the Minister, of whom:	24 25
		(a) at least two are to be persons appointed to represent the interests of environmental protection groups, and	26 27
		(b) at least two are to be persons appointed to represent the interests of water user groups, and	28 29
		(c) at least two are to be persons appointed to represent the interests of local councils, and	30 31

		(d)	at least one is to be a person appointed to represent the interests of catchment management boards and trusts, and	1
		(e)	at least one is to be an Aboriginal person appointed to represent the interests of Aboriginal persons, and	3
		(f)	at least one is to be a member of staff of the Department, and	5
		(g)	such other persons as are appointed to represent such interests as the Minister considers require representation, and	6
		(h)	one is to be a person (not being a member of staff of the Department and not being associated with any of the interests referred to above) who is appointed as an independent chairperson for the committee.	8 9 10 11
	(2)		egulations may make provision with respect to qualifications for nument as a member of a management committee.	12 13
	(3)		lule 6 has effect with respect to the constitution and procedure of agement committee.	14 15
11	Fun	ctions	of management committees	16
	(1)		rincipal function of a management committee is to carry out the or which it is appointed.	17 18
	(2)		ask for which a committee is appointed may include any one or of the following:	19 20
		(a)	to prepare a draft management plan for the water management area,	21 22
		(b)	to review a management plan that is in force for the water management area,	23 24
		(c)	to investigate such matters affecting the management of the water management area as the Minister refers to it for investigation,	25 26 27
		(d)	to report to the Minister on such matters affecting the management of the water management area as the Minister refers to it for report,	28 29 30
		(e)	to advise the Minister on such matters affecting the management of the water management area as the Minister refers to it for advice.	31 32 33

(3) It is the duty of a management committee to exercise its functions consistently with the principles of ecologically sustainable development.

1 2

Water man Manageme	nagement planning Chapter 2 ent plans Part 3										
Preliminary	Division 1	-									
Part 3	Part 3 Management plans										
Division	1 Preliminary										
12 Pr	eparation of draft management plan										
(1)	The Minister may, by the order by which a management committee is established or by a subsequent order in writing:										

13

(a) direct the committee to prepare a draft management plan, and 6 review any related implementation program, on any aspect of 7 water management, including (but not limited to): 8 water sharing, and 9 (i) (ii) water source protection, and 10 drainage management, and (iii) 11 floodplain management, and (iv) 12 (b) set terms of reference in accordance with which such a plan is 13 to be prepared. 14 (2) A management committee to which such an order is given is to prepare 15 a draft management plan in accordance with the terms of reference 16 specified in the order. 17 (3) If the management committee fails to prepare a draft management plan 18 in accordance with its terms of reference, the Minister may do so 19 instead. 20 Management plan to be consistent with other instruments 21 (1) A draft management plan must not be inconsistent with: 22 government policy, or (a) 23 (b) government obligations arising under any inter-governmental 24 agreement to which the government is a party, including the 25 Murray-Darling Basin Agreement referred to in the 26 Murray-Darling Basin Act 1992, or 27 government obligations arising in connection with any (c) 28 international agreement to which the government of the 29 Commonwealth is a party, or 30 any State environmental planning policy under the (d) 31 Environmental Planning and Assessment Act 1979, or 32

Clause 12

1

2

Clause 13 Chapter 2 Part 3 Division 1		V\ M	Water Management Bill 2000 Water management planning Management plans Preliminary	
		(e)	any catchment management strategy under the <i>Catchment Management Act 1989</i> , or	1
		(f)	any protection of the environment policy under the <i>Protection</i> of the Environment Operations Act 1997, or	3
		(g)	any regulation under the Sydney Water Catchment Management Act 1998, the Sydney Water Act 1994, the Hunter Water Act 1991 or the Googong Dam Catchment Area Act 1975.	5
	(2)		ne purposes of subsection (1), <i>government policy</i> includes such ers as are declared by the regulations to be government policy.	9
14	Pro	vision	s applicable to all management plans	10
			anagement plan for a water management area may contain the wing kinds of provisions:	11 12
		(a)	provisions with respect to the preservation and enhancement of the quality of water in the water sources in the area,	13 14
		(b)	provisions with respect to the kinds of monitoring and reporting requirements that should be imposed as conditions of approvals having effect within the area,	15 16 17
		(c)	provisions with respect to the conditions to which access licences and approvals having effect within the area are to be subject (<i>mandatory conditions</i>),	18 19 20
		(d)	provisions indicating the circumstances in which, the matters in respect of which and the extent to which the management plan may be amended by the Minister during the period for which it is in force,	21 22 23 24
		(e)	provisions with respect to such other matters as may be authorised by the regulations.	25 26
15	Soc	io-eco	onomic impact	27
		must	rmulating a draft management plan, the management committee have due regard to the socio-economic impacts of the proposals dered for inclusion in the draft plan.	28 29

Water Management Bill 2000	Clause 16
Water management planning	Chapter 2
Management plans	Part 3
Water sharing	Division 2

Division 2		2	Water sharing	1
16	App	olicatio	on of Division	2
			Division applies to a management plan to the extent to which it s with water sharing.	3
17	Cor	e prov	visions	5
	(1)		anagement plan for a water management area must have the wing provisions:	6 7
		(a)	provisions that establish environmental water principles for the area,	8
		(b)	provisions that identify requirements for water within the area to satisfy basic landholder rights,	10 11
		(c)	provisions that identify requirements for water for extraction under access licences,	12 13
		(d)	provisions that identify the existing stress, the potential risk and the intrinsic conservation value of the various water sources in the area,	14 15 16
		(e)	provisions that establish a bulk access regime for the extraction of water under access licences, having regard to the provisions referred to in paragraphs (a)–(d).	17 18 19
	(2)	The 1	bulk access regime referred to in subsection (1) (e):	20
		(a)	must recognise the effect of climatic variability on the availability of water, and	21 22
		(b)	may establish rules with respect to the priorities according to which access licences are to be adjusted as a consequence of any reduction in the availability of water.	23 24 25
	(3)		rules referred to in subsection (2) must comply with the priorities blished under Division 1 of Part 2 of Chapter 3.	26 27
18	Add	ditiona	al provisions	28
	(1)		anagement plan for a water management area may also contain the wing kinds of provisions:	29 30
		(a)	provisions with respect to:	31

Clause 18 Chapter 2 Part 3 Division 2		Water Management Bill 2000 Water management planning Management plans Water sharing				
			 (i) the rate at which water may be taken from any water source in the area, or (ii) the quantity of water that may be taken from any water source in the area, 			
		(b)	provisions with respect to the kinds of water supply works that may be constructed and used in the area,	:		
		(c)	provisions with respect to: (i) the rate at which water may be applied to land in any specified zone in the area, or (ii) the quantity of water that may be applied to land in any specified zone in the area,	1 1		
		(d)	provisions establishing transfer rules for the area.	1:		
	(2)		ransfer rules referred to in subsection (1) (d) must comply with linister's transfer principles.	1		
Divis	sion (3	Water source protection	1		
19	Арр	Application of Division				
			Division applies to a management plan to the extent to which it with water source protection.	1		
20	Wat	er sou	rce protection zones	19		
	(1)		unagement plan for a water management area may contain the wing provisions:	20		
		(a)	provisions identifying zones (<i>water source protection zones</i>) in which controlled activities must be regulated in order to discourage unacceptable effects on the water sources of the area,	2 2 2 2		
		(b)	provisions limiting the exercise of basic landholder rights within a water source protection zone.	2 2		
	(2)	subse plan i	nagement plan that contains provisions of the kind referred to in ction (1) may, but need not, contain a provision declaring that the s to have the same effect as a regional environmental plan under nvironmental Planning and Assessment Act 1979.	2 ¹ 2 ¹ 3 ¹ 3		

Water	sourc	e prote	ection Division 3	
	(3)	Subj	ect to the regulations:	1
		(a)	a management plan that contains a provision of the kind referred to in subsection (2) (a <i>deemed regional environmental plan</i>) is taken to be a regional environmental plan, and prevails over any other environmental planning instrument (other than a State environmental planning policy) to the extent of any inconsistency, and	2 3 4 5 6 7
		(b)	the Minister is taken to be the consent authority under the <i>Environmental Planning and Assessment Act 1979</i> in respect of all development to which the plan relates.	8 9 10
Division 4 Drainage management		4	Drainage management	11
21	App	olicatio	on of Division	12
			Division applies to a management plan to the extent to which it s with drainage management.	13 14
22	Dra	inage	management principles	15
			following principles are drainage management principles for the oses of this Act:	16 17
		(a)	the principle that wetting regimes in wetlands and related ecosystems should be maintained,	18 19
		(b)	the principle that habitats, animals (including fish) and plants should be preserved,	20 21
		(c)	the principle that groundwater dependent ecosystems should be maintained,	22 23
		(d)	the principle that geographical and other features of special significance to Aboriginal culture should be protected,	24 25
		(e)	the principle that drainage should not be permitted to have an adverse impact on water users,	26 27
		(f)	the principle that drainage should not be permitted to cause or exacerbate soil erosion, salinity or acidity,	28 29
		(g)	the principle that drainage should not be permitted to cause or exacerbate wastage of otherwise useable water.	30 31

Water management planning

Management plans

Clause 20

Chapter 2

Part 3

Part 3 Management plans Division 4 Drainage management			-	
23	Core provisions		isions	
	(1)	A ma	anagement plan for a water management area must have the wing provisions:	
		(a)	provisions that identify existing drainage works within the area and the way in which they are managed,	
		(b)	provisions that identify the impact of the existing drainage works on the environment, including their impact on: (i) natural surface flows, and (ii) groundwater, and (iii) soil erosion, salinity and acidity,	1
		(c)	provisions that, having regard to existing drainage works in the area, propose measures to be adopted to promote the drainage management principles.	1 1 1
	(2)	constr	measures proposed by the plan may include measures for the ruction of new drainage works and the modification or removal isting drainage works.	1 1 1
24		nagem ciples	ent plan to be consistent with drainage management	1 1
			nagement plan for drainage must be not be inconsistent with the age management principles.	1 2
Divis	sion (5	Floodplain management	2
25	App	olicatio	n of Division	2
			Division applies to a management plan to the extent to which it with floodplain management.	2 2
26	Flo	odplair	n management principles	2
			ollowing principles are floodplain management principles for the uses of this Act:	2
		(a)	the principle that the existing and future risk to human life and property arising from occupation of floodplains should be minimised,	2 2 3
		(b)	the principle that the environment should be protected, in particular in relation to: (i) the passage flow and distribution of floodwater and	3 3

Chapter 2

Water Management Bill 2000

Water management planning

Water Manag Flood				
			(ii) existing dominant floodways and exits from floodways, and	1 2
			 (iii) rates of flow, floodwater levels and duration of inundation, and (iv) downstream water flows, 	3 4 5
		(c)	the principle that natural flood regimes in wetlands and related ecosystems should be maintained,	6 7
		(d)	the principle that habitats, animals (including fish) and plants that benefit from periodic flooding should be preserved,	8
		(e)	the principle that geographical and other features of special significance to Aboriginal culture should be protected.	10 11
27	Cor	e pro	visions	12
	(1)		nanagement plan for a floodplain must have the following isions:	13 14
		(a)	provisions that identify the natural flooding regime in the floodplain, in terms of the frequency, duration, nature and extent of flooding and the risk to life and property from the effects of flooding,	15 16 17 18
		(b)	provisions that identify the ecological benefits of flooding in the floodplain, with particular regard to wetlands and other floodplain ecosystems,	19 20 21
		(c)	provisions that identify existing flood works in the floodplain, their benefits in terms of the protection they give to life and property and their environmental impact.	22 23 24
	(2)	the c exist be at	sures proposed by a management plan may include measures for construction of new flood works, the modification or removal of ing flood works and the control of activities that may affect, or to ffected by, the frequency, duration, nature or extent of flooding in the floodplain.	25 26 27 28 29
28		nagen nciples	nent plan to be consistent with floodplain management	30 31
		A ma	anagement plan for a floodplain must be not be inconsistent with:	32
		(a)	the floodplain management principles, or	33
		(b)	the general principles and policies set out in the <i>Floodplain Development Manual</i> .	34 35

Clause 29	Water Management Bill 2000
Chapter 2	Water management planning
Part 3	Management plans
Division 6	Procedures for making management plans

Division 6 Procedures for making management plans

Fo	rmat o	f management plan	2	
(1)	A m	anagement plan must include the following components:	3	
	(a)	a vision statement,	4	
	(b)	objectives consistent with the vision statement,	5	
	(c)	strategies for reaching those objectives,	ϵ	
	(d)	performance indicators to measure the success of those strategies.	7	
(2)	mana	ne case of a water management area for which an equivalent agement plan is already in force, the draft management plan may a the form of:	9 10 11	
	(a)	an amendment to the existing management plan, or	12	
	(b)	a new plan to replace the existing management plan.	13	
(3)	9	ect to subsections (1) and (2), the format of a management plan is as determined by the Minister.	14 15	
No	tification	on of certain persons and bodies	16	
(1)		In preparing a draft management plan, the following information must be notified to each person or body referred to in subsection (2):		
	(a)	the general aims and objectives of the draft plan,	19	
	(b)	a description of the water management area to which the draft plan is intended to apply,	20 21	
	(c)	such other matters as the Minister thinks fit.	22	
(2)	The	persons and bodies to be notified are as follows:	23	
	(a)	each local council within whose area the water management area is located,	24 25	
	(b)	each catchment management committee and catchment management trust within whose area of operations the water management area is located,	26 27 28	
	(c)	each holder of an access licence or approval who may be affected by the plan,	29 30	
	(d)	such other persons or bodies as the Minister may determine in relation to the plan	31 32	

	Management plans Procedures for making management plans				
	(3)		are to notify a person or body referred to in subsection (2) does not et the validity of a management plan.	1 2	
	written submissions to the Minister in draft management plan within 28 da		rson to whom information is notified under this section may make ten submissions to the Minister in relation to the preparation of the management plan within 28 days (or such longer period as may etermined by the Minister) after the information is notified.	3 4 5 6	
31	Ref	erenc	e of draft management plan to Minister	7	
	(1)	mana	r a draft management plan has been prepared, including a draft agement plan that has been referred back to it under this section, nanagement committee must submit the plan to the Minister.	8 9 10	
	(2)	with	e Minister is of the opinion that the draft plan does not comply the requirements of this Part, the Minister is to refer the draft plan to the management committee for further consideration.	11 12 13	
	(3)		section does not apply to a draft management plan prepared by Minister.	14 15	
32	Public exhibition of draft management plan				
	(1)	Once the Minister is satisfied that a draft management plan is suitable for public exhibition, the Minister:			
		(a)	must give public notice of the draft plan, and	19	
		(b)	must exhibit the draft plan (together with such other information as is appropriate or necessary to enable the draft plan and its implications to be understood) at the places, on the dates and during the times set out in the notice.	20 21 22 23	
	(2)	The	public notice referred to in subsection (1) (a):	24	
		(a)	must specify the places at which, the dates on which, and the times during which, the draft plan may be inspected by the public, and	25 26 27	
		(b)	must specify a period of at least 40 days during which submissions may be made to the Minister in relation to the plan (the <i>submission period</i>), and	28 29 30	
		(c)	must be published in a daily newspaper circulating throughout New South Wales and in a local newspaper.	31 32	

Water management planning

Clause 30

Chapter 2

Division 6		Prod	cedures for making management plans	
33	Sub	mission	s on draft management plan	1
	(1)		the submission period, any person may make written sions to the Minister on the draft management plan.	2
	(2)		inister must send a copy of each such submission to the ement committee by which the plan was prepared.	4
	(3)	Subsect the Mir	tion (2) does not apply to a draft management plan prepared by nister.	6
34	Res	ubmissi	on of draft management plan to Minister	8
	(1)	submiss the dra	on as practicable after completing its consideration of any sions received by it, the management committee must resubmit aft management plan to the Minister together with the tree's comments on the submissions.	9 10 11 12
	(2)		making any alterations to the draft management plan, the er must consult with the management committee.	13 14
	(3)	This see	ction does not apply to a draft management plan prepared by nister.	1: 10
35	Mal	ing of n	nanagement plan	17
	(1)	After co	omplying with the requirements of this Part, the Minister:	18
			may make a management plan in accordance with the draft plan, as finally submitted to the Minister, or	19 20
		1	may make a management plan in accordance with the draft plan, as finally submitted to the Minister, but with such alterations as the Minister thinks fit, or	21 22 23
		5	may cause the draft management plan to be re-exhibited (with such alterations as the Minister thinks fit) and resubmitted in accordance with this Part, or	24 25 26
		(d) 1	may decide not to proceed with the draft management plan.	27
	(2)		making a management plan, the Minister must obtain the rence of the Minister for the Environment to the making of the	28 29 30
	(3)		agement plan commences on the date on which it is published Gazette or on such later date as may be specified in the plan.	31 32

Chapter 2

Part 3

Water Management Bill 2000

Water management planning

Management plans

•	Procedures for making management plans Division 6		
	_		_
36	Am	endment and repeal of management plans	1
	(1)	A management plan may be amended by a subsequent management plan made in accordance with this Part.	nt 2 3
	(2)	A management plan may also be amended by the Minister, by notic published in the Gazette, but only in such circumstances, in relation t such matters and to such extent as the plan so provides.	
	(3)	The amendment of a management plan under subsection (2) take effect on the date of publication of the relevant notice or a later dat specified in the notice.	
37	Dur	ration of management plans	10
	(1)	Subject to this section, a management plan has effect for 5 years from the date on which it is made.	m 11
	(2)	Within the fifth year after it was made, the Minister is to review each management plan for the purpose of ascertaining whether it provisions remain adequate and appropriate.	
	(3)	Such a review is to be conducted in consultation with the Minister for the Environment.	or 16
	(4)	If following a review under this section the Minister is satisfied that it provisions remain adequate and appropriate, the Minister may, b notice published in the Gazette, extend the term of the plan by up to years.	y 19
	(5)	The power conferred on the Minister by subsection (4) may be exercised only once in relation to any particular management plan.	e 22 23
Divis	sion 7	7 Amendment of management plans by Minister	24
38	Am	endment of management plans in relation to bulk access regime	s 25

The Minister may, by order published in the Gazette, vary the bulk

access regime established by a management plan if satisfied that it is

in the public interest to do so.

Water Management Bill 2000

Water management planning

26

27

28

Clause 36 Chapter 2

Division	n 8	Miscellaneous	
Divisi	on 8	Miscellaneous	
39	Validity	y of management plans	
	an Ei	The validity of a management plan may not be called into question to legal proceedings other than those commenced in the Landonvironment Court within 3 months after the date of its publication of Gazette.	d and
40	Effect	of management plans on exercise of Minister's functions	
	re pl	When exercising functions under this Act, the Minister must tale as onable steps to give effect to the provisions of any manage lan and, in particular, to ensure that any environmental rinciples established by the plan are observed.	ement

41 Consideration of management plans by public authorities

Water Management Bill 2000

Water management planning

Management plans

- (1) When exercising its functions, a public authority must have regard to the provisions of any management plan to the extent to which they apply to the public authority.
- (2) For the purposes of this section, a management plan may be expressed so as to apply:
 - (a) to a specified public authority, to a specified class of public authorities or to public authorities generally, and
 - (b) to a specified function, to a specified class of functions or to functions generally.
- (3) This section neither restricts a public authority's statutory discretions nor authorises a public authority to do anything inconsistent with its statutory or other legal obligations.

Clause 39

Chapter 2

Part 3

1

Part 4 Minister's plans

2 Min	Minister's plans			
(1)	The Minister may, by order published in the Gazette, make a plan (a <i>Minister's plan</i>):			
	(a) for any part of the State that is not within a water management area, or	5 6		
	(b) for any water management area, or part of a water management area, for which a management plan is not in force, or	7 8		
	(c) for any water management area, or part of a water management area, for which a management plan is in force, but only so as to deal with matters not dealt with by the management plan.	9 10 11		
(2)	A Minister's plan must deal with any matters that a management plan is required to deal with, and may also deal with any other matters that a management plan is authorised to deal with, other than matters that are already dealt with by a management plan.	12 13 14 15		
(3)	Before making a plan that establishes environmental water principles, the Minister must obtain the concurrence of the Minister for the Environment to the establishment of those principles.	16 17 18		
(4)	A Minister's plan may not be made for the purpose of establishing water source protection zones.	19 20		
(5)	Except to the extent to which this Act otherwise provides, a Minister's plan has the same effect as a management plan.	21 22		
(6)	The Minister must cause each Minister's plan to be periodically reviewed at intervals of not more than 5 years.	23 24		

Part 5 Implementation programs

lmp	Implementation programs			
(1)	The Minister may, by order in writing, establish a program for implementing a management plan or Minister's plan (an <i>implementation program</i>).	3 4 5		
(2)	Before establishing the first implementation program for a management plan, the Minister must consult with the management committee by which the plan was prepared.	6 7 8		
(3)	An implementation program must set out the means by which the Minister intends that the objectives of the relevant management plan or Minister's plan are to be achieved.	9 10 11		
(4)	On establishing an implementation program, the Minister must ensure that:	12 13		
	(a) notice of that fact is published (in a local newspaper and in such other manner as the Minister considers appropriate) to persons in the area to which the program relates, and	14 15 16		
	(b) copies of the program (as currently in force) are made available for inspection during normal office hours, free of charge, at suitable locations within the area.	17 18 19		
(5)	The Minister is to ensure that the implementation program is reviewed each year for the purpose of determining whether the implementation program is effective in implementing the management plan or Minister's plan to which it relates.	20 21 22 23		
(6)	The results of each review under this section are to be included in the annual report for the Department under the <i>Annual Reports</i> (<i>Departments</i>) <i>Act</i> 1985.	24 25 26		

Domestic and stock rights	Division 1
Basic landholder rights	Part 1
Water management implementation	Chapter 3
Water Management Bill 2000	Clause 44

Chapter 3		r 3	Water management implementation	
Part	: 1 E	Basio	c landholder rights	2
Division 1 Domestic and stock rights				
44	Dor	nestic	and stock rights	4
	(1)		ndholder is entitled, without the need for an access licence, water ly work approval or water use approval:	5 6
		(a)	to take water from any river, estuary or lake to which the land has frontage or from any aquifer underlying the land, and	7 8
		(b)	to construct and use a water supply work for that purpose, and	9
		(c)	to use the water so taken for domestic consumption and stock watering, but not for any other purpose.	10 11
	(2)		ection (1) does not authorise a landholder to construct a dam or r bore without a water supply work approval.	12 13
	(3)	dome	maximum amount of water that can be used by a landholder for estic consumption or stock watering in any one year is the amount cribed by the regulations.	14 15 16
	(4)	This	section is subject to any deemed regional environmental plan.	17
	(5)	In thi	is section:	18
		norm	estic consumption, in relation to land, means consumption for all household purposes (including garden watering) in domestic isses situated on the land.	19 20 21
		raise	watering, in relation to land, means the watering of stock being d on the land, but does not include the use of water in connection intensive animal husbandry.	22 23 24

Chapter 3		V	Vater management implementation	
Part 1		Е	Basic landholder rights	
Division	Division 2		rvestable rights	
Division 2		on 2 Harvestable rights		1
45	Harvestable rights		ble rights	2
	(1)		ndholder within a harvestable rights area is entitled, without the for any access licence, water supply work approval or water use oval:	3 4 5
		(a)	to construct and use a dam for the purpose of capturing and storing rainwater run-off, and	6 7
		(b)	to use water that has been captured and stored by a dam so constructed,	8 9
			ecordance with the harvestable rights order by which the area is tituted.	10 11
	(2)	captu	ngle dam may be used both for rainwater run-off that has been ured and other water that has been lawfully taken from a water ce, but only if the harvestable rights order so provides.	12 13 14
	(3)	This	section does not allow a landholder:	15
		(a)	to supply any other land with water that has been captured and stored under this section, or	16 17
		(b)	to construct or use a dam that obstructs the flow of a river, unless the river is declared by the relevant harvestable rights order to be a minor stream for the purposes of this Division.	18 19 20
	(4)	This	section is subject to any deemed regional environmental plan.	21
46	Har	vestal	ble rights orders	22
	(1)		Minister may, by order published in the Gazette, constitute any as a harvestable rights area and may, by the same or a subsequent	23 24

order so published, name the area and fix its boundaries.

specify the following:

that average),

(a)

(b)

(2) The order by which a harvestable rights area is constituted must

rainwater run-off for a landholding in the area.

the proportion of the average rainwater run-off that may be

captured by landholders in the area (being no less than 10% of

the procedures to be followed for calculating the average

25

26

27

28

29

30

31

32

Clause 45

Water Management Bill 2000

Water Management Bill 2000	Clause 46
Water management implementation	Chapter 3
Basic landholder rights	Part 1
Harvestable rights	Division 2

The order may allow an existing dam to be used both for rainwater			
<u>♣</u>	•		
taken from a water source.	3		
The order may also deal with the following matters:	4		
(a) the types and locations of dams that may be used	by a 5		
landholder to capture and store rainwater run-off,	6		
(b) the means by which the maximum capacity of a dam that	t may 7		
be used by a landholder to capture and store rainwater ru	in-off 8		
is to be calculated,	9		
(c) the arrangements that may be made by landholders of adjo	ining 10		
land for the shared use of a single dam for the captu	re of		
rainwater run-off,	12		
(d) such other matters as are necessary or convenient to give	effect 13		
to the order.	14		
For the purpose of calculating any matter under an order under	r this 15		
that is valued under the Valuation of Land Act 1916, a reference	to the 17		
area of a portion or parcel of land that is separately valued unde	r that 18		
Act.	19		
An order under this section may deal with any matter by referer	ice to 20		
a map held in the head office of the Department.	21		
Any map that is so referred to is to be available for public inspec	ction. 22		
order relates.	25		
	run-off that has been captured and other water that has been law taken from a water source. The order may also deal with the following matters: (a) the types and locations of dams that may be used landholder to capture and store rainwater run-off, (b) the means by which the maximum capacity of a dam that be used by a landholder to capture and store rainwater run is to be calculated, (c) the arrangements that may be made by landholders of adjournal for the shared use of a single dam for the capturainwater run-off, (d) such other matters as are necessary or convenient to give to the order. For the purpose of calculating any matter under an order unde section, a reference in the order to an area of land is, in the case of that is valued under the <i>Valuation of Land Act 1916</i> , a reference area of a portion or parcel of land that is separately valued under Act. An order under this section may deal with any matter by reference a map held in the head office of the Department. Any map that is so referred to is to be available for public inspective of charge, during normal office hours at the head office of Department and at the regional office for the area to which the release of the property of the public inspective of the pu		

			Access licences Preliminary		
Part	2 /	Acce	ess licences	1	
Divis	Division 1 Preliminary		Preliminary	2	
47	Acc	Access licences			
	(1)	An a	access licence entitles its holder:	4	
		(a)	to specified shares in the available water within a specified water management area or from a specified water source (the <i>share component</i>), and	5 6 7	
		(b)	to take that share of water at specified times, at specified rates, in specified circumstances or in specified zones or locations, or in any combination of these (the <i>extraction component</i>).	8 9 10	
	(2)		out limiting subsection (1) (a), the share component of an access ce may be expressed:	11 12	
		(a)	as a specified volume over a specified period, or	13	
		(b)	as a specified proportion of the storage capacity of a dam or other storage work.	14 15	
	(3)		es in available water may be allocated generally or to specified gories of access licence.	16 17	
	(4)	acces	Minister may, by order published in the Gazette, declare that the ss licences for specified water management areas are to be issued rately for the share components and extraction components.	18 19 20	
	(5)	licen any r of an the h	the case of a water management area for which separate access are issued for the share component and extraction component, requirement of this or any other Act for a person to be the holder access licence is taken to be a requirement for the person to be holder of both an access licence for the share component and an ass licence for the extraction component.	21 22 23 24 25 26	
	(6)		he purposes of this Act, the entitlements under an access licence eferred to as the <i>water allocation</i> under the licence.	27 28	
48	Cat	egorie	es of licence	29	
		_	e are the following categories of access licence:	30	
			local water utility access licences	31	

Water management implementation

Clause 47 Chapter 3

Preliminary		Division 1	
	(b)	major utility access licences,	1
	(c)	regulated river (high security) access licences,	2
	(d)	regulated river (general security) access licences,	3
	(e)	regulated river (opportunistic water) access licences,	4
	(f)	unregulated river access licences,	5
	(g)	aquifer access licences,	6
	(h)	estuarine water access licences,	7
	(i)	coastal water access licences,	8
	(j)	such other categories of access licence as may be prescribed by the regulations.	9 10
	held lanyor acces	Local water utility access licences and major utility access licences will be by local water utilities and major utilities, respectively. Those utilities (like ne else) may also hold other categories of access licences. However, an ass licence does not become a local water utility access licence or major utility as licence merely because it is held by a local water utility or major utility.	11 12 13 14 15
49 Pric	orities	between different categories of licence	16
(1)		he purposes of this Act, the following priorities are to be observed lation to access licences:	17 18
	(a)	local water utility access licences and major utility access licences have priority over all other access licences,	19 20

regulated river (high security) access licences have priority over

regulated river (general security) access licences and regulated

regulated river (general security) access licences have priority

over regulated river (opportunistic water) access licences.

another access licence (the lower priority licence), then if the water

allocations under them have to be diminished, the water allocations of

the higher priority licence are to be diminished at a lesser rate than the

(2) If one access licence (the *higher priority licence*) has priority over

river (opportunistic water) access licences,

water allocations of the lower priority licence.

Available water determinations

(b)

(c)

50

Water Management Bill 2000

Access licences

Water management implementation

(1) From time to time the Minister may, by order in writing, determine the amount of water that is available in a specified water management area or water source.

Clause 48 Chapter 3

21

22

23

24

25

26

27

28

29

30

31

32

33

34

Part 2

Clause 50 Chapter 3 Part 2 Division 1		Water Management Bill 2000 Water management implementation Access licences Preliminary				
	(2)		termination under this section (a <i>general determination</i>) must be istent with:	1 2		
		(a)	any relevant management plan or Minister's plan, and	3		
		(b)	any relevant implementation plan.	4		
	(3)		regulations may make provision for or with respect to the manner hich a determination under this section is to be publicly notified.	5 6		
Division 2		2	Granting and renewal of access licences			
51 Gra		Granting and renewal of access licences				
	(1)		ect to any embargo, any person may apply to the Minister for an ss licence.	9 10		
	(2)	In the	e case of an application under subsection (1) for an access licence:	11		
		(a)	for water in an area that is not within a water management area, or	12 13		
		(b)	for water in a water management area for which there is no water sharing management plan in force,	14 15		
			Ainister must cause the application to be advertised in accordance the regulations.	16 17		
	(3)	befor	holder of an access licence may, at any time within the 12 months re the access licence expires, apply to the Minister for a renewal e access licence.	18 19 20		
52	Obj	Objections to granting of access licences				
	(1)		person may, in accordance with the regulations, object to the ting of an access licence:	22 23		
		(a)	for water in an area that is not within a water management area, or	24 25		
		(b)	for water in a water management area for which there is no water sharing management plan in force.	26 27		
	(2)	grounthe a	Minister must inform the applicant for the access licence of the nds of any objection to the granting of the licence and must allow pplicant a specified time within which to make a written response e Minister in relation to the objection.	28 29 30 31		

Water Management Bill 2000	Clause 52
Water management implementation	Chapter 3
Access licences	Part 2
Granting and renewal of access licences	Division 2

	(3)	The Minister:			
		(a)	may require an objector or applicant to provide additional information within a specified time if of the opinion that additional information would be relevant to consideration of the objection or response, and	2 3 4 5	
		(b)	may delay consideration of the objection or response until the information is provided or, if the information is not provided within the time specified, may refuse to consider the objection or response.	6 7 8 9	
	(4)	notify	re is a deficiency in an objection or response, the Minister may the objector or applicant accordingly and allow further time to e the deficiency to be rectified.	10 11 12	
	(5)	respected endead consu	re making a decision on an application for an access licence in ct of which any objection has been made, the Minister must avour to resolve the issues raised by the objection by means of ultation with the applicant and the objector, with a view to ing agreement on the matters raised by the objection.	13 14 15 16	
	(6)	of me or eva	he purpose of reaching such an agreement, the Minister may ose that the matters raised by the objection be dealt with by way ediation or neutral evaluation involving an independent mediator aluator selected by agreement between the applicant, the objector he Minister.	18 19 20 21 22	
	(7)	applic	opplication or objection may be dismissed by the Minister if the cant or objector, as the case may be, fails to participate in any ation or neutral evaluation proceedings referred to in subsection	23 24 25 26	
53	Det	ermina	ation of applications	27	
	(1)		Minister may determine an application for an access licence by ing or refusing to grant the licence.	28 29	
	(2)	An acthat:	ccess licence is not to be granted unless the Minister is satisfied	30 31	
		(a)	the granting of an access licence, or an access licence of the category to which the application relates, is permitted by the relevant provisions of a management plan or Minister's plan, and	32 33 34 35	

			Vater Management Bill 2000 Vater management implementation	
Part 2	J. 0		access licences	
		G	Granting and renewal of access licences	
		(b)	adequate arrangements are in force to ensure that minimal harm will be done to any water source as a consequence of water being taken from the water source under the licence.	1 2 3
54	Not	ice of	decision	4
		cause	determining an application under this Division, the Minister must enotice of the determination to be given to the applicant and, if pplication relates to:	5 6 7
		(a)	an area that is not within a water management area, or	8
		(b)	a water management area for which there is no water sharing management plan in force,	9 10
			ch person who has made an objection in connection with the cation.	11 12
55	Cor	ntrolle	d allocation of access licences	13
	(1)	The l	Minister may, by order published in the Gazette:	14
		(a)	impose an embargo on applications for access licences for a specified water management area, and	15 16
		(b)	declare that access licences for that area are to be allocated by auction, tender or any other means specified in the order.	17 18
	(2)	mana	e such an order is in force, access licences for the water agement area to which the order relates are to be allocated in rdance with the terms of the order and not otherwise.	19 20 21
Divis	ion :	3	Conditions and duration of access licences	22
56	Cor	ndition	ns of access licence	23
	(1)	An a	ccess licence is subject to:	24
		(a)	such conditions as are from time to time required to be imposed on the access licence by the relevant management plan or Minister's plan (<i>mandatory conditions</i>), and	25 26 27
		(b)	such other conditions as the Minister may from time to time impose on the access licence (<i>discretionary conditions</i>).	28 29

Acces Condi			Part 2 ration of access licences Division 3				
	(2)		andatory condition prevails over a discretionary condition to the nt of any inconsistency between them.	1 2			
			If a management plan or Minister's plan is replaced or amended during the of an access licence, the mandatory conditions applying to the access licence vary.	3 4 5			
57	lmp	Imposition of conditions after access licence is granted					
	(1)		retionary conditions may not be imposed on an access licence after s been granted unless the Minister:	7 8			
		(a)	has given written notice to the holder of the access licence that the Minister proposes to impose such conditions, and	9 10			
		(b)	has given the holder of the access licence a reasonable opportunity to make submissions to the Minister with respect to the proposed conditions, and	11 12 13			
		(c)	has taken any such submissions into consideration.	14			
	(2)		section (1) does not apply to conditions imposed on an access ce at the request of the holder of the access licence.	15 16			
	(3)	impo	Minister must cause written notice of any discretionary conditions used on an access licence after the time it is granted to be served the holder of the access licence.	17 18 19			
	(4)	the d	scretionary condition referred to in subsection (3) takes effect on lay on which the notice referred to in that subsection is served on nolder of the access licence or on such later day as may be ified in the notice in that regard.	20 21 22 23			
58	Rev	ocatio	on of conditions	24			
		whic	Minister may at any time revoke any discretionary conditions to the an access licence is subject, whether or not on the application e holder of the access licence.	25 26 27			
59	Dur	ation	of access licences	28			
			ccess licence has effect for such period as may be specified in the ce in that regard, being a period not exceeding:	29 30			
		(a)	20 years, in the case of a local water utility access licence or major utility access licence, or	31 32			

Water management implementation

Clause 56

Chapter 3

Clause 59 Chapter 3 Part 2 Division 3		V A	Water Management Bill 2000 Water management implementation Access licences Conditions and duration of access licences		
		(b)	2 years, in the case of a regulated river (opportunistic water) access licence, or	1 2	
		(c)	15 years, in any other case.	3	
Division 4		4	Access licence transfers	4	
60 Transfer		nsfer	principles	5	
			Minister may, by order published in the Gazette, establish transfer ciples for the purposes of this Division.	6 7	
61	Tra	Transfer of access licences		8	
	(1)		parties to a proposed transfer of an access licence may apply to the ster for consent to the transfer.	9 10	
	(2)	The application may relate to:		11	
		(a)	the whole of the access licence for the whole of the period for which the access licence is in force, or	12 13	
		(b)	the whole of the access licence for part only of the period for which the access licence is in force, or	14 15	
		(c)	part only of the access licence for the whole of the period for which the access licence is in force, or	16 17	
		(d)	part only of the access licence for part only of the period for which the access licence is in force.	18 19	
	(3)		pplication under this section is to be dealt with in accordance with ocal transfer rules.	20 21	
	(4)	Desp	pite subsections (1), (2) and (3):	22	
		(a)	the maximum period for which a local water utility access licence or major utility access licence may be transferred is one year, and	23 24 25	
		(b)	a regulated river (opportunistic water) access licence may not be transferred unless such a transfer is permitted by the relevant management plan or Minister's plan.	26 27 28	
	(5)	section	completing a transfer to which consent has been given under this on, the parties to the transfer must cause notice of that fact to be in to the Minister.	29 30 31	

Acces Acces		ces Part 2 ce transfers Division 4	
	(6)	The transfer takes effect on the date on which details of the transfer are entered on the register of access licences.	
62	Wat	er allocations may be transferred	3
	(1)	The water allocations conferred by an access licence may be transferred in accordance with this section.	4
	(2)	Two or more holders of access licences may apply to the Minister for the transfer of water allocations between their respective access licences.	6 7 8
	(3)	An application under this section is to be dealt with in accordance with the local transfer rules.	<u>9</u> 10
63	Inte	rstate transfers	11
	(1)	The Minister may enter into an agreement with a Minister of a participating State or Territory for the interstate transfer of access licences and their corresponding interstate equivalents.	12 13 14
	(2)	Any interstate equivalent to an access licence that is transferred in accordance with such an agreement so as to become available for use in New South Wales is taken to be an access licence for the purposes of this Act.	15 16 17 18
	(3)	In this section, <i>participating State or Territory</i> means a State or Territory that has laws relating to the distribution of water rights that are declared by the regulations under this Act to be substantially similar to the provisions of this Act.	19 20 21 22
Divis	ion (5 Water return flows	23
64	Wat	er return flow rules	24
		The Minister may, by order published in the Gazette, establish water return flow rules for the purposes of this Division.	25 26
65	Wat	er allocations may be credited	27
	(1)	Water allocations that have been used by the holder of an access licence may be regained in accordance with this section.	28 29
	(2)	The holder of an access licence may apply to the Minister for used water allocations to be recredited to the licence.	30 31

Water management implementation

Clause 61

Chapter 3

Chapter 3 Part 2		V A	Nater Management Bill 2000 Nater management implementation Access licences Nater return flows	
	(3)		application under this section is to be dealt with in accordance with water return flow rules.	:
Divis	Division 6 Surrender, suspension, cancellation and compulsory acquisition of access licences			2
66	Sur	rende	er of access licences	4
	(1)		holder of an access licence may surrender the licence at any time otice in writing sent to the Minister.	(
	(2)		surrender takes effect on the date on which the notice is received ne Minister or such later date as is specified in the notice.	Ş
67	Sus	pensi	ion and cancellation of access licences	10
	(1)		Minister may suspend or cancel an access licence on any one or e of the following grounds:	11 12
		(a)	that the holder of the licence has failed to comply with the conditions to which the licence is subject,	13 14
		(b)	that the holder of the licence has been convicted of an offence against this Act or the regulations,	15 16
		(c)	if any charges payable in respect of the licence have not been paid.	17 18
	(2)		on under this section may not be taken in relation to an access ace unless the Minister:	19 20
		(a)	has given written notice to the holder of the access licence that the Minister proposes to take such action, and	21 22
		(b)	has given the holder of the access licence a reasonable opportunity to make submissions to the Minister with respect to the proposed action, and	23 24 25
		(c)	has taken any such submissions into consideration.	26
68	Cor	npuls	ory acquisition of access licences	27
	(1)	The comp	Minister may, by notice in writing served on their holders, pulsorily acquire access licences if of the opinion that, in the ial circumstances of the case, the public interest requires their pulsory acquisition	28 29 30

Water Acces		gement implementation	Chapter 3 Part 2	
	nder, s	suspension, cancellation and compulsory acquisition of	Division 6	
	(2)	A person from whom an access licence is compulsor entitled to compensation from the State for the mark licence as at the time it was compulsorily acquired.		
	(3)	The amount of compensation payable is to be agreement between the Minister and the personal compensation or, if agreement cannot be reached, is to by the Valuer-General.	on entitled to	
	(4)	A person who is dissatisfied with the amount of competo the person under this section, or with any delay in compensation, may appeal to the Land and Environment	the payment of	1
	(5)	The regulations may make provision for or with respect of compensation under this section.	t to the payment	1
	(6)	Nothing in this section prevents the Ministerial Co acquiring an access licence by way of transfer.	rporation from	1
	(7)	For the avoidance of doubt, it is declared that a reducti allocations under an access licence as a consequence of the mandatory conditions of the licence does not compulsory acquisition of an access licence or any pa- licence.	of a variation in constitute the	1 1 1 1
Divis	sion	7 Embargoes on applications for access	licences	2
69	Ten	nporary embargo		2
	(1)	The Minister may, by order published in the Gaze embargo on the making of applications for access licent to any water management area.		2 2 2
	(2)	An order under this section may relate to all a applications of a specified kind or to all applications of a specified kind.		2 2 2
	(3)	A copy of any order under this section is to be publinewspaper.	ished in a local	2
	(4)	An order under this section takes effect on the date	on which it is	3

published in the Gazette and, unless sooner revoked, expires at the end

of 2 years after that date.

Water Management Bill 2000

31

32

Part 2				
Divisio	on 7	E	Embargoes on applications for access licences	
70	Per	mane	nt embargo	1
	(1)	an ei	Governor may, by proclamation published in the Gazette, declare mbargo on the making of applications for access licences with ect to any water management area.	2 3 4
	(2)	appli	oclamation under this section may relate to all applications, to ications of a specified kind or to all applications other than ications of a specified kind.	5 6 7
	(3)		py of any proclamation under this section is to be published in a newspaper.	8
	(4)	is pu	oclamation under this section takes effect on the date on which it blished in the Gazette and continues in force until it is revoked by ther proclamation so published.	10 11 12
71	Оре	Operation of embargo		13
	(1)	An e	embargo does not apply to:	14
		(a)	an application made before the embargo took effect, or	15
		(b)	an application to amend an application made before the embargo took effect (other than an application to amend an application for an access licence in such a manner as to increase the amount of water sought by the proposed access licence), or	16 17 18 19
		(c)	an application for a transfer of an access licence, or	20
		(d)	any other application of a kind prescribed by the regulations.	21
	(2)		application to which an embargo relates is a nullity and is not yed merely because the embargo is subsequently revoked.	22 23
Divis	sion 8	3	Registers and registration	24
72	Reg	Register of access licences		
	(1)	The	Minister is to cause a register to be kept of:	26
		(a)	every application for an access licence that is duly made under this Act, and	27 28
		(b)	every access licence that is granted, renewed, transferred, surrendered, suspended or cancelled under this Act.	29 30

Chapter 3

Water Management Bill 2000

Water management implementation

	Water management implementation Chapter 3 Access licences Part 2				
			stration	Part 2 Division 8	
	(2)	whic	regulations may make provision for or with resp ch such a register is to be kept and the particul rded in such a register.		1 2 3
	(3)		register must be made available at the heartment for inspection, free of charge, by members		4 5
73	Reg	gister	of available water determinations		6
	(1)		Minister is to cause a register to be kept of eacrmination made under section 50.	h available water	7 8
	(2)	whic	regulations may make provision for or with resp ch such a register is to be kept and the particul rded in such a register.		9 10 11
	(3)		register must be made available at the heartment for inspection, free of charge, by member		12 13
74	Wat	ter all	ocation accounts		14
	(1)	For e	each access licence, the Minister is to cause an a	account to be kept	15 16
		(a)	the water allocations that are acquired or acci time under the licence, and	rued from time to	17 18
		(b)	the water allocations that are used or transfer time under the licence.	rred from time to	19 20
	(2)	whic	regulations may make provision for or with respect such an account is to be kept and the particularded in such an account.		21 22 23
75	Inte	rest i	n access licence to be created by instrument	t in writing	24
	(1)		gal or equitable interest in an access licence may osed of except by instrument in writing.	not be created or	25 26
	(2)	not a	creation of a legal or equitable interest in an acuffect the liability of the holder of the licence for litions of the licence or of any of the provisions lations.	any breach of the	27 28 29 30
76	Reg	gistrat	ion of certain interests		31
	(1)	•	person claiming a legal or equitable interest in apply for registration of the interest.	an access licence	32 33

Chapter 3 \ Part 2		Water Management Bill 2000			
			Water management implementation Access licences		
Divisio	n 8	R	egisters and registration		
	(2)	An application must be lodged with the Minister and accompanied by documentary evidence of the legal or equitable interest concerned.		1	
	(3)	of the	eceipt of the application, particulars of the documentary evidence e legal or equitable interest are to be entered in the register of ss licences.	2	
	(4)	On a	pplication by the holder of the interest or otherwise, the Minister:	Ć	
		(a)	may make such amendments to the register as are appropriate to reflect dealings in the interest, and	7	
		(b)	may cancel the registration of the interest if satisfied that the interest has ceased to exist.	<u>9</u> 10	
	(5)		registration of an interest under this section is not to be taken to ridence of the existence of the interest.	11 12	
(6)		For the purposes of any legal proceedings concerning an access licence:		13 14	
		(a)	a registered interest has priority over an unregistered interest, and	15 16	
		(b)	an earlier registered interest has priority over a later registered interest.	17 18	
77	Dev	olutio	n of rights of holder of access licence	19	
		A pe	rson on whom the rights of the holder of an access licence have	20	
			lved by operation of law may apply to the Minister to have that	2	
			on's name recorded as the holder of the licence and, if the Minister	22	
		is satisfied that those rights have so devolved, the Minister may so record the name of the applicant.			
			••		
Divis	ion 9	9	Compensation when management plan bulk	25	
			access regime varied	26	
78			ation payable for reductions in water allocations arising from	27	
	Min	ıster's	amendment of management plan	28	
	(1)		e end of each prescribed period, each holder of an access licence	29	
			r than a regulated river (opportunistic water) access licence)	30	
			se water allocations have been adversely affected during that	31	
		perio	d as a consequence of the variation of a bulk access regime under	32	

Access licences			Chapter 3 Part 2		
			Division 9		
			on 38 may claim compensation for any loss suffections equence of that reduction.	ered by the holder	1 2
	(2)		regulations may make provision for or with respondent in which such a claim is to be made.	spect to the manner	
	(3)	The Minister, with the concurrence of the Treasurer, may determine whether or not compensation should be paid and, if so, the amount of any such compensation and the manner and timing of any such payments.			5 6 7 8
	 (4) No appeal lies against any decision of the Minister of under this section, and any such decision is not to question in any legal proceedings. (5) Payment of compensation under this section is to be a Consolidated Fund which is, to the extent necessar accordingly. (6) In this section, <i>prescribed period</i> means: 			9 10 11	
			solidated Fund which is, to the extent necessary		12 13 14
			s section, <i>prescribed period</i> means:		15
		(a)	in the case of access licences for a water man which a management plan is in force, the peri that plan is in force, or		16 17 18
		(b)	in any other case, any of the successive pe commencing on the date on which this section		19 20
Division 10 Miscellaneous				21	
79	Reg	julatio	ns		22
			regulations may make provision for or with wing matters:	n respect to the	23 24
		(a)	the procedures to be followed in relation to applications under this Part,	o the making of	25 26
		(b)	the procedures to be followed in relation renewal and transfer of access licences,	to the granting,	27 28
		(c)	the procedures to be followed in relation to the allocations under an access licence,	transfer of water	29 30
		(d)	the procedures to be followed in relation to t water allocations under an access licence.	he recrediting of	31 32

32

Division 1		Preliminary		
Part	3 /	Approvals	1	
Divis	ion '	1 Preliminary	2	
80	Wat	ter use approvals	3	
		A water use approval confers a right on its holder to use water for a particular purpose at a particular location.	4 5	
81	Water management work approvals			
	(1)	There are three kinds of water management work approvals, namely, water supply work approvals, drainage work approvals and flood work approvals.	7 8 9	
	(2)	A water supply work approval authorises its holder to construct and use a specified water supply work at a specified location.	10 11	
	(3)	A drainage work approval confers a right on its holder to construct and use a specified drainage work at a specified location.	12 13	
	(4)	A flood work approval confers a right on its holder to construct and use a specified flood work at a specified location within a floodplain.	14 15	
82	Act	ivity approvals	16	
	(1)	There are two kinds of activity approvals, namely, controlled activity approvals and aquifer interference approvals.	17 18	
	(2)	A controlled activity approval confers a right on its holder to carry out a specified controlled activity at a specified location in waterfront land or at a specified location in a water source protection zone.	19 20 21	
	(3)	An aquifer interference approval confers a right on its holder to penetrate or interfere with an aquifer at a specified location, or in a specified area, in the course of carrying out specified activities.	22 23 24	
		Note. Examples of where an aquifer interference approval may be needed include mining operations, road construction and any other large scale activity that involves excavation.	25 26 27	

Chapter 3

Part 3

Water Management Bill 2000

Approvals

Water management implementation

Water Management Bill 2000	Clause 83
Water management implementation	Chapter 3
Approvals	Part 3
Preliminary	Division 1

83	Exemption from requirement for approval			
	(1)	The Minister may, by order in writing, exempt:		2
		(a)	any person or class of persons, or	3
		(b)	any works or class of works, or	۷
		(c)	any controlled activity or class of controlled activities, or	5
		(d)	any building or class of buildings,	ϵ
	from the need for an approval.			
	(2)	2) Unless sooner revoked, an order under this section has effect for such period (not exceeding 3 years) as is specified in the order.		
	(3)		is section, <i>controlled activity</i> includes the activity of penetrating quifer, or interfering with water in an aquifer or obstructing its	10 11 12
Divis	ion 2	2	Applications for approvals	13
84	Applications for approvals			
	(1)		application for an approval must be made to the Minister in rdance with the regulations.	15 16
	(2)		regulations may require the application to be accompanied by a agement program for the land to which the application relates.	17 18
	(3)	An application may relate to more than one approval, whether of the same or of a different kind, unless the Minister requires a separate application to be made in relation to one or more of them.		19 20 21
	(4)	The I	Minister:	22
		(a)	may require an applicant for an approval to provide additional information within a specified time if of the opinion that additional information would be relevant to consideration of the application, and	23 24 25 26
		(b)	may delay consideration of the application until the information is provided or, if the information is not provided within the time specified, may refuse to consider the application.	27 28 29
	(5)		Minister may refuse to accept an application for an approval if it ars to the Minister that the application is incomplete.	30 31

Clause Chapte Part 3 Divisio	er 3	Water Management Bill 2000 Water management implementation Approvals Applications for approvals		
	(6)	must	e case of an application for a water use approval, the Minister cause the application to be advertised in accordance with the ations.	
		Asses	Approvals are listed in section 91 of the <i>Environmental Planning and</i> sment Act 1979. Development that requires an approval is consequently ated development for the purposes of that Act.	4
85	Obj	ection	s to applications for water use approvals	7
	(1)		person may, in accordance with the regulations, object to the ing of a water use approval.	9
	(2)	grour allow	Minister must inform the applicant for a water use approval of the ads of any objection to the granting of the approval and must the applicant a specified time within which to make a written use to the Minister in relation to the objection.	10 11 12 13
	(3)	The N	Minister:	14
		(a)	may require an objector or applicant to provide additional information within a specified time if of the opinion that additional information would be relevant to consideration of the objection or response, and	15 16 17 18
		(b)	may delay consideration of the objection or response until the information is provided or, if the information is not provided within the time specified, may refuse to consider the objection or response.	19 20 21 22
	(4)	notify	re is a deficiency in an objection or response, the Minister may the objector or applicant accordingly and allow further time to e the deficiency to be rectified.	23 24 25
	(5)	in res endea consu	re making a decision on an application for a water use approval spect of which any objection has been made, the Minister must avour to resolve the issues raised by the objection by means of altation with the applicant and the objector, with a view to ing agreement on the matters raised by the objection.	26 27 28 29 30
	(6)	For the proper of me or ev	the purpose of reaching such an agreement, the Minister may ose that the matters raised by the objection be dealt with by way ediation or neutral evaluation involving an independent mediator valuator appointed by agreement between the applicant, the tor and the Minister.	31 32 33 34 35
	(7)		osts of any such mediation or neutral evaluation are to be paid for e. Minister	36 37

Appro Applic		for approval		art 3 ivision 2		
	(8)	applicant	cation or objection may be dismissed by the Min or objector, as the case may be, fails to particip n or neutral evaluation proceedings referred to in	oate in any	1 2 3 4	
86	Determination of applications affected by public inquiries under the Environmental Planning and Assessment Act 1979					
	(1)	which a C	ion applies to any application for an approval in Commission of Inquiry has given a section 120A not before the Minister makes a decision on the appli	otice to the	7 8 9	
	(2)	The Minis	ster:		10	
		(a) mu (i) (ii)	relation to the application), and	ading any	11 12 13 14 15	
			nether the application or objection is made or lod after the section 120A notice is received, and	ged before	16 17	
			ust defer making any decision on the application ceives the Commission of Inquiry's section 119 r		18 19	
	(3)	to the find	g a decision on the application, the Minister must have and recommendations contained in the Comsection 119 report.		20 21 22	
	(4)	In this sec	ction:		23	
			ion of Inquiry means a Commission of Inquiry of the Environmental Planning and Asse		24 25 26	
			19 report means a report referred to in section 119 nental Planning and Assessment Act 1979.	9 (6) of the	27 28	
			20A notice means a notice referred to in section 1 conmental Planning and Assessment Act 1979.	20A (1) of	29 30	
87	Det	ermination	of applications		31	
	(1)		nsidering an application and all matters releven, the Minister is to determine the application:	ant to the	32 33	
		(a) by	granting the approval to which the application re	elates, or	34	

Water Management Bill 2000 Water management implementation Clause 85

Chapter 3

Clause 87		Water Management Bill 2000			
Chapter 3 Part 3 Division 2		Water management implementation			
		Applications for approvals			
		Applications for approvals	_		
		(b) by refusing the application.	1		
	(2)	A single approval may be granted in relation to more than one wat	er 2		
		management work or activity and in relation to more than one kind water management work or activity.	of 3 4		
	(3)	An approval may not be granted in contravention of the provisions any relevant management plan.	of 5		
88 Ma		atters affecting consideration of applications	7		
		In considering whether or not to grant an approval, the Minister mutake into account:	st 8 9		
		(a) such matters as are prescribed by the regulations, and	10		
		(b) such other matters as the Minister considers to be relevant.	11		
89	Gro	ounds of refusal of certain applications	12		
	(1)	A water use approval is not to be granted unless the Minister	is 13		
		satisfied that adequate arrangements are in force to ensure that minim			
		harm will be done to any water source, or its dependent ecosystems,			
		a consequence of the proposed use of water on the land in respect which the approval is to be granted.	of 16 17		
	(2)	A drainage work approval is not to be granted unless the Minister	is 18		
		satisfied that adequate arrangements are in force to ensure that minimal			
		harm will be done to any water source, or its dependent ecosystems,			
		a consequence of the construction or use of the proposed drainage work.	ge 21 22		
	(3)	A flood work approval is not to be granted unless the Minister	is 23		
		satisfied that adequate arrangements are in force to ensure that minim			
		harm will be done to any water source, or its dependent ecosystems,			
		a consequence of the construction or use of the proposed flood wor	k. 26		
	(4)	A controlled activity approval is not to be granted unless the Minist	er 27		
		is satisfied that adequate arrangements are in force to ensure the			
		minimal harm will be done to any waterfront land or water source			
		protection zone as a consequence of the carrying out of the propose controlled activity.	ed 30 31		
	(5)	•	rk 32		
	(-)	that is, or is proposed to be, situated on land not owned by the			
		applicant is not to be granted unless the Minister is satisfied:	34		

Appro		for ap	provals Part 3 Division 2	
		(a)	that the applicant is likely to become the owner of the land within a reasonable time, or	
		(b)	that the land is subject to an easement that authorises the construction and use of such a work for the benefit of the applicant's land.	
	(6)	Mini that ecos	aquifer interference approval is not to be granted unless the ister is satisfied that adequate arrangements are in force to ensure minimal harm will be done to the aquifer, or its dependent ystems, as a consequence of its being interfered with in the course he activities to which the approval relates.	1
90	Not	ice of	decision	1
		caus the a	r determining an application under this Division, the Minister must e notice of the determination to be given to the applicant and, if application relates to a water use approval, to each person who has e an objection to the Minister in connection with the application.	1 1 1
91	Pro	cedur	res in relation to integrated development	1
	(1)	deve	is section, consent authority, development consent and integrated dopment have the same meanings as they have in the ironmental Planning and Assessment Act 1979.	1 1 1
	(2)	appl	provisions of this Act with respect to the advertising of an ication for a water use approval do not apply if the application es to integrated development.	2 2 2
	(3)	deve term cons	consent authority grants development consent to integrated elopment, either as a consequence of having obtained the general as of any approval proposed to be issued by the Minister or as a sequence of the Minister having failed to inform the consent ority as to whether or not an approval will be granted:	2 2 2 2 2
		(a)	no person has any right to lodge an objection under this Act to the granting of the approval, and	2 2
		(b)	no person has any right under this Act to appeal to the Land and Environment Court against the granting of the approval.	3
	(4)		section (3) (b) does not affect any right of appeal to which an ctor may be entitled under section 98 of the <i>Environmental</i>	3

Planning and Assessment Act 1979.

Water Management Bill 2000

Water management implementation

34

Clause 89

Chapter 3

Clause 92	water Management Bill 2000	
Chapter 3	Water management implementation	
Part 3	Approvals	
Division 3	Conditions and duration of approvals	

Division 3 Conditions and duration of approvals

92	Cor	dition	s of approval generally	2	
	(1)	An a	pproval is subject to:	3	
		(a)	such conditions as are from time to time required to be imposed	4	
		` /	on the approval by the relevant management plan or Minister's	5	
			plan (mandatory conditions), and	6	
		(b)	such other conditions as the Minister may from time to time	7	
			impose on the approval (discretionary conditions).	8	
	(2)	A ma	andatory condition prevails over a discretionary condition to the	9	
			nt of any inconsistency between them.	10	
		Note.	If a management plan or Minister's plan is replaced or amended during the	11	
		term c	of an approval, the mandatory conditions applying to the approval may vary.	12	
93	Cor	dition	s of approval for joint water supply schemes	13	
	(1)	This section applies to a water supply work approval granted in			
			on to a joint water supply scheme, that is, a scheme under which	15	
		landholders of different parcels of land hold a single water supply		16 17	
		work approval for a water supply work located on, or passing through,			
			those parcels.	18	
	(2)		ater supply work approval for a water supply work the subject of	19	
			nt water supply scheme is subject to such conditions as are	20	
			red to be imposed on the approval by the regulations, being	21	
			itions as to the rights and duties of the landholders concerned in on to:	22 23	
		(a)	the granting of access to the work, and	24	
		(b)	the apportionment of water supplied by means of the work, and	25	
		(c)	the apportionment of the cost of constructing and maintaining	26	
			the work, and	27	
		(d)	the apportionment of such other costs relating to the work as	28	
			are prescribed by the regulations, and	29	
		(e)	the exclusion of land from the operation of the scheme.	30	
94	lmp	ositio	n of conditions after approval is granted	31	
	(1)	Disci	retionary conditions may not be imposed on an approval after it	32	
			een granted unless the Minister:	33	

1

Appro	vals		nt implementation Chapter 3 Part 3 ration of approvals Division 3	
Condi	tions a	ina aai	Elvision 3	
		(a)	has given written notice to the holder of the approval that the Minister proposes to impose such conditions, and	1 2
		(b)	has given the holder of the approval a reasonable opportunity to make submissions to the Minister with respect to the proposed conditions, and	3 4 5
		(c)	has taken any such submissions into consideration.	6
	(2)		section (1) does not apply to conditions imposed on an approval are request of the holder of the approval.	7 8
	(3)	impo	Minister must cause written notice of any discretionary conditions osed on an approval after the time it is granted to be served on the ler of the approval.	9 10 11
	(4)	the c	scretionary condition referred to in subsection (3) takes effect on day on which the notice referred to in that subsection is served on nolder of the approval or on such later day as may be specified in notice in that regard.	12 13 14 15
95	Rev	ocati	on of conditions	16
		whic	Minister may at any time revoke any discretionary conditions to ch an approval is subject, whether or not on the application of the ler of the approval.	17 18 19
96	Dur	ation	of approval	20
	(1)		approval has effect for such period as may be specified in the oval in that regard, being a period not exceeding:	21 22
		(a)	3 years, in the case of a controlled activity approval, or	23
		(b)	10 years, in the case of a water use approval or aquifer interference approval, or	24 25
		(c)	20 years, in the case of a water management work approval.	26
	(2)		application for renewal of an approval is lodged before it expires, term of the expiring approval is extended until:	27 28
		(a)	the date of the final decision on the application, or	29
		(b)	a date fixed by the Minister for the approval,	30
		whic	chever is the later date.	31

(3) If:

32

Clause 96		W	Vater Management Bill 2000	
Chapte	er 3	·		
Part 3 Approvals Division 3 Conditions and duration of approvals		Approvals Conditions and duration of approvals		
		——————————————————————————————————————		
		(a)	an approval expires without an application for its renewal being made, and	1 2
		(b)	an application for its renewal is subsequently made by the former holder of an approval and is accompanied by a statutory declaration of the reasons for the delay in making the application, and	3 4 5 6
		(c)	the reasons are accepted by the Minister,	7
		appli	erm of the approval is taken to have been extended, and the cation may be dealt with, as if the application had been made the the approval expired.	8 9 10
97		d ben roval	nefited by water use approval or water management work	11 12
	(1)	be he time	atter use approval or water management work approval is taken to eld by, and for the benefit of, each successive landholder for the being of the land specified in the approval as the land benefited e approval.	13 14 15 16
	(2)	While	e the approval is in force, each such landholder:	17
		(a)	is under a duty to comply with the conditions to which the approval is subject, and	18 19
		(b)	is entitled to commence and maintain legal proceedings against any other such landholder with respect to that other landholder's failure to comply with those conditions.	20 21 22
	(3)		section does not apply in such circumstances as may be ribed by the regulations.	23 24
Division 4		1	Amendment, surrender, suspension and cancellation of approvals	25 26
98		Amendment of water management work approval on application by holder of approval		27 28
	(1)		he application of the holder of a water management work oval, the Minister may amend the approval:	29 30
		(a)	so as to allow an alteration or extension of the relevant water management work, or	31 32

Appro	Vater management implementation Approvals Amendment, surrender, suspension and cancellation of approvals			Chapter 3 Part 3 Division 4	
		(b)	so as to allow the construction or use of an management work on adjoining land occupied the approval.		1 2 3
	(2)	appli mana same	Minister may treat an application under this ication for the grant, on surrender of the agement work approval, of a replacement approve authority as would have been conferred by the east been amended in accordance with the application.	existing water ral conferring the existing approval	5 6 7 8
99	Sur	rende	er of approvals		ç
	(1)		holder of an approval may surrender the approvate in writing sent to the Minister.	al at any time by	10 11
	(2)		surrender takes effect on the date on which the nee Minister or such later date as is specified in the		12 13
100	Sus	pensi	ion and cancellation of approvals		14
	(1)		Minister may suspend or cancel an approval on the following grounds:	any one or more	15 16
		(a)	that the holder of the approval has failed to conditions to which the approval is subject,	comply with the	17 18
		(b)	that the holder of the approval has been convic against this Act or the regulations,	ted of an offence	19 20
		(c)	that the holder of the approval has failed to may with respect to any fee or charge that is payab in relation to the approval,		21 22 23
		(d)	in the case of an approval granted in rela management work, the holder of the approv comply with any direction given to the holder connection with the work.	val has failed to	24 25 26 27
	(2)		on under this section may not be taken in relations the Minister:	n to an approval	28 29
		(a)	has given written notice to the holder of the a Minister proposes to take such action, and	approval that the	30 31
		(b)	has given the holder of the approval a reason to make submissions to the Minister with proposed action, and		32 33 34

has taken any such submissions into consideration.

Water Management Bill 2000

(c)

35

Part 3		Approvals	
Division 5		Embargoes on applications for approvals	
Divis	sion (Embargoes on applications for approvals	1
101	Ten	nporary embargo	2
	(1)	The Minister may, by order published in the Gazette, declare an embargo on the making of applications for water use approvals or water management work approvals with respect to any water management area.	3 4 5 6
	(2)	An order under this section may relate to all applications, to applications of a specified kind or to all applications other than applications of a specified kind.	7 8 9
	(3)	A copy of any order under this section is to be published in a local newspaper.	10 11
	(4)	An order under this section takes effect on the date on which it is published in the Gazette and, unless sooner revoked, expires at the end of 2 years after that date.	12 13 14
102	Per	manent embargo	15
	(1)	The Governor may, by proclamation published in the Gazette, declare an embargo on the making of applications for water use approvals or water management work approvals with respect to any water management area.	1 <i>6</i> 17 18
	(2)	A proclamation under this section may relate to all applications, to applications of a specified kind or to all applications other than applications of a specified kind.	20 21 22
	(3)	A copy of any proclamation under this section is to be published in a local newspaper.	23 24
	(4)	A proclamation under this section takes effect on the date on which it is published in the Gazette and continues in force until it is revoked by a further proclamation so published.	25 26 27
103	Оре	eration of embargo	28
	(1)	An embargo does not apply to:	29
		(a) an application made before the embargo took effect, or	30
		(b) an application to amend an application made before the embargo took effect, or	31 32
		(c) any other application of a kind prescribed by the regulations.	33

Clause 101

Chapter 3

Water Management Bill 2000

Water management implementation

	Water management implementation Chapter 3 Approvals Part 3		
Embar	goes	on applications for approvals	Division 5
	(2)	An application to which an embargo rela revived merely because the embargo is sub	•
Divis	ion (6 Registers	
104	Reg	gister of approvals	
	(1)	The Minister is to cause a register to be ke	pt of:
		(a) every application for an approval th Act, and	at is duly made under this
		(b) every approval that is granted surrendered, suspended or cancelled	
	(2)	The regulations may make provision for or which such a register is to be kept and the recorded in such a register.	
	(3)	The register must be made available at Department for inspection, free of charge,	

Part	4 I	inance		1
105	Det	rmination of recove	rable costs	2
	(1)	The Minister may fro	om time to time make determinations as to:	3
		(a) the costs of ac	lministering this Act, including in particular:	4
			sts associated with the management of water	5
			es, including information collection and gement, monitoring, water allocation, water	6
			ement planning and environmental remediation,	7 8
		and	y	9
			associated with the supply of water, including	10
			delivery operation, asset management and an	11
			mic return on asset investments, and sts associated with compliance management, and	12 13
			those costs are to be recovered from water users	
		\ /	rious water management areas and water sources	14 15
		in the State.	Trous water management areas and water sources	16
	(2)	The regulations may	make provision for or with respect to the manner	17
			s costs referred to in subsection (1) are to be	18
		calculated.		19
	(3)		e after making a determination under this section,	20
			use notice of the determination to be published in	21
		the Gazette.		22
106	Min	ster's guidelines for	charges	23
	(1)	The Minister may	from time to time establish guidelines (the	24
	` ,	Minister's guidelines	y) with respect to the manner in which charges are	25
		to be imposed in con	nection with:	26
		(a) the granting, approvals, and	renewal and transfer of access licences and	27 28
			of applications for the granting, renewal and cess licences and approvals, and	29 30
		(c) the supply of	water to holders of access licences, and	31
		(d) special levies	and other miscellaneous charges.	32
	(2)	The Minister's guide	lines may allow charges to be set differentially:	33
		(a) by reference t	o different areas or water sources, or	34

		(b)	by reference to different classes of access licence or approval, or	1 2
		(c)	by reference to different water usages, or	3
		(d)	by reference to such other factors as are specified in the guidelines.	5
	(3)		Minister's guidelines must minimise cross-subsidies between rent classes of water users.	6
	(4)	varie	Minister's guidelines may provide for charges to be automatically deach year by reference to movements in a specified index (such consumer price index).	8 9 10
	(5)	in the	Minister's guidelines may permit charges to include a component e nature of a resource rent or royalty for water taken under the prity of an access licence.	11 12 13
	(6)		Minister must ensure that the guidelines are periodically reviewed ervals of not more than 5 years.	14 15
107	Cha	arges	payable for access licences and approvals	16
		be as	charges payable in respect of an access licence or approval are to determined by the Minister in accordance with the Minister's elines.	17 18 19
108	Cer	tificate	e as to charges outstanding	20
	(1)		Minister may, in relation to any land, issue a certificate to the t that, as at the date on which the certificate is issued:	21 22
		(a)	a specified amount is payable in relation to the land pursuant to charges imposed under this Part, or	23 24
		(b)	no amount is so payable.	25
	(2)		a certificate is conclusive proof, in favour of a purchaser in good and for value of the land to which it relates, that, as at the date on	2 <i>6</i>

Divisio	n 1 Preliminary	
Cha	pter 4 Joint private works	1
Part	1 Irrigation corporations	2
Divis	ion 1 Preliminary	3
109	Application of Part	4
	This Part applies to all irrigation corporations under the former	5
	Irrigation Corporations Act 1994 that were in existence immediately	6
	before the repeal of that Act, but does not authorise the establishment of any new irrigation corporations.	7 8
110	Definitions	9
	In this Part:	10
	area of operations of an irrigation corporation means the area of land	11
	comprising the areas and districts specified in Schedule 1 in relation to	12
	the corporation, being areas and districts constituted under the former	13
	Irrigation Act 1912 and former Private Irrigation Districts Act 1973.	14
	irrigation corporation means a corporation named in Schedule 1.	15
	operating licence means an operating licence referred to in section	16
	115.	17
111	Requirements for access licences and approvals	18
	(1) Nothing in this Part authorises an irrigation corporation to take water	19
	from a water source otherwise than in accordance with an access	20
	licence and water supply work approval held by the corporation.	21
	(2) Nothing in this Part authorises a landholder of land within an irrigation	22
	corporation's area of operations to use water otherwise than in	23
	accordance with a water use approval held by the corporation.	24

Clause 109

Chapter 4

Part 1

Water Management Bill 2000

Joint private works

Irrigation corporations

Water Management Bill 2000	Clause 112
Joint private works	Chapter 4
Irrigation corporations	Part 1
Irrigation corporations	Division 2

Divis	ion 2	2	Irrigation corporations	1
112	Ass	ets of	irrigation corporations	2
	(1)	An ir	rigation corporation is the owner of all water management works	3
			led in or on land by the corporation (whether or not the land is ed by the corporation).	4 5
	(2)	The p	provisions of Part 5 and section 88 of the Irrigation Corporations	6
			994, as in force immediately before their repeal, continue to apply	7
			ne purpose of enabling the Ministerial Corporation to transfer to	8
			rigation corporation any of its assets, rights and liabilities with	9
			ect to an irrigation scheme area with which the irrigation oration was formerly connected.	10 11
113	Ent	ry on t	to land	12
	(1)	An ir	rigation corporation may, by authorised persons, enter land for	13
		any o	one or more of the following purposes:	14
		(a)	to install, operate, repair, replace, maintain, remove, extend,	15
			expand, connect, disconnect, improve or do any other things	16
			that the corporation considers are necessary or appropriate to	17
			any of its water management works or to construct new water	18
			management works and, for these purposes, to carry out any	19
			work on, below or above the surface of the land,	20
		(b)	to read a meter that:	21
			(i) measures water supplied by the corporation, or	22
			(ii) monitors drainage for quantity or quality or both,	23
		(c)	to find the source of pollution of water within the area of	24
			operations of the corporation,	25
		(d)	to ascertain whether a water supply contract or other contract	26
			for the provision of services by the corporation is being	27
			breached,	28
		(e)	to rectify defective or improper work that has not been rectified	29
			by a customer in accordance with a notice served by the	30
			corporation on the customer under a water supply contract or	31
			other contract,	32

consequential on a corporation ceasing to be an irrigation corporation.

1

3

4

5

8

31

Water Management Bill 2000 Claus	se 115
Joint private works Chap	oter 4
Irrigation corporations Part	1
Operating licences Divis	ion 3

Divis	ion (3	Operating licences	1
115	Aut	hority	conferred by operating licence	2
			operating licence for an irrigation corporation authorises the	3
			pration to carry on the business of supplying water provided to it	4
		•	e Ministerial Corporation and to exercise its functions under this	5
		Part.		6
116	Ter	ms an	d conditions of operating licence	7
	(1)		perating licence is subject to the terms and conditions determined e Governor.	8
	(2)	•	nples of terms and conditions that may be included are as follows:	10
		(a)	a requirement that the irrigation corporation will (in accordance	11
			with any applicable management program and the corporation's	12
			business plan) provide, construct, maintain, manage and	13
			operate:	14
			(i) efficient, co-ordinated and commercially viable systems	15
			and services for supplying water from both surface and	16
			subsurface sources, and (ii) surface and subsurface drainage networks that have	17
			(ii) surface and subsurface drainage networks that have sufficient capacity having regard to specified factors,	18 19
			including the amount of water supplied by the	20
			corporation to users,	21
		(b)	a requirement that the irrigation corporation must be the holder	22
		` /	of all relevant licences or other authorities,	23
		(c)	a requirement that the irrigation corporation is to comply with	24
		(•)	the provisions of any applicable management program, either	25
			in all respects or in certain respects,	26
		(d)	a requirement that, in supplying water to its members, the	27
		()	irrigation corporation is to give priority to certain councils or	28
			other local water utilities for domestic water supply,	29
		(e)	a requirement as to how the irrigation corporation is to spend	30
		` '	and otherwise deal with any money provided to it out of money	31
			appropriated from the Consolidated Fund or other public	32
			money.	33
	(3)		perating licence may be amended only in the manner specified in perating licence.	34 35

Division 3		Operating licences		
117	Ter	n of operating licence		1
	(1)	An operating licence has effect fin that regard, but may be renew	For the period specified in the licence led at the end of that period.	2 3
	(2)	An operating licence may be ren	newed even if its term has expired.	4
	(3)		application for the renewal of an ounds as the Governor considers	5 6 7
118	Cor	travention of operating licence		8
	(1)		an irrigation corporation contravenes or may direct that one of the following	9 10 11
		(a) a letter of reprimand by irrigation corporation,	the Minister is to be served on the	12 13
			notice to be served on the irrigation to rectify the contravention within a	14 15 16
	(2)	contravene an operating licenc subsection (1) (a) or fails to rec	an irrigation corporation continues to e after the issue of a letter under tify a contravention as required by a he Governor may do either or both of	17 18 19 20 21
			hat the irrigation corporation is to pay amount (not exceeding \$150,000) to vernor,	22 23 24
		(b) the Governor may cancel	the operating licence.	25
	(3)	section does not prevent the Gov	rected that action be taken under this ternor directing that the same or other on if the contravention continues or a	26 27 28 29
	(4)		provision for advice to be furnished with the exercise of the Minister's	30 31 32
	(5)	A penalty imposed under this se of competent jurisdiction as if it	ection may be recovered in any court were a debt due to the Crown.	33 34

Clause 117

Chapter 4

Part 1

Water Management Bill 2000

Joint private works

Irrigation corporations

Water Management Bill 2000	Clause 119
Joint private works	Chapter 4
Irrigation corporations	Part 1
Operating licences	Division 3

119	Cancellation of operating licence				
	(1)	An o	operating licence may be cancelled by the Governor, but only:	2	
		(a)	if the irrigation corporation fails or ceases to hold a relevant licence or other authority, or	3	
		(b)	if the irrigation corporation is, in the opinion of the Minister, in material default in compliance with the operating licence, viewed in terms of the operation of the operating licence as a whole, or	5 6 7 8	
		(c)	if the irrigation corporation is an externally-administered corporation within the meaning of the <i>Corporations Law</i> , or	9 10	
		(d)	if the irrigation corporation has been convicted of a criminal offence that is punishable by a fine of at least \$10,000 or, if the corporation were a natural person, imprisonment for 12 months or more, or	11 12 13 14	
		(e)	in the circumstances set out in section 118 (2).	15	
	(2)	in su cance	beginning licence may not be cancelled on the ground referred to bsection (1) (a) if an appeal against a decision not to renew or to el the relevant licence or other authority has been made but not mined or withdrawn.	16 17 18	
	(3)	with	ection (1) (d) does not apply where the offence is unconnected the functions of the irrigation corporation relating to the supply or tage of water.	20 21 22	
120	Irriç	gation	corporation may make arrangements with subsidiaries	23	
	(1)	agree of th	irrigation corporation may arrange, whether by an agency ement or in any other way (except by assignment), for a subsidiary be irrigation corporation to exercise any or all of the irrigation coration's functions under this Part or under an operating licence.	24 25 26 27	
	(2)	irriga	Minister may treat any act or omission of the subsidiary of an ation corporation as an act or omission of the irrigation corporation ne purpose of determining:	28 29 30	
		(a)	whether the irrigation corporation has contravened its operating licence, or	31 32	
		(b)	whether the licence should be suspended or cancelled,	33	
			only if the irrigation corporation has made arrangements for the cise of functions by the subsidiary.	34 35	

			ater Management Bill 2000 int private works	
Part 1			igation corporations	
Divisio	n 3		perating licences	
	(3)	a corp	s section, <i>subsidiary</i> means a corporation that is a subsidiary of poration as determined in accordance with the <i>Corporations Law</i> other applicable law.	1 2 3
Division 4 Miscellaneous			4	
121	Suc	cessor	r in title liable for unpaid contract charges	5
	Division 4 121 Suc		change of ownership of land, the new landholder of the land is	ϵ
Division Division 121			to an irrigation corporation for the amount of any charges levied	7
			e irrigation corporation in relation to the land and unpaid by the bus landholder as if the new landholder had entered into a	9
			act with the irrigation corporation for the supply of the service or	10
			es to which the unpaid charges relate.	11
122	Inde	emnitie	es	12
	(1)		rson does not have any right or claim to relief of any kind	13
			ever in any legal or other proceeding against an irrigation	14
			ration or officer or employee of an irrigation corporation in et of any nuisance connected with or in any way arising out of:	15 16
		-		
		(a)	the design, construction, alteration, maintenance, non-maintenance, operation, repair, disrepair or non-repair of	17 18
			a water management work owned or controlled by the irrigation	19
			corporation, or	20
		(b)	the destruction or damage, or partial destruction or partial	21
		,	damage, by flood, storm, tempest or otherwise of a water	22
			management work owned or controlled by the irrigation	23
			corporation, or	24
		(c)	the exercise, in respect of a water management work owned or	25
			controlled by the irrigation corporation, by the irrigation	26
			corporation of any function conferred or imposed on the irrigation corporation under this or any other Act.	27 28
	(2)	Sectio	on 733 of the <i>Local Government Act 1993</i> applies to and in	29
	` /	respec	ct of an irrigation corporation in the same way as it applies to and	30
		in resp	pect of a council.	31

Water Management Bill 2000	Clause 123
Joint private works	Chapter 4
Irrigation corporations	Part 1
Miscellaneous	Division 4

123	Reg	jister (of licences, audits and management programs	1
	(1)	1) The Ministerial Corporation must maintain at its head office a register containing copies of the following:		2 3
		(a)	operating licences,	4
		(b)	audit reports,	5
		(c)	applicable management programs,	6
		(d)	recommendations of the Minister to the Governor under this Part.	7 8
	(2)		register must be made available for public inspection during nal business hours.	9 10
	(3)	-	person may obtain a copy of any information on the register on nent of the reasonable cost of providing the copy.	11 12

Part	: 2 Pr	rivate irrigation districts	1
Divis	sion 1	Preliminary	2
124	Applie	cation of Part	3
	<i>I</i> t	This Part applies to all private districts under the former <i>Private Irrigation Districts Act 1973</i> that were in existence immediately before the repeal of that Act, and also authorises the establishment of further private irrigation districts.	5
125	Defini	itions	8
	I	In this Part:	ç
		<i>trigated holding</i> means a holding in respect of which water is supplied by a private irrigation board for irrigation.	10 11
		new holding means each part of an original holding that, after subdivision, is separately owned.	12 13
		non-irrigated holding means a holding in respect of which water is supplied by a private irrigation board for domestic and stock use only.	14 15
		<i>priginal holding</i> means a holding in a private irrigation district, being a holding:	16 17
	((a) in existence at the date of constitution of that district, or	18
	((b) comprising lands added to that district under Division 2,	19
	b	out does not include any holding, or any part of a holding, that has become a new holding or that has been excised from the private rrigation district under Division 2.	20 21 22
	n	private irrigation board, in relation to a private irrigation district, means the board of management for that district elected under Division 3.	23 24 25
	ŗ	private irrigation district means a private water supply district or a private water supply and irrigation district constituted under Division 2.	26 27 28

Joint private works Private irrigation districts

Preliminary

Clause 124 Chapter 4

Part 2 Division 1

Prelim	eliminary			Division 1	
126	Rec	quirem	ents f	or access licences and approvals	1
	(1)	other	wise th	this Part authorises a private irrigation board to take water nan in accordance with an access licence and water supply val held by the board.	2 3 4
	(2)	irriga	tion di	this Part authorises a landholder of land within a private istrict to use water otherwise than in accordance with a pproval held by the private irrigation board.	5 6 7
Divis	sion 2	2	Priv	rate irrigation districts	8
127	Pet	ition			9
	(1)	or me	ore ho	s who are landholders of lands that are being worked as 3 oldings may lodge with the Minister a petition for the of those lands:	10 11 12
		(a)	as a	private water supply district, or	13
		(b)	as a	private water supply and irrigation district.	14
	(2)	The r	etition	1:	15
	()	(a)	must	t contain the names, addresses and occupations of all of the ioners and be signed by all of the petitioners, and	16 17
		(b)	must (i) (ii)	the location of the lands proposed to be supplied with water, in relation to the river, estuary or lake from which water is proposed to be obtained, and the lands on which a water supply work is proposed to be constructed or located in connection with the proposed water supply, and the site on those lands of that proposed water supply work, and	18 19 20 21 22 23 24 25
		(c)	must (i) (ii) (iii)	t be accompanied by: particulars of the title of the lands within the proposed private irrigation district, and particulars of the area of land within the proposed private irrigation district owned by each petitioner, and an estimate of the quantity of water proposed to be taken annually for the purposes of the proposed private irrigation district, and	26 27 28 29 30 31 32

Private irrigation districts

Joint private works

Clause 126

Chapter 4 Part 2

Clause 127 Chapter 4 Part 2		· ·					
		Joint private works Private irrigation districts					
			(iv) particulars of any water use approval under which any	1			
			lands within the proposed private irrigation district are,	2			
			at the date of lodgment of the petition, authorised to be	3			
			irrigated, and	4			
		(d)	must contain the names, addresses and occupations of the	5			
			landholders of the lands referred to in paragraph (b) (ii) and be	6			
			accompanied by separate particulars of the title of those lands.	7			
	(3)	The I	Minister may cause a notice containing particulars of the petition	8			
		to be	published in the Gazette and in a local newspaper.	9			
	(4)	Such	a notice may not be published unless:	10			
		(a)	the Minister is satisfied that the establishment of a private	11			
		()	irrigation district in accordance with the petition would be of	12			
			benefit to the landholders of land within the proposed private	13			
			irrigation district, and	14			
		(b)	the Minister has had regard to the Competition Principles	15			
			Agreement, and	16			
		(c)	the Premier has concurred in the publication of the notice.	17			
	(5)	In th	nis section, Competition Principles Agreement means the	18			
		agree	ement of that name between the Commonwealth, the States and	19			
			erritories that was entered into, for and on behalf of New South	20			
		Wale	es, on 11 April 1995.	21			
128	Cor	nstituti	ion of private irrigation districts	22			
	(1)	If a si	upplementary petition in relation to, or an objection to, a petition	23			
		is not	t duly lodged, the Governor may, by proclamation in the Gazette,	24			
		const	titute the lands described in the petition:	25			
		(a)	as a private water supply district, or	26			
		(b)	as a private water supply and irrigation district,	27			
		which	hever was requested in the petition.	28			
	(2)	If a si	upplementary petition in relation to, or an objection to, a petition	29			
	` /		lodged but the Minister recommends the granting of the petition	30			
		•	ther with respect to all of the lands described in the petition or	31			
			e only of them), the Governor may, by proclamation in the	32			
		Gaze	ette, constitute the lands to which the recommendation relates:	33			
		(a)	as a private water supply district, or	34			

Private	e irriga	tion di	stricts Division 2	
		(b)	as a private water supply and irrigation district,	
		whic	hever was requested in the petition.	
	(3)		oclamation under this section:	
	(-)	(a)	must assign a name to the private irrigation district and a	
		` /	corporate name to the private irrigation board, and	
		(b)	must define the boundaries of the private irrigation district, and	
		(c)	must specify at which office of the Ministerial Corporation a plan of the private irrigation district is exhibited, and	
		(d)	must fix a time and place for the first election of the members of the private irrigation board.	
129	Add	dition	of lands to private irrigation districts	
	(1)	A laı	ndholder of lands adjacent to or near a private irrigation district	
			lodge with the Minister a petition for the addition of those lands	
			e private irrigation district.	
	(2)	-	petition:	
		(a)	must specify the name, address and occupation of the petitioner and must be signed by the petitioner, and	
		(b)	must be accompanied by plans showing:	
			(i) the location of the additional lands proposed to be	
			supplied with water, in relation to the river, estuary or lake from which water is proposed to be obtained, and	
			(ii) the lands on which any additional water supply work is	
			proposed to be constructed or located in connection	
			with the proposed water supply, and the site on those	
			lands of that proposed additional water supply work,	
			and	
		(c)	must be accompanied by:	
		•	(i) particulars of the title and the area of the additional	
			lands, and	
			(ii) an estimate of the quantity of water proposed to be	
			taken annually by the petitioner for the purposes of	
			those lands, and	
			(iii) particulars of any water use approval under which those	
			lands are, at the date of lodgment of the petition,	
			authorised to be irrigated, and	

Private irrigation districts

Joint private works

Clause 128

Chapter 4 Part 2

1

3

4

5

6

7

8

Clause 129

Water Management Bill 2000

section does not affect the liability of any person for any rates or

charges levied or leviable in respect of those lands while they were in

the private irrigation district.

Clause 130

1

3

4

5

6

8

33

34

35

132 Supplementary petitions and objections

- (1) Within a period of 28 days after the publication of the notice of a petition for the constitution of a private irrigation district:
 - (a) a person who is the landholder of lands that the person desires to be included in the district to which that notice relates may lodge with the Minister a supplementary petition for the inclusion in that district of lands owned by the person that are being worked as a separate property, or

1

2

3

4

5

6

7

8

Q

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

29

30

31

32

33

34

35

36

37

38

- (b) any person may lodge with the Minister an objection in writing to the petition referred to in the notice on the ground that the granting of that petition would adversely affect the person's interests.
- (2) Within a period of 28 days after the publication of the notice of a petition for the addition of lands to a private irrigation district:
 - (a) the private irrigation board for the district to which that notice relates may lodge with the Minister an objection in writing to the petition referred to in the notice, or
 - (b) any person may lodge with the Minister an objection in writing to the granting of that petition on the ground that the granting of that petition would adversely affect the person's interests.
- (3) A person may not lodge an objection on the ground that the person's interests would, if the petition were granted, be adversely affected for reasons relating to the quantity of water available from the river, estuary or lake from which it is proposed to take water for the purposes of the proposed private irrigation district, and any objection lodged on that ground is not to be entertained.
- (4) A supplementary petition:
 - (a) must contain the name, address and occupation of the supplementary petitioner, and
 - (b) must be accompanied by plans showing:
 - (i) the location of the lands that the supplementary petitioner desires to be included in the proposed district, and
 - (ii) the lands on which any additional water supply work to be used for the taking of water for the use of the lands that the petitioner desires to be included in the proposed district is proposed to be constructed or is located and the site on those lands of that water supply work, and

Private	oint private works Private irrigation districts Private irrigation districts Division 2					
		(c)	must be accompanied by particulars of the title and area of the lands referred to in paragraph (b) (i) and of any water use approval under which those lands are, at the date of lodgment of the supplementary petition, authorised to be irrigated, and	1 2 3 4		
	(d) must contain the names, addresses and occupations of the landholders of the lands referred to in paragraph (b) (ii) and be accompanied by separate particulars of the title of those lands.					
	(5)		objection must be in writing and must state particulars of the nds of objection.	8 9		
	(6)	Mini	upplementary petition or an objection has been lodged with the ster, the Minister must, after consultation with the petitioner or etor, make a recommendation with respect to the petition or etion.	10 11 12 13		
Divis	ion (3	Private irrigation boards	14		
133	Priv	ate irr	rigation boards	15		
	(1)		each private irrigation district there is to be a board of agement.	16 17		
	(2)		board is a corporation under the corporate name assigned to it by roclamation by which its private irrigation district is constituted.	18 19		
	(3)		ard is to consist of such number of members, being not less than more than 10:	20 21		
		(a)	in the case of the first board elected for a private irrigation district, as may be determined by the Minister, and	22 23		
		(b)	in the case of any subsequent board elected for a private irrigation district, as may be determined before the election by the board for that private irrigation district.	24 25 26		
	(4)	The r	regulations may make provision for or with respect to:	27		
		(a)	the conduct of elections for the members of a board, and	28		
		(b)	other matters concerning the constitution and procedure of a board.	29 30		

Division 3			Private irrigation districts Private irrigation boards	
134	Ele	ction (of members of private irrigation boards	1
	(1)	on th	election of the members of a private irrigation board must be held ne day and at the time fixed by the proclamation by which it is tituted and thereafter:	2 3 4
		(a)	if the third anniversary of the declaration of the poll for the previous election of members of that board is a Saturday, on that Saturday, or	5 6 7
		(b)	in any other case, on the Saturday following the third anniversary of the declaration of the poll for the previous election of members of that board.	8 9 10
	(2)		orporation that is a member of a private irrigation board must orise an individual to represent it as a member of the board.	11 12
	(3)		ect to this Division, the members of a private irrigation board hold e until the date of the declaration of the poll for the next election.	13 14
135			of members of private irrigation board from office and ent of administrator	15 16
	(1)		Governor may, by proclamation in the Gazette, remove all the abers of a private irrigation board from office:	17 18
		(a)	if there are not sufficient members of the board to form a quorum, or	19 20
		(b)	if, in the opinion of the Governor, the board has failed or neglected to make or levy rates or charges required by this Part, or otherwise to exercise its functions under this Part.	21 22 23

(2) If there are insufficient members of a private irrigation board to form

(3) If the Governor has removed the members of a private irrigation board

(4) An administrator so appointed has and may exercise all of the

appoint an administrator for the board.

a quorum, the Governor, instead of removing the members from office,

may, by proclamation in the Gazette, appoint persons (each being

eligible for election) to fill the vacancies in the membership of the

board and to hold office until elections are held to fill those vacancies.

from office, the Governor may, in the proclamation removing the

members from office or in a subsequent proclamation in the Gazette,

functions of a private irrigation board and is entitled to be paid, out of

the funds of the board, such remuneration as the Governor may

24

25

26

27

28

29

30

31

32

33

34

35

36

determine.

Clause 134

Chapter 4

Water Management Bill 2000

Joint private works

Joint properties of the private of t	irriga	tion di	stricts Part 2	
	(5)	offic than	e members of a private irrigation board have been removed from the tee, the Minister may, and if the board's term of office has more 12 months to run, must, by notice in the Gazette, order that an action of members be held on a day specified in the order.	1 2 3 4
	(6)	date	day so specified must be a day not more than 6 months after the of removal of the members from office and not less than 4 weeks the date on which the notification is published in the Gazette.	5 6 7
	(7)	offic priva for the	member elected at an election held under this section is to hold be until the time when the terms of office of the members of a late irrigation board who have been removed from office would but heir removal have expired and no longer, but is eligible for retion if otherwise qualified.	8 9 10 11 12
	(8)	appo	sufficient members of a private irrigation board, by election or bintment under this section, taking office to form a quorum, the tions of the administrator cease.	13 14 15
	(9)	offic subse an el at the satist know	e members of a private irrigation board have been removed from the because of the board's failure or neglect as referred to in ection (1) (b), each of those members is ineligible for election at ection ordered under this section or, if no such election is ordered, the next election of members of the board, unless the Minister is fied that the failure or neglect took place without the member's wledge or consent and, by instrument in writing, declares that the liber is eligible for election at any such election.	16 17 18 19 20 21 22 23
136	Abo	olition	of private irrigation districts	24
		any j	Governor may, by proclamation published in the Gazette, abolish private irrigation district for which a private irrigation board has been elected.	25 26 27
137	Win	ding-	up of private irrigation boards	28
	(1)	If the	e Governor:	29
		(a)	is of the opinion that a private irrigation board has ceased to function satisfactorily, or	30 31
		(b)	is satisfied that a private irrigation board has made a request to the Minister that it be wound up,	32 33

the Governor may order that the board be wound up.

Water Management Bill 2000

34

Clause 137		Water Management Bill 2000					
Chapte	er 4	Joint private works					
Part 2		Private irrigation districts					
Division 3		Private irrigation boards					
	(2)	An order under subsection (1) must be published in the Gazette and must appoint a liquidator for the private irrigation board.	1 2				
	(3)	A winding up of a private irrigation board under this section commences on the publication of the order in the Gazette.					
(4)		The regulations may make provision for or with respect to the winding up of a private irrigation board and for the disposal of any residual assets of the board.					
	(5)	If the Governor is satisfied that the winding up of a private irrigation board has been completed under this section, the Governor may, by proclamation published in the Gazette, abolish the board.	8 9 10				
138	Em	ployees	11				
	(1)	Each private irrigation board may from time to time employ such persons as may be necessary to assist it in the exercise of its functions.	12 13				
	(2)	A person who has ceased to be a member of a private irrigation board is not eligible to be employed by the board until 6 months have elapsed after the person's so ceasing to be a member.	14 15 16				
	(3)	All employees of a private irrigation board are subject to the control and governance of the board and to the provisions of any by-laws made by the board in that behalf.	17 18 19				
		A private irrigation board may fix wages and conditions of employment of its employees if they are not fixed in accordance with the provisions of any other Act.	20 21 22				
Divis	ion 4	Construction and taking over of works	23				
139	Aut	ithorised sites					
		For the purposes of this Division, a site is an <i>authorised site</i> for a water supply work if:	25 26				
		(a) it is the site shown on the plans that accompanied: (i) the petition for the constitution of the district, or (ii) any subsequent petition for the addition of lands to the district,	27 28 29 30				
		as the site of any work proposed to be constructed as a water supply work, or is that site as varied by the Minister as a consequence of any objection to the petition, or	31 32 33				

Consti	ruction	and ta	king over of works	Division 4	
		(b)	it is the site on which the Go construction or taking over of a w		1 2
		(c)	it is the site of any works of which the control, use and maintenance u		3 4
140	Cor	struc	tion, maintenance and operation o	of water supply works	5
	(1)		ivate irrigation board may construct r supply work that is located on an a		6 7
	(2)	section	ivate irrigation board must not execute in respect of any authorised site ing works unless it has taken of tenance of those works under this D	on which are situated any over the control, use and	8 9 10 11
	(3)	irriga site s mana	the purpose of exercising its powers ation board may enter any authorised ituated on lands that belong to, or a agement of a public authority) and to ctive material.	site (not being an authorised re under the care, control or	12 13 14 15 16
141	Priv	ate irı	rigation board may apply to take o	ver water supply works	17
	(1)	take (ivate irrigation board may apply to to over any water management work that or that work.		18 19 20
	(2)	The a	application:		21
		(a)	must contain the names, address landholders of the lands on whi referred to in the application constructed or is or are located, an	ich the water supply work is or are proposed to be	22 23 24 25
		(b)	must be accompanied by: (i) a plan showing those lands of that water supply work, (ii) particulars of the title of the		26 27 28 29
	(3)	notic	Minister must consider any such age containing particulars of the applicate and in a local newspaper.		30 31 32

Private irrigation districts

Joint private works

Clause 139

Chapter 4 Part 2

Part 2 Division 4		Private irrigation districts					
DIVISIO	n 4	Construction and taking over of works					
142	Obj	ections to applications					
	(1)	Within 28 days after the publication of the relevant notice, any person may lodge with the Minister an objection in writing to the granting of the application referred to in the notice on the ground that the granting of the application would adversely affect the person's interests.					
	(2)	A person may not make an objection on the ground that the person's interests would, if the petition were granted, be adversely affected for reasons relating to the quantity of water available from the river, estuary or lake from which it is proposed to take water by means of the proposed water supply work, and any objection made on that ground is not to be entertained.					
	(3)	An objection must state particulars of the grounds of objection.					
	(4)	If an objection is lodged with the Minister, the Minister must, after consultation with the objector, make a recommendation with respect to the objection.					
143	Governor may authorise construction or taking over of water supply work						
		If an objection to an application is not duly lodged, or is duly lodged but the Minister recommends that the application be granted, the Governor may, by proclamation in the Gazette, authorise the construction or taking over of the water supply work on the site concerned.					
144	Tak	ing over works					
	(1)	A private irrigation board may serve a notice in writing on the landholder of any lands on which is situated any water supply work on an authorised site informing the landholder that the board proposes to take over the work.					
	(2)	The control and management of the work specified in the notice vests in the private irrigation board on and from the day specified in the notice.					
	(3)	A private irrigation board must not serve such a notice after the expiration of 12 months after the constitution of the private irrigation district.					
	(4)	A private irrigation board must not serve such a notice on any person in respect of a work that belongs to, or is under the control or management of, a public authority.					

35 36

Clause 142

Chapter 4

Water Management Bill 2000

Joint private works

		ition dis	Part 2 Division 4			
145	Tra	nsfer (of lands		1	
	(1)	mont	ivate irrigation board, by notice in v hs after the constitution of the private holder of any lands:		2 3 4	
		(a)	that, immediately before the constitution district, were vested in the landhole supplied with water under a single was and	der as a trustee of lands	5 6 7 8	
		(b)	that on the constitution of the private the whole or part of the private irriga		9 10	
		•	require the landholder to transfer the land n such period as may be specified in the		11 12	
	(2)	land,	otice is served on a landholder under the the land is taken to be held by the land te irrigation board by which the notice	holder as a trustee for the	13 14 15	
146	Power of entry					
	(1)	A private irrigation board may, by its employees or agents, at any reasonable time enter any lands and thereon carry out any investigation or inspection, take levels, drill test bore-holes, make surveys and marks and fix pegs or stakes for the purpose of determining the site of any proposed water supply work.				
	(2)					
Divis	ion (5	Compensation		25	
147	No	comp	ensation for surveys on land within p	private irrigation district	26	
	(1)	A pe	rson is not entitled to compensation by	reason of:	27	
		(a)	any water supply works becoming virrigation board has the control, use		28 29	
		(b)	the exercise by a private irrigation be of entry conferred on it by this Part private irrigation district of that boar	on any lands within the	30 31 32	

Joint private works

Clause 145 Chapter 4

Clause 147 Chapter 4 Part 2 Division 5		J. P	Vater Management Bill 2000 oint private works Private irrigation districts Compensation				
	(2)		ection (1) (a) has effect subject to any terms and conditions odied in a decision of the Land and Environment Court.	1			
148	Cor	npens	sation for surveys on lands outside private irrigation district	3			
		susta	pensation is payable by a private irrigation board for all damage ined by any person as a result of a board's exercise of its power rry out surveys on lands outside its private irrigation district.	5 6			
149	Cor wor		sation where private irrigation board constructs or takes over	7			
	(1)	a priv	pensation assessed in accordance with this Division is payable if wate irrigation board exercises its powers to construct or take over water supply works.	9 10 11			
	(2)	the rewher	pensation is not payable by a private irrigation board in respect of epair, operation or maintenance of any water supply work except e, in repairing, operating or maintaining any such work or works, board causes damage to any lands outside its private irrigation ct.	12 13 14 15			
	(3)	If immediately before the control and management of any water supply work becomes vested in a private irrigation board under this Part there was in force a legally binding agreement or arrangement between the person who then had the control and management of the work and some other landholder of land in the board's private irrigation district (being an agreement or arrangement under which that other person was entitled to exercise any powers in relation to that work):		17 18 19 20 21 22 23			
		(a)	that agreement or arrangement is taken to be an agreement or arrangement between that board and that other person, and	24 25			
		(b)	any compensation to which that other person may be entitled under this Division must be assessed, having regard to his or her obligations under that agreement or arrangement.	26 27 28			
150	Det	Determination of amount of compensation					
			ompensation is payable under this Division, the amount of pensation must be determined:	30			
		(a)	by agreement between the private irrigation board and the person entitled to claim compensation, or	32			

Joint private works Private irrigation districts Compensation			•	
		(b)	if such an agreement has not been reached, by the Land and Environment Court in accordance with the provisions of this Division.	1 2 3
151	Recovery of compensation		4	
		this 1	amount payable to a claimant as compensation in accordance with Division may be recovered from the private irrigation board as a in any court of competent jurisdiction.	5 6 7
Division 6			Rates and charges for water	8
152	Fixing of rates and charges			9
	(1)	As soon as practicable after 1 July in each year, a private irrigation board:		10 11
		(a)	must fix a rate per hectare, for all holdings within its private irrigation district, so as to produce a total amount sufficient to meet the estimated liabilities of that board during that year and any outstanding liabilities of that board, and	12 13 14 15
		(b)	must fix a rate per hectare for water, or a charge for the quantity of water, to be supplied during that year for domestic and stock purposes to all holdings within the private irrigation district, and	16 17 18 19
		(c)	 in the case of a board constituted for a private water supply and irrigation district: (i) must determine the total quantity of water that it proposes to supply to all holdings for the purpose of irrigation during that year, and (ii) must fix the charges in respect of the quantities of water allocated under Division 7 for that purpose in respect of all holdings within the district. 	20 21 22 23 24 25 26 27
	(2)	fixed	rates and charges referred to in subsection (1) (b) and (c) must be d so as to produce in the year for which they are fixed the amount nated by the private irrigation board as being required in that year:	28 29 30
		(a)	to defray the cost of constructing, maintaining and operating its water supply works, and	31 32
		(b)	to pay the interest on and repay the capital of any loans raised by the board, and	33 34

			Vater Management Bill 2000		
Chapte	er 4		oint private works		
Part 2 Divisio			Private irrigation districts Rates and charges for water		
		(c)	to meet any outstanding liabilities of the board and the costs and expenses of administering the private irrigation district and of doing all such things as the board may lawfully do.	1 2 3	
153	Spe	cial ra	ates and special charges	4	
	(1)	For the	he purpose of raising money:	5	
		(a)	for the fulfilment of any contract, or	6	
		(b)	for the payment of any debt that may be due or become due by the board, or	7 8	
		(c)	for any other purpose for which the board is authorised to exercise its functions,	9 10	
			vate irrigation board may from time to time fix a special rate per ure to be paid in respect of all holdings in its private irrigation ct.	11 12 13	
	(2)	A pri	vate irrigation board may also fix special charges in respect of:	14	
		(a)	the quantities of water allocated for the purpose of irrigation, or	15	
		(b)	the quantities of water determined for domestic and stock purposes,	16 17	
		in res	spect of all holdings in its private irrigation district.	18	
154	Ass	19			
	(1)	Rates fixed by a private irrigation board must be assessed, and must be levied as prescribed, in respect of the area of each holding within its private irrigation district.			
	(2)	and	ges fixed by a private irrigation board for a private water supply irrigation district must be assessed, and must be levied as cribed, in respect of:	23 24 25	
		(a)	the quantity of water allocated for irrigation by the board under Division 7, or	26 27	
		(b)	the quantity of water determined by the board for domestic and stock purposes,	28 29	
		in res	spect of each holding within the district.	30	
155	Ass	essm	ent of rates and charges	31	
	(1)		oon as practicable after fixing any rates and charges, a private	32 33	

Rates	and c	harges for water	Division 6				
	(2)	assessment of any rates or charge	time finds it has made an error in the s for water in respect of any holding assess the rates or charges in respect cted and if it does so:	1 2 3 4			
		(a) must refund any amounts	overpaid, and	5			
		(b) may levy any additional a	mount found to be due.	6			
156	Pro	vision for determining areas of h	noldings	7			
		Division, a holding that has an are	essing and levying rates under this ea equal to a number of hectares and is taken to have an area in hectares ber.	8 9 10 11			
157	Liability for rates and charges for water						
	(1)	charges are be payable to the priva	er this Division in respect of rates or ate irrigation board by the landholder in they were levied, and must be paid by the landholder.	13 14 15 16			
	(2)		yable to and recoverable by a private of one month after service of notice	17 18 19			
	(3)	may be levied on any one or m irrigation board may recover the whom they are so levied, but no	ore of the land, the rates or charges ore of those persons, and a private rates or charges from any person on othing in this subsection entitles the all amount of the rates or charges.	20 21 22 23 24			
	(4)	jointly and severally liable to the or charges, but as between them	two or more landholders, they are private irrigation board for the rates selves they are each liable only for s is proportionate to their interests in	25 26 27 28 29			
	(5)		e private irrigation board more than landholder may recover the excess	30 31 32			

Private irrigation districts

Joint private works

Clause 155

Chapter 4 Part 2

Chapter 4 Part 2 Division 6		Р	oint private works rivate irrigation districts ates and charges for water			
	(6)	If any land in a holding reverts to the Crown during any year for which rates or charges are or are to be assessed, the person who immediately before the reversion was the landholder of the land is liable for payment of only that part of the rates or charges proportionate to the part of the year for which the land was held by the person, and any excess payment by the person must be refunded to the person.		1 2 3 4 5		
158	Cha	rge or	n land	7		
	(1)	privat any s	s and charges under this Division, and any costs awarded to a te irrigation board by any court in proceedings for the recovery of such rates and charges, are a charge on the land in respect of the the rates and charges have been levied.	8 9 10 11		
	(2)	in go	arge created by subsection (1) has no effect as against a purchaser od faith for value who at the time of purchase made due inquiry ad no notice of the liability.	12 13 14		
	(3)	made priva	the purposes of subsection (2), a purchaser is not taken to have a due inquiry unless the purchaser obtained a certificate from the te irrigation board as to the amount, if any, due in respect of rates, tes or costs.	15 16 17 18		
159	Aba	ndonr	ment of rates	19		
		the b certification off is	and charges may be abandoned or written off in accordance with by-laws made by the private irrigation board but only on the licate of the auditor of the board that the abandonment or writing in accordance with the by-laws and on the unanimous resolution be board.	20 21 22 23 24		
160	Assessment book					
	(1)	(in th	private irrigation board must cause to be kept a book or record his Part referred to as the <i>assessment book</i>) in which must be ded the following:	26 27 28		
		(a)	the name, address and occupation of each landholder of land within the private irrigation district,	29 30		
		(b)	particulars of the area of each parcel of land owned by each such landholder,	31 32		
		(c)	particulars identifying the separate holdings within the private irrigation district,	33 34		
		(d)	particulars of the total area of land owned by all landholders,	35		

Water Management Bill 2000

Rates	and c	harges	for water Division 6	
		(e)	 in the case of a board for a private water supply and irrigation district: (i) particulars of the quantity of water last allocated by the board to each irrigated holding within the district, and (ii) particulars of the quantity of water last determined by the board for domestic and stock purposes in respect of 	
		(f)	each such holding, such particulars relating to the fixing, assessing and levying of rates and charges as may be prescribed.	
	(2)		ivate irrigation board may from time to time rectify any errors or sions in the assessment book.	
Division 7 Supply of water		7	Supply of water	
161	Det	ermin	ation of allocation	
		deter	pard for a private water supply and irrigation district must mine the quantity of water, if any, to be allocated for irrigation to holding within the private irrigation district.	
162	Sup	ply of	water states	
		At th	e times fixed by it, a private irrigation board:	
		(a)	must supply water for domestic and stock purposes (in such quantities as it may determine): (i) to the boundary of each holding within its private irrigation district, and (ii) to such other points as may be agreed on by the board and the landholder of the holding, and	
		(b)	 in the case of a board for a private water supply and irrigation district, must supply water for irrigation (in the quantities allocated by it under this Division): (i) to the boundary of each holding within its private irrigation district for which an allocation of water for irrigation has been made by the board, and (ii) to such other points as may be agreed on by the board and the landholder of the holding. 	

Private irrigation districts

Joint private works

Clause 160

Chapter 4 Part 2

Division 7		Supply of water		
163	Dis	Discontinuance or reduction of supply of water		
	(1)	A private irrigation board may at any time refuse to deliver water to any holding or may discontinue any delivery of water to a holding:		
		 (a) in the case of a holding for which it has made an allocation of water for irrigation: (i) if the land to be irrigated is not in its opinion properly prepared for irrigation or on which the ditches or channels to be used for the distribution of water within the holding are in the opinion of the board inadequate or in a bad state of repair, or 	4 5 6 7 8 9	
		(ii) if the water is for the irrigation of grasses or pastures that are not sown grasses or improved pastures, or	11 12	
		(b) if any rates or charges for water in respect of the holding are, and have been for a period of 2 months or more after the due date of payment, unpaid, or	13 14 15	
		(c) if the landholder of the holding does not comply with any requirement specified in a notice given to the landholder under this Division, or	16 17 18	
		(d) if the water is being used for a purpose that is not authorised by a water use approval.	19 20	
164	Circ wat	cumstances in which private irrigation board not obliged to supply ter	21 22	
	(1)	Nothing in this Part requires a private irrigation board to supply water to any land or landholder if, by reason of drought, accident or otherwise, the board is of the opinion that it is impracticable to do so.	23 24 25	
	(2)	Unless the private irrigation board otherwise determines, any failure to deliver water to a holding does not relieve the landholder of the holding of any liability for payment of rates and charges, and rates and charges continue to be leviable in respect of the holding despite any such failure.	26 27 28 29 30	
165	Lan	ndholders may be required to provide distribution works.	31	
		A private irrigation board may, by notice in writing given to the landholder of any holding, require the landholder:	32 33	
		(a) to provide on his or her holding:	34	

Chapter 4

Part 2

Water Management Bill 2000

Private irrigation districts

Joint private works

Joint p	orivate e irriga	works	t Bill 2000 Clause 165 Chapter 4 Part 2 Division 7	
		(b)	 (i) ditches or channels of such a size and capacity as will enable water to be delivered to his or her land at not less than such rate of delivery as the board may stipulate in the notice, and (ii) water storage works, in such locations and of such nature and extent as the board may stipulate in the notice, for the water to be supplied by it for domestic and stock purposes, and to maintain, continuously and effectively, any ditches or channels referred to in paragraph (a) (i) so that water may be carried or passed at not less than the rate so stipulated. 	1 2 3 4 5 6 7 8 9 10
166	Sale	e of su	irplus water	12
	(2)	suppli a priv supply landh irrigat agreed If the additi licence autho irrigat	full quantity of water to be supplied under this Division has been ied or, despite any sales made under this section, will be supplied, vate irrigation board may, subject to the conditions of any water y work approval held by it, agree to sell by measure to the colder of any holding water from the works of the private tion district subject to such terms and conditions as may be d to by the board and that landholder. Minister authorises a private irrigation board to take water in ion to the water the board is entitled to take under its access ce, the board may agree to sell by measure the water it is so orised to take to the landholder of any holding within its private tion district, subject to such terms and conditions as may be d on by the board and that landholder.	13 14 15 16 17 18 19 20 21 22 23 24 25
Divis	sion 8	3	Effect of new subdivisions	26
167	Supply of water for domestic and stock purposes to new holdings resulting from subdivisions			27 28
	(1)	subdito a si irrigat board	holding, whether an irrigated or non-irrigated holding, is vided, a new holding resulting from the subdivision is not entitled supply of water for domestic and stock purposes from a private tion board's water supply works until a date determined by the I (not being a date earlier than the date on which the board me aware of the disposition of that new holding).	29 30 31 32 33 34

capable of being irrigated from the works of the private

irrigation district or to which, in the opinion of the board, it is

impracticable to convey water for irrigation from those works,

a holding nominated by the landholder is too small to justify an

the allocation of water in the manner nominated is otherwise

detrimental to the administration of the private irrigation district

27

28

29 30

31

32

33

34

35

(c)

(d)

allocation of water, or

concerned.

Water Management Bill 2000	Clause 168
Joint private works	Chapter 4
Private irrigation districts	Part 2
Effect of new subdivisions	Division 8

- (4) If a private irrigation board makes an allocation under subsection (3), it is to notify the landholder of the subdivided holding and the landholder of each new holding of the allocation.
- (5) An allocation of water to a new holding made under this section has effect on and from a date to be specified in the instrument by which the allocation is made (not being a date earlier than the date on which the private irrigation board became aware of the first disposition of any of the new holdings resulting from the subdivision).
- (6) The landholder of a new holding resulting from a subdivision referred to in subsection (1) is not entitled to an allocation of water by the private irrigation board for irrigation purposes otherwise than in accordance with this section.
- (7) If the charges for water have not been levied for the current year for the subdivided holding, the private irrigation board must levy the charges for the whole of that year in respect of the new holding to which the water previously allocated to the subdivided holding has been allocated in accordance with this section.

169 Additional works required as a result of subdivision

- (1) The person who, immediately before the disposition of a new holding resulting from a subdivision, was the landholder of the holding (the *previous landholder*) must construct at his or her own cost such works as are necessary to provide:
 - (a) means of conveying water to the new holding from the private irrigation board's water supply works and, if an allocation of water is made to the new holding for irrigation, means of measuring the water so supplied, and
 - (b) means of access from roads to any works of the private irrigation district or any works provided for the purposes of paragraph (a) if that access would not be available except by crossing a channel of the private irrigation district, and
 - (c) means of access across a channel of the private irrigation district to the new holding if that means of access is required by reason of the subdivision.

Division 8	Effect of new subdivisions	
(2)	All works to be constructed under subsection (1):]
	(a) in respect of the supply of water to a new non-irrigated holding, must be constructed before the new holding is disposed of or within such period after the disposition of the new holding as the private irrigation board may in any particular case allow, and	
	(b) in respect of the supply of water to a new irrigated holding, must be constructed within such period as the private irrigation board may, by notice in writing, have notified to the landholder of the holding that was subdivided.	? { 9
(3)	All works constructed or to be constructed under subsection (1) must be constructed in accordance with the approval in writing of the private irrigation board in respect of location, design, form, dimensions and construction.	11 12 13 14
(4)	At the request of the previous landholder, a private irrigation board may undertake, at the landholder's cost, the construction of any works required by this section.	15 16 17
(5)	A private irrigation board may construct such works as have not been constructed by the previous landholder, and any costs and expenses (including any compensation paid or payable by the board under Division 5 by reason of the construction of the works) are payable to the board either by the previous landholder or by the new landholder, as the board may determine.	18 19 20 21 22 23
(6)	If any part of the costs and expenses referred to in subsection (5) is recovered by the private irrigation board from the new landholder, the new landholder may recover from the previous landholder the whole or that part of those costs or expenses, as the case may be.	24 25 26 27
(7)	On their completion, the control and management of any works constructed under this section is vested in the private irrigation board.	28 29
Division	9 Meetings of landholders	30
170 Ge	neral meeting	31
(1)	_	32 33 34

Chapter 4

Part 2

Water Management Bill 2000

Private irrigation districts

Joint private works

	e irriga	works Chapter 4 Ition districts Part 2 Ilandholders Division 9	
	(2)	A private irrigation board must, within 21 days after the receipt of a requisition signed by not less than one-fifth in number of the landholders of the holdings in the private irrigation district, convene a general meeting of those landholders.	1 2 3 4
	(3)	Seven days' notice of every general meeting must be sent to every landholder at the address shown in the private irrigation board's assessment book informing the landholder of the time and place of the general meeting.	5 6 7 8
171	Vot	ing rights	9
	(1)	A corporation may, by any person authorised by it in writing, attend general meetings and vote.	10 11
	(2)	If there is more than one landholder of any holding, each landholder may attend general meetings but only one of them may vote.	12 13
Divis	ion '	10 Finance	14
172	Boo	oks of account	15
		Each private irrigation board must cause to be kept, in relation to its funds, proper books of account that must be audited as often as the board considers it advisable so to do, but at least once in every year, by a registered company auditor (within the meaning of the <i>Corporations Law</i>).	16 17 18 19 20
173	Acc	counts to be rendered	21
		Each private irrigation board must as soon as practicable, and in any case before 31 October in each year, forward to the Minister a copy of the income and expenditure account, balance sheet and rate account as last audited, together with a copy of the certificate of audit relating to them.	22 23 24 25 26
174	Bar	king of money	27
	(1)	All money received by or on account of a private irrigation board must be paid into a bank or authorised deposit-taking institution chosen by the board.	28 29 30

Clause 170

Clause 174			Vater Management Bill 2000		
Chapte	er 4	·			
Part 2 Division 10		Private irrigation districts			
DIVISIO	n 10		inance		
	(2)		y payment of \$2.00 or more by or on behalf of a private irrigation d must be by cheque on the bank or authorised deposit-taking	1 2	
			aution drawn and countersigned as prescribed by the regulations.	3	
	(3)		nents of less than \$2.00 may be made out of a petty cash fund, enished from time to time by cheque drawn and countersigned as	4	
		-	cribed by the regulations.	5	
175	Ten	npora	ry accommodation	7	
	(1)		he temporary accommodation of a private irrigation board it may	8	
			in advances by way of overdraft of current account in any one or	9	
			e banks or authorised deposit-taking institutions on the security of noome of the board.	10	
				11	
	(2)	The	amount of any such overdraft must be limited to:	12	
		(a)	one-half of the income of the private irrigation board as shown by the last audited accounts, or	13 14	
		(b)	if there are no audited accounts, one-half of the income of the	15	
		(-)	private irrigation board estimated by the board in respect of the	16	
			year commencing on 1 July immediately preceding the date on	17	
			which the overdraft is proposed to be obtained.	18	
	(3)	No g	greater amount may be borrowed under this section than the	19	
			unt stated in a certificate of the auditor of the private irrigation	20	
			d as being the sum that may be borrowed within the limits	21	
		ımpo	osed by this section.	22	
Divis	ion '	11	Miscellaneous	23	
176	Del	egatio	on	24	
•		_	ivate irrigation board may by instrument in writing delegate:	25	
		(a)	to any member of the board, or		
		` ′	•	26	
		(b)	to any employee of the board,	27	
		any o	of its functions, other than this power of delegation.	28	

Water Management Bill 2000	Clause 177
Joint private works	Chapter 4
Private irrigation districts	Part 2
Miscellaneous	Division 11

177	Ass	essm	ent book admissible as evidence	1
		In an	ny legal proceedings for the recovery of rates or charges:	2
		(a)	a private irrigation board's assessment book, or	3
		(b)	any document purporting to contain a copy of any part of a private irrigation board's assessment book that is certified as a true copy by the chairperson of the private irrigation board, or by a person authorised by the chairperson in that regard,	4 5 6 7
		is ad	missible in evidence.	8
178	Ser	vice o	of notice of proceedings on a private irrigation board	9
			document required to be served on a private irrigation board may erved:	10 11
		(a)	by leaving it with some person apparently employed by the board at the office of the board, or	12 13
		(b)	by posting it to the board at its office.	14
179	Rec	overy	of rates	15
		this	rates, charges or money due to a private irrigation board under Part may be recovered as a debt in a court of competent diction.	16 17 18
180	Am	endm	ent of proclamations	19
		proc	proclamation under this Part may be amended by a further lamation for the purpose of correcting any error in the earlier lamation.	20 21 22
181	By-	laws		23
	mal		rivate irrigation board may, with the approval of the Governor, e by-laws not inconsistent with this Part or the regulations for or respect to the following:	24 25 26
		(a)	the convening and holding of general meetings of landholders,	27
		(b)	the appointment by a landholder of a proxy for voting purposes,	28
		(c)	the supply of water for domestic and stock purposes or for irrigation.	29 30

24

0.5 penalty units.

Clause 181

Water Management Bill 2000

Prelim	inary Division 1	
Part	3 Private drainage boards	1
Divis	ion 1 Preliminary	2
182	Application of Part	3
	This Part applies to all drainage unions under the former <i>Drainage Act</i> 1939 that were in existence immediately before the repeal of that Act, but does not authorise the establishment of any new drainage unions.	4 5 6
183	Definitions	7
	In this Part:	8
	director means director of a private drainage board.	9
	drain includes a natural watercourse.	10
	<i>drainage district</i> means the area in respect of which a drainage union is constituted, having boundaries as varied from time to time in accordance with this Part.	11 12 13
	private drainage board means board of directors of a drainage union.	14
	<i>ratable land</i> means land in respect of which rates or charges are payable under this Part.	15 16
	ratable person means the owner of ratable land.	17
184	Requirements for approvals	18
	Nothing in this Part authorises a private drainage board to construct or	19
	use a drainage work otherwise than in accordance with a drainage work approval held by the board.	20 21
Divis	ion 2 Private drainage boards	22
185	Private drainage boards	23
	Each private drainage board is a corporation under the corporate name by which it was constituted.	24 25

Joint private works

Private drainage boards

Clause 182

Chapter 4 Part 3

Clause 186	Water Management Bill 2000
Chapter 4	Joint private works
Part 3	Private drainage boards
Division 2	Private drainage boards

186	App	oointn	nent and election of directors	1		
	(1)		ivate drainage board is to have not less than 3, and not more than rectors.	2 3		
	(2)		y Crown lands are included in the private drainage board's district, of the directors is to be appointed by the Minister (the <i>appointed</i> etor).	4 5 6		
	(3)		remaining directors (the <i>elected directors</i>) are to be elected by holders within the drainage district.	7 8		
	(4)	The	regulations may make provision for or with respect to:	9		
		(a)	the conduct of elections for the elected members of a private drainage board, and	10 11		
		(b)	other matters concerning the constitution and procedure of a private drainage board.	12 13		
187	Functions of private drainage boards					
	(1)	A private drainage board has the following functions:				
		(a)	to prepare, review and implement a management program for its drainage district,	16 17		
		(b)	to maintain in a state of efficiency the drainage works under its charge, and renew such drainage works if necessary,	18 19		
		(c)	to construct, alter, or extend any drainage works in accordance with any authority and consent given under this Part,	20 21		
		(d)	to make, levy and collect rates,	22		
		(e)	to appoint such officers and employees as may be required,	23		
		(f)	to institute legal proceedings for the recovery of outstanding rates or other amounts,	24 25		
		(g)	to keep the prescribed books and accounts,	26		
		(h)	to manage the affairs of the drainage union,	27		
		(i)	to do such acts as may be necessary or desirable for carrying out the purposes of this Part.	28 29		
	(2)	the I	drainage works affecting navigable waters (within the meaning of <i>Protection of the Environment Operations Act 1997</i>) are to be menced without the approval of the Governor.	30 31 32		

Private	te drainage boards Division 2			
188	Ent	ry on	lands	
	(1)		ivate drainage board or any person authorised by it may enter any within or outside its drainage district:	
		(a)	for the purpose of making inspections or surveys, and	
		(b)	for the purpose of constructing, maintaining and effecting extensions and alterations to the drainage works, and	
		(c)	for any other purpose in the exercise of its powers and duties.	
	(2)		private drainage board must make full compensation for damage	
		this l	sioned to any land in the exercise of its powers and duties under Part.	,
189	Extension of drainage works			
		A pr	ivate drainage board is to carry out such extensions of drainage	
			ks as are authorised by a majority of votes cast at a general meeting	1
		at wl	hich a quorum is present.	1
190	Am	endm	ent of boundaries	1
	(1)		boundaries of a drainage district may from time to time, on ication by the private drainage board, be amended by the	1

(2) Any such application must contain particulars of the lands proposed to

(3) The Minister must cause notice of any such application to be published

(5) After expiration of the appointed time and on consideration of any

objection lodged the Minister may recommend and the Governor may

approve an application with such modification, if any, as the Minister,

from the drainage district, and

be added to or excised from the drainage district, and that have or are

capable of being increased in value or that have decreased in value

respectively by reason of the operations of the private drainage board.

give particulars of the lands proposed to be added to or excised

appoint a time (not being earlier than 8 weeks after the date of

publication of the notice) and place at which objections may be

Water Management Bill 2000

Governor.

in a local newspaper.

lodged.

on investigation, may recommend.

(4) The notice must:

(a)

(b)

Joint private works
Private drainage boards

18

19

20

21

22

23

24

25

26

27

28

29

30

31

32

33

34

Clause 188 Chapter 4

Part 3

Clause 190 Chapter 4 Part 3 Division 2		Water Management Bill 2000 Joint private works Private drainage boards Private drainage boards		
	(6)	The approval of the Governor, and particulars of the alterar in the boundaries of the drainage district, must be notif Minister in the Gazette.		
	(7) As from the date of publication of such notification, the boundaries of the drainage district are taken to be altered accordingly.		undaries of	
Divis	sion (3 Rating		
191	App	pointment of valuer		
	(1)	Each private drainage board is to appoint a valuer to make in accordance with this Part.	valuations	
	(2)	A valuer appointed under this Part is to determine the incre accrued, accruing or to accrue to each portion of land drainage district by reason of drainage works used or prop used by the private drainage board.	within the 1	
	(3)	Division 4 applies to valuations made under this Part.	14	
	(4)	A valuation made for the purposes of this Part comes into January following the date on which it is made.	effect on 1	
192	Anr	nual estimates	1	
		For each year commencing 1 January, the private drainage to cause an estimate to be made of the amount required for the following purposes:		
		(a) construction, maintenance, extension and alteration works,	of drainage 2	
		(b) payment to the Treasurer of any amounts due or become	oming due, 2	
		(c) repayment of loans, and payment of interest thereon	n, 24	
		(d) defraying costs of administration,	2.	
		(e) meeting all other expenditure whatsoever in carry powers, authorities, duties and functions,	ing out its 2	
		(f) the preparation and implementation of management	programs. 2	

Water Management Bill 2000	Clause 193
Joint private works	Chapter 4
Private drainage boards	Part 3
Rating	Division 3

193 Rates, making and levying 1 (1) For each year commencing 1 January, the private drainage board is to 2 make and levy rates on all ratable land in the drainage district in 3 respect of the increased values as shown in the current valuation book 4 sufficient to meet the estimated amount required for that year. 5 (2) Rates so made and levied must not, except with the consent of the 6 Minister previously obtained, exceed 10% on the aggregate of the 7 increased value of all land within the drainage district, or the 8 maximum amount levied in any year before the commencement of this 9 Part, whichever is the greater. 10 (3) Every such rate must: 11 be made by resolution of the private drainage board, and 12 (b) be levied by the service of a rate notice. 13 (4) Every rate must be levied in respect of a separate parcel of land, but 14 any rate in respect thereof may be included in the same rate notice with 15 any rate in respect of the same or different land. 16 (5) If the name of any owner liable to pay the rate is not known to the 17 private drainage board, it is sufficient to rate such owner by the 18 designation of "owner" without stating the owner's name. 19 (6) If land becomes ratable, the rate on the land is to be proportionate to 20 the portion of the year during which the land is ratable and, in that 21 case, the first valuation of the land made after it becomes ratable is 22 taken to have come into force when the land became ratable. 23 (7) If ratable land has not been valued, the first valuation of the land made 24 after discovery of the omission is taken to have come into force on 1 25 January of the current year, and the rate may be levied accordingly. 26 (8) Rates may be levied in respect of all ratable land within the drainage 27 district. 28 (9) All land within a drainage district in respect of which any increased 29 value has been determined under this Part is ratable, except land 30 owned by the Crown (not being land held under a lease from the 31 Crown by any person for private purposes) that is situated within the 32

drainage district.

33

Clause 194		Water Management Bill 2000				
Chapter 4		Joint private works				
Part 3		Private drainage boards				
Divisio	n 3	Rating				
194	Irre	gularities	1			
		If for any reason any rate is not made within or by the time prescribed	2			
		by or under this Part, or if any irregularity in making or levying any	3			
		rate affects or may be considered to affect the validity of any rate, the	4			
		Governor may extend the time for the making of the rate, and may	5			
		authorise the doing by the private drainage board of such acts as may be necessary to cure the irregularity and to validate the rate.	6 7			
195	Rat	e book	8			
	(1)	Every rate must be entered in a rate book which must be in the prescribed form.	9 10			
	(2)	An alteration or amendment in the rate book may be made in respect of any rate by altering such of the particulars entered therein as may be prescribed.	11 12 13			
	(3)	An alteration or amendment in the rate-book has effect on adoption by the private drainage board as though made when the rate was made.	14 15			
	(4)	An alteration or amendment in the rate-book made in conformity with a resolution of the private drainage board must be made in the prescribed manner.	16 17 18			
196	Due date		19			
		Every rate becomes due and payable to and recoverable by the private	20			
		drainage board on the expiration of one month after service of the rate	21			
		notice.	22			
197	Lial	pility of owner	23			
		Except where this Part otherwise expressly provides, every rate is to be	24			
		paid to the private drainage board by the owner of the land in respect	25			
		of which the rate is levied.	26			
198	Lial	pility of current owner	27			
	(1)		28			
		amount owing by way of rates from the owner of any land, the private	29			
		drainage board may serve on any tenant of that land a notice requiring	30			

that any rent then due or thereafter to become due by the tenant in

respect of the land be paid by the tenant as it falls due to the board in

liquidation of the amount owing.

31

32

33

Private	drain				
Rating			Division 3		
	(2)	from	efault of payment of rent, the private drainage board may recover the tenant of the land the amount owing as a debt in any court of petent jurisdiction.	1 2 3	
	(3)	cons	payment to the private drainage board under this section titutes a valid discharge to the payer for such rent as against all r persons.	4 5 6	
	(4)	beha	ning in this section applies to a person who is a tenant for or on lf the Crown, as an officer or employee of the Crown or as an loyee of a local council.	7 8 9	
199	Les	sees	of land owned by the Crown	10	
	(1)	lease	e land is owned by the Crown and is held by any person under a from the Crown, the rate must be paid to the private drainage d by the holder of the lease.	11 12 13	
	(2)	perso	If the land is held under a lease from the Crown by two or more persons successively in the same year, the private drainage board may do any of the following:		
		(a)	it may make such adjustment (if any) of the rate, whether paid or unpaid, as it thinks proper between such persons,	17 18	
		(b)	it may recover from each of such persons that person's proportion of the rate as fixed by the adjustment,	19 20	
		(c)	it may make any refund in accordance with the adjustment,	21	
		(d)	it may write off any amount in respect of the interval between the those persons' holdings.	22 23	
Divis	ion 4	4	Valuations	24	
200	Valuation periods		periods	25	
	(1) A val made		luation in respect of all lands within the drainage district must be e:	26 27	
		(a)	at least once in every 3 years, and	28	
		(b)	for a valuation period, that is to say, for a period not exceeding 3 years.	29 30	

Joint private works

Clause 198

Chapter 4

Chapter 4 Part 3		Joint private works Private drainage boards				
Divisio	n 4	Valuations				
	(2)	Despite any other provision in this Part the private drainage board may cause any valuation to be made at any time whether during the currency of any valuation period or otherwise.	1 2 3			
	(3)	A valuation (together with any amendments made in accordance with law) remains in force until a fresh valuation comes into force.	5			
201	Incl	usive valuations	6			
	(1)	If several parcels of land adjoin, are owned by the same person, are of the same class of tenure, and no part is let to any person, they must be included in one valuation, unless the private drainage board otherwise directs.	7 8 9 10			
	(2)	Any such parcels of land must be valued separately if any buildings erected on them are obviously adapted to separate occupation.	11 12			
	(3)	If several parcels of land adjoin, are owned by the same person, are of the same class of tenure, and are all let to one person, they must be included in one valuation, unless the private drainage board otherwise directs.	13 14 15 16			
202	Sep	parate valuations	17			
	(1)	If several parcels of land owned by the same person are not of the same class of tenure, or are separately let to different persons, they must be separately valued.	18 19 20			
	(2)	Lands that are separately owned, or that do not adjoin, must be separately valued, except that lands that are separated by a road generally used by the public may be included in one valuation if they are owned by the same person and are worked as one holding.	21 22 23 24			
	(3)	If a part of a parcel of land that has been valued is sold, conveyed or resumed, fresh valuations must be made of the portion sold, conveyed or resumed and of the part remaining.	25 26 27			
	(4)	If a part only of a parcel of land is ratable, the part that is ratable must be separately valued.	28 29			
203	Lan	Land extending outside the area				
		If any land in respect of which one valuation would otherwise be made under this Division is not wholly within the drainage district but extends both inside and outside the drainage district, the part within the drainage district must be separately valued.	31 32 33 34			

Water Management Bill 2000

Valuat		nage bo	oards Part 3 Division 4		
204	Valuation book				
	(1)		ry valuation made for a private drainage board must be entered into board's valuation book.	:	
	(2)	The	valuation book must be kept as prescribed by the regulations.	2	
	(3)	•	particulars in the valuation book may be entered in the board's book.		
205	Alterations in valuation book				
	(1)		rations may be made in the valuation book by resolution of the ate drainage board at any time if:	9	
		(a)	it is found there has been an error as to the area, ownership or occupation of land, or	1	
		(b)	the valuer certifies in writing that the valuer has made an arithmetical mistake in calculation of values, or	12 13	
		(c)	there has been a change of owners and it is necessary in consequence to alter the names and other particulars forming part of the entry of the valuation.	14 1: 10	
	(2)	of th	y land has been subdivided, and a portion sold or let, the valuation ne land and any unpaid rates may be apportioned accordingly by private drainage board on the recommendation of the valuer.	17 18 19	
	(3)		rations and apportionments under this section are, for the purposes otice and objection, taken to be valuations.	20 21	

206 Notice of valuation

Water Management Bill 2000

Joint private works

(1) Notice of valuation must be given by service of a valuation notice.

(4) Other minor errors in the valuation book not affecting the amount of

(5) An alteration in the valuation book made in conformity with this

Division must be initialled by the chairperson of the private drainage

board and, if a resolution of the private drainage board is necessary to

authorise the alteration, a reference to the minute of the resolution must

(2) The notice must comply with the following requirements:

the valuation may be altered as prescribed.

be inserted in the valuation book.

(a) it must be in or to the effect of the form prescribed by the regulations,

22

23

24

25

26

27

28

29

30

31

32

33

Clause 204

Chapter 4

Chapter 4 Part 3 Division 4		Р	oint private works Private drainage boards /aluations	
		•		
		(b)	it may include any number of separate valuations provided that each valuation must be separately shown,	1 2
		(c)	it may be included in a rate notice,	3
		(d)	it may designate any ratable person, whose name is unknown to the private drainage board, as "the owner" without stating the person's name,	4 5 6
		(e)	it must state a time (being at least 30 days after service) within which any person ratable in respect of the land may lodge with the private drainage board a written objection to the valuation,	7 8 9
		(f)	it may be served in any manner prescribed by or under this Part for the service of notices by the private drainage board,	10 11
		(g)	it must be served on every person ratable in respect of the land.	12
	(3)	is ser	ore than one person is ratable in respect of any land, and the notice rved on one or more of such persons, the omission to serve the se on any other ratable person does not affect the liability of the or of any ratable person on whom the notice is served.	13 14 15 16
	(4)	to rec	e case of the lessee of private land, the liability for and the right cover rates is not affected by reason only of the fact that notice has been given to the lessee, unless the name of the lessee is on the roll oters kept in accordance with the regulations.	17 18 19 20
	(5)	comp	e Crown is the ratable person, the private drainage board must ply with any request of the Treasurer for the grouping or ration of rates on rate notices.	21 22 23
207	Obj	ection	us .	24
	(1)		ratable person may object to any valuation of which notice has served on the person in accordance with this Part.	25 26
	(2)	The o	objection must:	27
		(a)	be in or to the effect of the form prescribed by the regulations, and	28 29
		(b)	be lodged with the secretary of the drainage board within the time stated in the valuation notice.	30 31
	(3)	The j	private drainage board must forward the objection to the valuer.	32
	(4)		only grounds on which objection may be made under this section ne following:	33 34
		(a)	that the values assigned are wrong,	35

Water Management Bill 2000

Valuat		nage bo	oards Part 3 Division 4		
		(b)	that separate valuations have been made if one valuation should have been made,	1 2	
		(c)	that one valuation has been made if separate valuations should have been made,	3 4	
		(d)	that any apportionment of the valuation is not correct,	5	
		(e)	that the person named in or served with the valuation notice as a ratable person in respect of the ratable land is not such a ratable person,	6 7 8	
		(f)	that the land is not ratable.	9	
	(5)	that i	fact that an objection has been lodged does not affect the valuation is objected to, and any rate may be made, levied, and recovered on valuation as if no objection were pending.	10 11 12	
	(6)	and	e valuation is altered on objection, a due adjustment must be made, any amount paid in excess must be refunded and any amount aid is recoverable as arrears.	13 14 15	
208	App	eals t	to Local Court	16	
	(1)		objector who is dissatisfied with the valuer's decision on the ction may appeal to a Local Court against the decision.	17 18	
	(2)		Local Court may order the valuation to be altered if it is satisfied it is erroneous.	19 20	
	(3)	reco	costs incurred by the valuer in relation to the appeal are verable from the private drainage board in any court of competent diction as a debt due to the valuer.	21 22 23	
209	Inva	Invalid valuations			
	(1)	than	Local Court holds that a valuation is invalid for any reason other that the land is not ratable, the private drainage board may cause esh valuation to be made in place of the valuation held to be lid.	25 26 27 28	
	(2)		valuation made for the purposes of any rate made or to be made ken to have come into force when the invalid valuation would if	29 30	

valid have come into force.

Water Management Bill 2000

Joint private works

31

Clause 207

Chapter 4

Clause 210	Water Management Bill 2000
Chapter 4	Joint private works
Part 3	Private drainage boards
Division 5	Effect of new subdivisions

Division 5 Effect of new subdivisions

Division 5		5	Effect of new subdivisions	1
210	Sup	ply of	water to new holdings resulting from subdivisions	2
	(1)	If a	holding is subdivided, a new holding resulting from the	3
	()		ivision is not entitled to be connected to a private drainage board's	4
		drain	age works until a date determined by the board (not being a date	5
			er than the date on which the board became aware of the	6
		dispo	osition of that new holding).	7
	(2)	If rate	es for a period or year ending on 30 June, being the period or year	8
	, ,	durin	g which the date determined by the board under subsection (1)	9
		occui	rred, have not, before the date so determined, been levied in	10
			ect of the holding that was subdivided, the board must levy the	11
			for the whole of that period or year in respect of each of the new	12
		holdi	ngs that resulted from the subdivision, and that was disposed of.	13
211	Add	ditiona	ll works required as a result of subdivision	14
	(1)	The p	person who, immediately before the disposition of a new holding	15
		result	ting from a subdivision of land within a private drainage board's	16
			age district, was the landholder of the holding (the previous	17
			<i>holder</i>) must construct at his or her own cost such works as are	18
		neces	ssary to provide:	19
		(a)	means of conveying water to the board's drainage works from	20
			the new holding, and	21
		(b)	means of access from roads to any works of the board or any	22
			works provided for the purposes of paragraph (a) if that access	23
			would not be available except by crossing a channel of the	24
			drainage district, and	25
		(c)	means of access across a channel of the drainage district to the	26
			new holding if that means of access is required by reason of the	27
			subdivision.	28
	(2)	All w	vorks to be constructed under subsection (1) must be constructed	29
		befor	re the new holding is disposed of or within such period after the	30
			osition of the new holding as the board may in any particular case	31
		allow	7.	32
	(3)		vorks constructed or to be constructed under subsection (1) must	33
			onstructed in accordance with the approval in writing of the board	34
		in res	spect of location, design, form, dimensions and construction.	35

•		-		
Effect	of nev	v subdivisions Division 5		
	(4)	At the request of the previous landholder, the board may undertake, at the landholder's cost, the construction of any works required by this section.	1 2 3	
	(5)	The board may construct such works as have not been constructed by the previous landholder, and any costs and expenses are payable to the board either by the previous landholder or by the new landholder, as the board may determine.	4 5 6 7	
	(6)	If any part of the costs and expenses referred to in subsection (5) is recovered by the board from the new landholder, the new landholder may recover from the previous landholder the whole or that part of those costs or expenses, as the case may be.	8 9 10 11	
	(7)	On their completion, the control and management of any works constructed under this section is vested in the board.	12 13	
Division 6 Miscellaneous		14		
212	Diss	solution	15	
	(1)	When all amounts due by the private drainage board of any drainage union have been repaid, the landholders (being not less than one-third in number of those within the drainage district) may present a petition to the Governor for the dissolution of the union.	16 17 18 19	
	(2)	The Governor may notify such petition in a local newspaper, and if no sufficient cause to the contrary is shown by other landholders within the drainage district, may proclaim that the union is dissolved.	20 21 22	
	(3)	The Governor may dissolve any union if its drainage works cease to be the subject of a drainage work approval.	23 24	
	(4)	The regulations may make provision for or with respect to the winding up of a private drainage board and for the disposal of any residual assets of the board.	25 26 27	
213	Deb	nts	28	
		Any rate, charge, fee, or money due to a private drainage board under the provisions of this Part or of any regulation under this Part may be recovered as a debt or liquidated demand in any court of competent jurisdiction.	29 30 31 32	

Joint private works

Clause 211

Chapter 4

Clause 214	Water Management Bill 2000
Chapter 4	Joint private works
Part 3	Private drainage boards
Division 6	Miscellaneous

214	Accounts					
	(1)	The a year.	accounts of a private drainage board must be audited once every	2 3		
	(2)	a state and e of the	vate drainage board must each year publish in a local newspaper ement of the receipts and payments or alternatively of the income expenditure for the next preceding year, certified under the hands e chairperson of the board or manager and the auditor, or forward y of such statement to each landholder of land within the drainage ct.	4 5 6 7 8 9		
	(3)	The p	private drainage board must forward a copy of the statement to the ster.	10 11		
215	Regulations					
			regulations may make provisions for or with respect to the wing matters:	13 14		
		(a)	the appointment, payment and dismissal of officers and employees of a drainage board,	15 16		
		(b)	the making and levying of rates and the time within which rates must be made,	17 18		
		(c)	the keeping of accounts of private drainage boards,	19		
		(d)	the qualifications to be held by auditors appointed by a private drainage board,	20 21		
		(e)	the appointment, powers and duties of valuers appointed by a private drainage board,	22 23		
		(f)	the form, preparation, sealing, keeping and inspection of the valuation book,	24 25		
		(g)	the entries that are to be made in the valuation book.	26		

Prelim	Preliminary Division 1		
Part	4 I	Private water trusts	1
Divis	sion '	1 Preliminary	2
216	App	olication of Part	3
		This Part applies to all private water trusts under Part 3 of the former	4
		Water Act 1912 that were in existence immediately before the repeal	5
		of that Part, but does not authorise the establishment of any new private water trusts.	6 7
217	Def	initions	8
		In this Part:	9
		<i>member</i> , in relation to a private water trust, means a member appointed or elected under section 219.	10 11
		<i>private water trust</i> means a trust constituted under Part 3 of the former <i>Water Act 1912</i> .	12 13
		<i>ratepayer</i> means person paying rates in respect of land within a water supply district.	14 15
		water supply district means the district in respect of which a private water trust is constituted.	16 17
218	Rec	uirements for access licences and approvals	18
	(1)	Nothing in this Part authorises a private water trust to take water	19
		otherwise than in accordance with an access licence and water supply	20
		work approval held by the trust.	21
	(2)		22
		supply district to use water otherwise than in accordance with a water use approval held by the private water trust.	23 24
Divis	sion 2	2 Private water trusts	25
219	App	pointment and election of members	26
	(1)	For private water trusts with 3 members, 1 is to be appointed by the Minister, and 2 are to be elected.	27 28

Joint private works

Private water trusts

Clause 216

Chapter 4 Part 4

Chapter 4 Part 4			oint private works Private water trusts		
Divisio	n 2	Р	Private water trusts		
	(2)		private trusts with 5 members, 2 are to be appointed by the ster, and 3 are to be elected.	1 2	
	(3)		of the members appointed by the Minister is to be appointed as person.	3 4	
	(4)	The 1	regulations may make provision for or with respect to:	5	
		(a)	the conduct of elections for the elected members of a private water trust, and	6 7	
		(b)	other matters concerning the constitution and procedure of a trust.	8	
220	Alte	eration	of water supply districts	10	
	(1)	forward suppl petiti	ro-thirds of the landholders of lands within any area sign and ard to the Minister a petition that such area be included in a water ly district, the Minister must, unless of the opinion that the ion should be refused, refer the petition to the members of the ite water trust for the district.	11 12 13 14 15	
	(2)	been of win	eceipt from the members of notice that the proposed alteration has approved by a special general meeting of the voters of the trust, hich at least 14 days' notice has been given in the prescribed her, the Minister may, by notification in the Gazette, alter the daries of the district accordingly.	16 17 18 19 20	
221	Removal of land from water supply district				
	(1)	water perio the M the M	y lands within a water supply district have not benefited from the r management works of the private water trust for a continuous of of 3 years or more and the landholder of such lands applies to Minister for the lands to be excised from the water supply district, Minister must refer the application to the members of the private r trust for the district.	22 23 24 25 26 27	
	(2)	lands voter the p	members must refer the question of the proposed excision of the s from the water supply district to a special general meeting of the s of the trust (of which at least 14 days' notice has been given in rescribed manner) and must convey to the Minister the decision e voters.	28 29 30 31 32	
	(3)	The 1	members must also advise the Minister as to:	33	
		(a)	the reasons why such lands have not so benefited or have ceased so to benefit, and	34 35	

Water Management Bill 2000

Joint private works Private water trusts Private water trusts			Chapter 4 Part 4 Division 2	
		(b)	the practicability or otherwise of extending or improving the water management works of the trust so as to benefit the said lands, and	1 2 3
		(c)	the effect that the granting of the request would have on the general administration and finances of the trust.	4 5
	(4)	the la	finister is then to decide whether or not the whole or any part of nds referred to in the request are to be excised from the water y district.	6 7 8
	(5)	from	Minister may, by notification in the Gazette, excise such lands the water supply district and on the publication of such cation the water supply district is taken to be altered accordingly.	9 10 11
222	Unio	on of ti	rusts	12
		district separate been goin the	e joint application of members of any two adjoining water supply ets, duly approved by a majority of the voters of each district at atte special general meetings (of which at least 14 days' notice has given in the prescribed manner), the Minister may, by notification Gazette, transfer any part of one water supply district to the other supply district.	13 14 15 16 17 18
Divis	ion 3	3	Functions of members	19
223	Duties of members			
			nembers of a private water trust have the function of maintaining dministering the trust's water supply works.	21 22
224	Pow	ers an	nd duties of members	23
	(1)		nembers of a private water trust have and may exercise the ving functions:	24 25
		(a)	to establish and maintain a management program for the water supply district,	26 27
		(b)	to maintain in a state of efficiency the water management works under their charge and renew such works if necessary,	28 29
		(c)	to fix and levy rates to provide for the maintenance, renewal and management of such water management works, and for interest, charges, and a sinking fund,	30 31 32

Clause 221

Clause			/ater Management Bill 2000	
-	•		oint private works rivate water trusts	
Divisio	n 3		unctions of members	
		(d)	to keep proper accounts of all money received and paid,	1
		(e)	to pay to the Treasury, at such times as may be fixed by the	2
			Minister, the interest and charges payable by them, and if necessary make due provision for a sinking fund,	3 4
		(f)	to appoint, with the approval of the Minister, such officers or employees as may be required,	5 6
		(g)	to borrow money, but only as provided in this section.	7
	(2)	A loa	n by way of limited overdraft may be obtained for the purpose of	8
			ing necessary legitimate expenditure prior to the collection of rates	9
			or the purpose of carrying out urgent works of renewal, or	10
			cement, or other emergency, for which sufficient funds are not ediately available.	11 12
	(2)		•	
	(3)		pplication by the members of a private water trust the Minister	13
			issue a certificate of limit of overdraft, in which must be named urpose in respect of which the loan may be obtained and the limit	14 15
			nount to be borrowed.	16
	(4)	The s	sum to be stated in the certificate as the limit of the overdraft is in	17
	(')		discretion of the Minister, but must not exceed the estimated	18
			ant required for the purpose mentioned therein plus 10% of such	19
		amou	int.	20
	(5)	The 1	oan is subject to any conditions inserted by the Minister in the	21
			icate, and the money borrowed is taken to be secured on the	22
			ne of the trust and must be repaid within the time fixed in the	23
		certif	icate.	24
	(6)		ums received on account of a trust's sinking fund must be carried	25
			e Treasurer to a special account, to be called "The Water Supply	26
		Loan	Redemption Fund", and all other sums to the Consolidated Fund.	27
	(7)		members of a private water trust may on application by any	28
			ayer defer or suspend payment of rates by the ratepayer for such	29
		_	d or periods and on such terms and conditions as the Minister	30
		may a	approve.	31
225	Sup	ply of	water	32
	(1)		nembers of a private water trust may for the more beneficial use	33
			ficient distribution of the water supplied by the trust's water	34
		suppl	y works:	35

	(a)	cease to supply water through any portion of the works in the water supply district or to any lands within such district, or	1 2
	(b)	deviate the course of a water supply work or otherwise alter in any way the works in the water supply district.	3 4
(2)		e taking any such action, the members must obtain the written nt of any ratepayer:	5 6
	(a)	whose benefit from those works will be diminished by the action of the members, or	7 8
	(b)	the supply of water or means of supply of water to whose lands will be affected,	9 10
		nust also obtain the written approval of the Minister of the eers' proposed action.	11 12
(3)		nembers of a private water trust may cut off or withhold the y of water to any land:	13 14
	(a)	if any meter used to measure such supply or any outlet is out of repair or, in the opinion of the members, unsatisfactory for the expeditious or effective supply of water to such land, or	15 16 17
	(b)	if, in the opinion of the members, such course is necessary owing to drought or any accident or other unavoidable cause, or	18 19 20
	(c)	if the landholder or person requiring a supply of water neglects to comply with the lawful requirements of the members as to the installation of outlets or meters or instruments for measuring the quantity of water, or	21 22 23 24
	(d)	if the landholder or person requiring a supply of water neglects to comply with any lawful requirements of the members to repair or alter water connections, outlets, channels, ditches, pipes, fittings or appliances connected to the water management works under the control of the members, or	25 26 27 28 29
	(e)	if the landholder of the land fails to take such steps as may be necessary to ensure compliance with any order or public notice of the members requiring consumers of water to economise its use in time of drought or scarcity of supply, or	30 31 32 33
	(f)	if any rates fixed and levied in respect of the land are, after the due date of payment, unpaid and approval to the supply of water to the land being cut off or withheld is given:	34 35 36

Clause 225		Water Management Bill 2000			
Chapter 4 Joint private works					
Part 4	Part 4 Private water trusts				
Divisio	Division 3 Functions of members				
		4) 10.1			
			constituted by 3 members, by the	1	
			e trust and all the elected members, or constituted by 5 members, by the	2 3	
			ne trust and a majority of the elected	5 5	
	(4)	If the Minister:		6	
	(+)		1 1 6		
		(a) reduces or discontinues a private water trust, or	the supply of water to the members of	7 8	
		(b) directs the members of a		9	
			intity of water being taken or diverted	10	
			y water source, or	11	
			e taking or diverting of water from a	12	
		water source,		13	
			off or withhold the supply of water to	14	
		any land.		15	
	(5)		or reduction of the supply of water by	16	
		<u> </u>	trust under this section does not affect	17	
			respect of the rates on the land the	18	
		supply of water to which has be	een cut off, withheld or reduced.	19	
226	Pov	ver of members to enter and in	spect	20	
		In the exercise of their function	ns, by themselves or their officers, the	21	
			st may enter any land within the water	22	
			inspection or survey they consider	23	
			alterations to any water management	24	
		loss, injury or damage.	oid, as far as practicable, causing any	25 26	
227	No	compensation		27	
		No compensation is payable in exercise of its functions under	connection with a private water trust's his Part.	28 29	

Water Management Bill 2000	Clause 228
Joint private works	Chapter 4
Private water trusts	Part 4
Rating	Division 4

(1) For the purpose of providing money in connection with the exercise of their functions under this Part, the members of a private water trust may fix and levy rates on the lands within the water supply district as follows: (a) in connection with the supply of water for stock purposes: (i) a rate per hectare of the land benefited by the water management works must be fixed, or (ii) if water is supplied down a natural channel, a rate per kilometre of the lands so benefited, measured according to the frontage to the channel, may be fixed, which rate may vary in proportion to the benefit received, (b) in connection with the supply of water for domestic purposes, a rate for each separate holding in the water supply district must be fixed, which rate may vary in proportion to the benefit received, (c) in connection with the construction and maintenance of flood works, a rate per hectare of the land benefited by the works must be fixed, which rate may vary: (i) according to the distance of the land from works for the prevention of floods or the control of flood waters, and (ii) in proportion to the benefit received, (2) (d) in connection with irrigation, a rate must be levied on the land within the water supply district: (i) that is suitable for production under irrigation, and (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part is used.	Division 4			Rating	1		
their functions under this Part, the members of a private water trust may fix and levy rates on the lands within the water supply district as follows: (a) in connection with the supply of water for stock purposes: (i) a rate per hectare of the land benefited by the water management works must be fixed, or (ii) if water is supplied down a natural channel, a rate per kilometre of the lands so benefited, measured according to the frontage to the channel, may be fixed, which rate may vary in proportion to the benefit received, (b) in connection with the supply of water for domestic purposes, a rate for each separate holding in the water supply district must be fixed, which rate may vary in proportion to the benefit received, (c) in connection with the construction and maintenance of flood works, a rate per hectare of the land benefited by the works must be fixed, which rate may vary: (i) according to the distance of the land from works for the prevention of floods or the control of flood waters, and (ii) in proportion to the benefit received, (d) in connection with irrigation, a rate must be levied on the land within the water supply district: (i) that is suitable for production under irrigation, and (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part	228	Rate	es		2		
may fix and levy rates on the lands within the water supply district as follows: (a) in connection with the supply of water for stock purposes: (i) a rate per hectare of the land benefited by the water management works must be fixed, or (ii) if water is supplied down a natural channel, a rate per kilometre of the lands so benefited, measured according to the frontage to the channel, may be fixed, which rate may vary in proportion to the benefit received, (b) in connection with the supply of water for domestic purposes, a rate for each separate holding in the water supply district must be fixed, which rate may vary in proportion to the benefit received, (c) in connection with the construction and maintenance of flood works, a rate per hectare of the land benefited by the works must be fixed, which rate may vary: (i) according to the distance of the land from works for the prevention of floods or the control of flood waters, and (ii) in proportion to the benefit received, (d) in connection with irrigation, a rate must be levied on the land within the water supply district: (i) that is suitable for production under irrigation, and (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part		(1)	For th	e purpose of providing money in connection with the exercise of	3		
follows: (a) in connection with the supply of water for stock purposes: (i) a rate per hectare of the land benefited by the water management works must be fixed, or (ii) if water is supplied down a natural channel, a rate per kilometre of the lands so benefited, measured according to the frontage to the channel, may be fixed, which rate may vary in proportion to the benefit received, (b) in connection with the supply of water for domestic purposes, a rate for each separate holding in the water supply district must be fixed, which rate may vary in proportion to the benefit received, (c) in connection with the construction and maintenance of flood works, a rate per hectare of the land benefited by the works must be fixed, which rate may vary: (i) according to the distance of the land from works for the prevention of floods or the control of flood waters, and (ii) in proportion to the benefit received, (d) in connection with irrigation, a rate must be levied on the land within the water supply district: (i) that is suitable for production under irrigation, and (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part							
(a) in connection with the supply of water for stock purposes: (i) a rate per hectare of the land benefited by the water management works must be fixed, or (ii) if water is supplied down a natural channel, a rate per kilometre of the lands so benefited, measured according to the frontage to the channel, may be fixed, which rate may vary in proportion to the benefit received, (b) in connection with the supply of water for domestic purposes, a rate for each separate holding in the water supply district must be fixed, which rate may vary in proportion to the benefit received, (c) in connection with the construction and maintenance of flood works, a rate per hectare of the land benefited by the works must be fixed, which rate may vary: (i) according to the distance of the land from works for the prevention of floods or the control of flood waters, and (ii) in proportion to the benefit received, (d) in connection with irrigation, a rate must be levied on the land within the water supply district: (i) that is suitable for production under irrigation, and (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part							
(i) a rate per hectare of the land benefited by the water management works must be fixed, or (ii) if water is supplied down a natural channel, a rate per kilometre of the lands so benefited, measured according to the frontage to the channel, may be fixed, which rate may vary in proportion to the benefit received, (b) in connection with the supply of water for domestic purposes, a rate for each separate holding in the water supply district must be fixed, which rate may vary in proportion to the benefit received, (c) in connection with the construction and maintenance of flood works, a rate per hectare of the land benefited by the works must be fixed, which rate may vary: (i) according to the distance of the land from works for the prevention of floods or the control of flood waters, and (ii) in proportion to the benefit received, (d) in connection with irrigation, a rate must be levied on the land within the water supply district: (i) that is suitable for production under irrigation, and (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part							
management works must be fixed, or (ii) if water is supplied down a natural channel, a rate per kilometre of the lands so benefited, measured according to the frontage to the channel, may be fixed, which rate may vary in proportion to the benefit received, (b) in connection with the supply of water for domestic purposes, a rate for each separate holding in the water supply district must be fixed, which rate may vary in proportion to the benefit received, (c) in connection with the construction and maintenance of flood works, a rate per hectare of the land benefited by the works must be fixed, which rate may vary: (i) according to the distance of the land from works for the prevention of floods or the control of flood waters, and (ii) in proportion to the benefit received, (d) in connection with irrigation, a rate must be levied on the land within the water supply district: (i) that is suitable for production under irrigation, and (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part			(a)				
(ii) if water is supplied down a natural channel, a rate per kilometre of the lands so benefited, measured according to the frontage to the channel, may be fixed, which rate may vary in proportion to the benefit received, (b) in connection with the supply of water for domestic purposes, a rate for each separate holding in the water supply district must be fixed, which rate may vary in proportion to the benefit received, (c) in connection with the construction and maintenance of flood works, a rate per hectare of the land benefited by the works must be fixed, which rate may vary: (i) according to the distance of the land from works for the prevention of floods or the control of flood waters, and (ii) in proportion to the benefit received, (d) in connection with irrigation, a rate must be levied on the land within the water supply district: (i) that is suitable for production under irrigation, and (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part							
kilometre of the lands so benefited, measured according to the frontage to the channel, may be fixed, which rate may vary in proportion to the benefit received, (b) in connection with the supply of water for domestic purposes, a rate for each separate holding in the water supply district must be fixed, which rate may vary in proportion to the benefit received, (c) in connection with the construction and maintenance of flood works, a rate per hectare of the land benefited by the works must be fixed, which rate may vary: (i) according to the distance of the land from works for the prevention of floods or the control of flood waters, and (ii) in proportion to the benefit received, (d) in connection with irrigation, a rate must be levied on the land within the water supply district: (i) that is suitable for production under irrigation, and (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part				(ii) if water is supplied down a natural channel, a rate per			
to the frontage to the channel, may be fixed, which rate may vary in proportion to the benefit received, (b) in connection with the supply of water for domestic purposes, a rate for each separate holding in the water supply district must be fixed, which rate may vary in proportion to the benefit received, (c) in connection with the construction and maintenance of flood works, a rate per hectare of the land benefited by the works must be fixed, which rate may vary: (i) according to the distance of the land from works for the prevention of floods or the control of flood waters, and (ii) in proportion to the benefit received, (d) in connection with irrigation, a rate must be levied on the land within the water supply district: (i) that is suitable for production under irrigation, and (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part							
(b) in connection with the supply of water for domestic purposes, a rate for each separate holding in the water supply district must be fixed, which rate may vary in proportion to the benefit received, 17 (c) in connection with the construction and maintenance of flood works, a rate per hectare of the land benefited by the works must be fixed, which rate may vary: 20 (i) according to the distance of the land from works for the prevention of floods or the control of flood waters, and (ii) in proportion to the benefit received, 23 (d) in connection with irrigation, a rate must be levied on the land within the water supply district: 25 (i) that is suitable for production under irrigation, and 26 (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. 28 (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: 31 (a) the fact that any such part is or is not actually used for production under irrigation, or 33 (b) the type of production under irrigation for which any such part 34					12		
a rate for each separate holding in the water supply district must be fixed, which rate may vary in proportion to the benefit received, (c) in connection with the construction and maintenance of flood works, a rate per hectare of the land benefited by the works must be fixed, which rate may vary: (i) according to the distance of the land from works for the prevention of floods or the control of flood waters, and (ii) in proportion to the benefit received, (d) in connection with irrigation, a rate must be levied on the land within the water supply district: (i) that is suitable for production under irrigation, and (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part				which rate may vary in proportion to the benefit received,	13		
be fixed, which rate may vary in proportion to the benefit received, (c) in connection with the construction and maintenance of flood works, a rate per hectare of the land benefited by the works must be fixed, which rate may vary: (i) according to the distance of the land from works for the prevention of floods or the control of flood waters, and (ii) in proportion to the benefit received, (d) in connection with irrigation, a rate must be levied on the land within the water supply district: (i) that is suitable for production under irrigation, and (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part			(b)	in connection with the supply of water for domestic purposes,	14		
received, (c) in connection with the construction and maintenance of flood works, a rate per hectare of the land benefited by the works must be fixed, which rate may vary: (i) according to the distance of the land from works for the prevention of floods or the control of flood waters, and (ii) in proportion to the benefit received, (d) in connection with irrigation, a rate must be levied on the land within the water supply district: (i) that is suitable for production under irrigation, and (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part					15		
(c) in connection with the construction and maintenance of flood works, a rate per hectare of the land benefited by the works must be fixed, which rate may vary: (i) according to the distance of the land from works for the prevention of floods or the control of flood waters, and (ii) in proportion to the benefit received, (d) in connection with irrigation, a rate must be levied on the land within the water supply district: (i) that is suitable for production under irrigation, and (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part							
works, a rate per hectare of the land benefited by the works must be fixed, which rate may vary: (i) according to the distance of the land from works for the prevention of floods or the control of flood waters, and (ii) in proportion to the benefit received, (d) in connection with irrigation, a rate must be levied on the land within the water supply district: (i) that is suitable for production under irrigation, and (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part					17		
must be fixed, which rate may vary: (i) according to the distance of the land from works for the prevention of floods or the control of flood waters, and (ii) in proportion to the benefit received, (d) in connection with irrigation, a rate must be levied on the land within the water supply district: (i) that is suitable for production under irrigation, and (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part			(c)				
(i) according to the distance of the land from works for the prevention of floods or the control of flood waters, and (ii) in proportion to the benefit received, 23 (d) in connection with irrigation, a rate must be levied on the land within the water supply district: 25 (i) that is suitable for production under irrigation, and (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. 28 (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or 33 (b) the type of production under irrigation for which any such part							
prevention of floods or the control of flood waters, and (ii) in proportion to the benefit received, (d) in connection with irrigation, a rate must be levied on the land within the water supply district: (i) that is suitable for production under irrigation, and (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part							
(ii) in proportion to the benefit received, (d) in connection with irrigation, a rate must be levied on the land within the water supply district: (i) that is suitable for production under irrigation, and (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part							
within the water supply district: (i) that is suitable for production under irrigation, and (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part 34							
within the water supply district: (i) that is suitable for production under irrigation, and (ii) that is accessible to the works of the trust by means of recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part 34			(d)	in connection with irrigation, a rate must be levied on the land	24		
(ii) that is accessible to the works of the trust by means of recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part 34					25		
recognised methods of irrigation. (2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part 34							
(2) If land is liable for rates under subsection (1) (d), the members of a private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part 34				· · ·			
private water trust may, in fixing the rate, fix different amounts for different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part 30 31 32 33 33					28		
different parts of the land, having regard to: (a) the fact that any such part is or is not actually used for production under irrigation, or (b) the type of production under irrigation for which any such part 31 32 33		(2)					
(a) the fact that any such part is or is not actually used for production under irrigation, or 33 (b) the type of production under irrigation for which any such part 34							
production under irrigation, or 33 (b) the type of production under irrigation for which any such part 34							
(b) the type of production under irrigation for which any such part 34			(a)				
			(1.)				
			(b)	• • • • • • • • • • • • • • • • • • • •			

Chapter 4 Part 4 Division 4	Joint private works Private water trusts Rating		
(.	3) In any case for which this section does not provide, a rate per hectare of the land benefited, directly or indirectly, by the works must be fixed yearly, and must, as far as practicable, be in proportion to the benefit received.	1 2 3 4	
(4	In the case of a supply for more than one purpose, separate rates may be fixed, calculated on the basis set out for each such purpose.	5 6	
(:	All rates are a charge on the land in respect of which they are levied and are payable by the landholder.	7 8	
(Any landholder aggrieved by the amount of a rate may appeal to a Local Court, which must hear and determine the matter, and may confirm or vary such amount.	9 10 11	
(*	If in any such appeal the Local Court reduces the amount at which the appellant is rated, it must increase the other ratings of the trust in such amounts as it thinks just, if it considers such course necessary, in order to secure that the total amount to be received by the trust for rates is not to be diminished by the reduction.	12 13 14 15 16	
229 S	upplementary rating powers	17	
(If the members of a private water trust are unable to meet the cost of administration or of an extension or improvement of the works under their control or other liability, they may in writing petition the Minister to approve an increase in the maximum rate that may be assessed by them, and must set out in such petition the reasons for their request.	18 19 20 21 22	
(2	2) The Minister must inform the members whether or not the Minister is prepared to grant their petition and, if so, must inform them of the maximum rate that the Minister is prepared to approve.	23 24 25	
(:	The members must submit the proposed increase in maximum rate that the Minister is prepared to approve to a special general meeting of the voters of the trust (of which at least 14 days' notice has been given in the prescribed manner).	26 27 28 29	
(4	The members must report to the Minister whether an affirmative or negative vote has been carried at the meeting.	30 31	
(:	6) If an affirmative vote has been cast the Minister must grant the petition of the members by notification sent to them and published in the Gazette, and thereafter the members may assess rates to the maximum approved by the Minister.	32 33 34 35	

Clause 228

Private Rating		er trusts Part 4 Divisio	n 4
230	Sur	rplus water	1
		When a private water trust has any surplus water that is not recommon for the purpose for which the trust was constituted, the trust matthe water, by measure or otherwise:	
		(a) to any ratepayer for any other purpose, or	5
		(b) if no ratepayer requires it, to any other person.	6
231	Acc	counts	7
	(1)	The members of a private water trust must submit to the Minister every year, and at such other times as the Minister may dire audited statement of the accounts of the trust.	
	(2)	The Minister may, if of opinion that the audited statement is def or unsatisfactory, appoint an auditor to prepare a proper statem accounts, at the cost of the trust.	
Divis	ion (5 Effect of new subdivisions	14
232	Sup	pply of water to new holdings resulting from subdivisions	15
	(1)	If a holding is subdivided, a new holding resulting from subdivision is not entitled to a supply of water from a private trust's water supply works until a date determined by the trust being a date earlier than the date on which the board became aw the disposition of that new holding).	water 17 st (not 18
	(2)		or year 21 on (1) 22 ied in 23 e rates 24 e new 25

Joint private works

Clause 230

Chapter 4

Clause 233 Water Management Bill 2000

Chapter 4 Joint private works
Part 4 Private water trusts
Division 5 Effect of new subdivisions

233 Additional works required as a result of subdivision

- (1) The person who, immediately before the disposition of a new holding resulting from a subdivision of land within a private water trust's water supply district, was the landholder of the holding (the *previous landholder*) must construct at his or her own cost such works as are necessary to provide:
 - (a) means of conveying water to the new holding from the trust's water supply works, and

Q

2.1

- (b) means of access from roads to any works of the trust or any works provided for the purposes of paragraph (a) if that access would not be available except by crossing a channel of the water supply district, and
- (c) means of access across a channel of the water supply district to the new holding if that means of access is required by reason of the subdivision.
- (2) All works to be constructed under subsection (1) must be constructed before the new holding is disposed of or within such period after the disposition of the new holding as the trust may in any particular case allow.
- (3) All works constructed or to be constructed under subsection (1) must be constructed in accordance with the approval in writing of the trust in respect of location, design, form, dimensions and construction.
- (4) At the request of the previous landholder, the trust may undertake, at the landholder's cost, the construction of any works required by this section.
- (5) The trust may construct such works as have not been constructed by the previous landholder, and any costs and expenses are payable to the trust either by the previous landholder or by the new landholder, as the trust may determine.
- (6) If any part of the costs and expenses referred to in subsection (5) is recovered by the trust from the new landholder, the new landholder may recover from the previous landholder the whole or that part of those costs or expenses, as the case may be.
- (7) On their completion, the control and management of any works constructed under this section is vested in the trust.

Water Management Bill 2000	Clause 234
Joint private works	Chapter 4
Private water trusts	Part 4
Miscellaneous	Division 6

Divis	ion (6	Miscellaneous	1
234	234 Min		nay remove members and Governor may dissolve trust	2
	(1)	In the	event of any delay in the election of members of a private water	3
			or in the event of any default by members of a private water trust	4
			e discharge of their duties under this Part or of the ratepayers	5
			g to elect members, or the required number of members, the ster may, by notification in the Gazette:	7
		(a)	remove a member or members from office, and direct the	8
		()	election of another member or other members, or	9
		(b)	remove a member or members (if any) and assume all of the	10
			functions of the members, and may appoint a manager of the	11
			affairs of the trust.	12
	(2)		Minister may at any time direct the election of members of a	13
			te water trust, and on such election all of the functions of the	14
		mem	bers are revested in the members.	15
	(3)		Governor may, if in the Governor's opinion sufficient reason	16
			s, dissolve a trust, and may extinguish any liability of such trust crown.	17 18
	(4)			
	(4)		egulations may make provision for or with respect to the winding	19
		the tr	a private water trust and for the disposal of any residual assets of	20 21
235	Reg	gulatio	ns	22
			regulations may make provision for or with respect to the	23
		follov	wing matters:	24
		(a)	the election of members of a private water trust and the making	25
			and revision of the rolls of voters, and the mode of voting,	26
		(b)	the intervals within which meetings of members of a private	27
			water trust must be held,	28
		(c)	the procedure at such meetings,	29
		(d)	the appointment, payment, and dismissal of officers and	30
			employees of a private water trust,	31
		(e)	the fixing and notifying of rates,	32
		(f)	the hearing of appeals,	33

Clause 235 Chapter 4 Part 4	J	Vater Management Bill 2000 Ioint private works Private water trusts	
Division 6	Miscellaneous		
	(g)	prescribing the method or methods to be adopted to determine the quantity of water for which payment is to be made by the trust,	1 2 3
	(h)	the keeping of the accounts of private water trusts,	4
	(i)	regulating the payment to the Crown of interest and charges by private water trusts,	5 6
	(j)	regulating special meetings of voters of a private water trust in pursuance of this Part.	7 8

Preliminary	Division 1
Fish River water supply works	Part 1
Public works	Chapter 5
Water Management Bill 2000	Clause 236

Cha	pter 5	Public works
Part	1 Fish	River water supply works
Divis	ion 1	Preliminary
236	Definition	ns en
	In th	is Part:
		River water supply works means the concrete dam on Fish River k at Oberon, together with:
	(a)	its associated gravitation main, concrete reservoirs and reticulation systems, and
	(b)	the pumping station at Oberon, and
	(c)	all incidental and connected works, and
	(d)	all additions, amplifications, improvements and extensions to those works.
	-	<i>tal area</i> means an area of land declared under Division 3 to be a sal area.
237	Fish Rive	r water supply works controlled by the Minister
		Fish River water supply works are to be controlled and nistered by the Minister.
238	Minister r	may repair works
	Mini wate	the purpose of supplying any person with water under this Part, the ster may at all times use and repair any portion of the Fish River r supply works even if a local council is charged with the care and agement of those works.

Divisio	on 2	Supply of water	=
Divis	sion 2	2 Supply of water	
239	Min	ister may supply water	
	(1)	The Minister may supply water from the Fish River water supply works to any person who enters into a contract with the Minister to accept a supply of water.	
	(2)	Any such contract may contain such terms and conditions as the Minister thinks fit.	e
	(3)	The Minister may from time to time, by order published in the Gazette	:
		(a) fix the price per kilolitre to be paid by any person for wate supplied by the Minister in any year, and	r
		(b) fix the minimum amount to be paid by any person in any one year irrespective of the quantity of water supplied.	e
240	Cou	ıncil not to supply outside area without consent	
		Except with the consent of the Minister, a local council that is supplied with water from the Fish River water supply works must not supply water to any premises situated outside the area of the council.	
241	No	compensation for failure of supply	
		The Minister is at all times entitled to discontinue the supply to a local council or other person whenever the Minister considers it necessary in the interests of public health or for the purpose of ensuring a proper supply.	y
Divis	ion (Special areas	
242	Spe	ecial areas	
	(1)	The Governor may, on the recommendation of the Minister, by orde published in the Gazette, declare an area of land described in the orde to be a special area in relation to the Fish River water supply works.	r
	(2)	The Minister may recommend an order only if of the opinion that the exercise of the State's water rights could be adversely affected unless the order is made.	

Fish River water supply works

Public works

Clause 239 Chapter 5

Part 1

Specia	al area	IS	Division 3	
243	Crown land in special areas			
	(1)		on may not be taken under the <i>Crown Lands Act 1989</i> in respect and within a special area unless:	2
		(a)	the Minister has given approval in writing, and	4
		(b)	any conditions to which the approval is subject are complied with.	6
	(2)	appo	Minister may, in a special area, exercise the functions of a trustee binted under Part 5 of the <i>Crown Lands Act 1989</i> without being binted as such.	5 9
244	Exe	rcise	of certain functions in special areas	10
	(1)	exer	ublic agency may not, in relation to land within a special area, cise functions other than functions under this Part unless notice is given, as prescribed by the regulations, to the Minister.	11 12 13
	(2)		receipt of such a notice, the Minister may make such esentations to the public agency as the Minister thinks fit.	14 15
	(3)		ublic agency may not exercise functions contrary to any such esentations made by the Minister unless:	16 17
		(a)	at least 14 days' notice has been given to the Minister of the functions intended to be exercised, and	18 19
		(b)	that period has expired without the matter being referred as a dispute for determination under this Act.	20 21
	(4)	In th	is section, <i>public agency</i> means:	22
		(a)	the Governor, or	23
		(b)	a Minister of the Crown, or	24
		(c)	a statutory body.	25
Divis	sion 4	4	Finance	26
245	Acc	ounts	s to be kept in Special Deposits Account	27
		Trea	re is to be established in the Special Deposits Account in the sury an account to be called the "Fish River Water Supply bunt".	28 29 30

Fish River water supply works

Public works

Clause 243

Chapter 5 Part 1

Division 4		Finance	
246	6 Allocation of money		
		Within the Fish River Water Supply Account is to be kept an account to be called the "Fish River Water Supply Working Account", in this Part referred to as the <i>Working Account</i> .	:
	(2)	There must be credited to the Working Account all revenue received by the Minister in pursuance of this Part and such amounts as may from time to time be appropriated by Parliament for the purpose.	
	(3)	There must be debited to the Working Account the following charges, in the following order:	
		(a) firstly, the cost of administration, operation, repair and minor renewal of the Fish River water supply works by the Minister,	1
		(b) secondly, interest on the interest-bearing part of the capital debt for any year at a rate or rates to be determined by the Treasurer, but not exceeding the average rate payable during such year by the Government for loan money and exchange at a rate or rates to be determined by the Treasurer, the amount of which interest and exchange must be credited to the Consolidated Fund,	1 1 1 1 1 1
		(c) thirdly, such contributions (if any) to a sinking fund as the Treasurer may direct.	1
247	Pay	yments by Treasurer	20
	(1)	When in any year the revenue received by the Minister is insufficient to meet the charges payable to the Working Account under this Division, the Treasurer may, out of money provided by Parliament, pay to the Working Account the amount of such deficiency.	2 2 2 2 2
	(2)	Any money paid to the Working Account must, after payment of the charges payable to the Working Account under this Division, be repaid to the Treasurer out of any balance remaining in the Working Account with interest at such rate and at such times and by such instalments as the Treasurer may determine.	2: 2: 2: 2:
Divis	sion (5 Miscellaneous	30
248	Reg	gulations	3:
	(1)	The regulations may make provision for or with respect to special areas.	32

Chapter 5

Part 1

Water Management Bill 2000

Fish River water supply works

Public works

Water Management Bill 2000	Clause 248
Public works	Chapter 5
Fish River water supply works	Part 1
Miscellaneous	Division 5

(2)	A regulation made for or with respect to a special area prevails to the	1
()	extent of any inconsistency with a statutory instrument made under	2
	another Act, including an environmental planning instrument made	3
	under the Environmental Planning and Assessment Act 1979.	4

Part	2 Hunt	ter Valley flood mitigation works	1
Divis	sion 1	Preliminary	2
249	Applicati	on of Part	3
	This	Part applies to and in respect of land in the Hunter Valley.	4
250	Definition	ns	5
	In th	is Part:	6
	Man	tter Catchment Management Trust means the Hunter Catchment magement Trust established by section 20 of the Catchment magement Act 1989.	7 8 9
	into	tter River includes all tributaries that, directly or indirectly, flow the Hunter River, and includes the estuary of the Hunter River and rms and branches of that estuary.	10 11 12
	inclu depo	ther Valley means the catchment area of the Hunter River, and adds the land shown on the plan marked "DWR PLA No 20/2551" osited in the head office of the Department, and also includes such r land as is declared by the regulations to form part of the Hunter ev.	13 14 15 16
	<i>leved</i> of o excl	e bank means a levee bank designed or intended for the purpose or that could or might have the effect of excluding or partially uding the waters of the Hunter River or waters overflowing from Hunter River from any land.	18 19 20 21
		ective works means works for the protection of any works vested otherwise under the control of a public authority:	22 23
	(a)	against flooding by waters overflowing from the Hunter River, or	24 25
	(b)	against the effects of any such flooding, or	26
	(c)	against the effects of river bank erosion.	27
	tree	includes sapling and shrub.	28

Chapter 5

Division 1

Part 2

Water Management Bill 2000

Hunter Valley flood mitigation works

Public works

Preliminary

Water Management Bill 2000	Clause 251
Public works	Chapter 5
Hunter Valley flood mitigation works	Part 2
Powers of Minister	Division 2

Division 2		2	Powers of Minister	
251	And	illary (construction powers	2
			he purpose of constructing, operating or maintaining any flood in the Hunter Valley, any authorised officer:	3
		(a)	may enter any land, and	5
		(b)	may take and use any extractive material from any land, and	6
		(c)	may take and use any extractive material from the bed of the Hunter River, and	7
		(d)	may lop, cut down or remove any tree within 40 metres of the Hunter River.	9 10
252	Res	trictio	n on use of lands adjacent to levee banks	11
	(1)		evee bank has been constructed adjacent to the Hunter River, the ster may, by notice in writing to:	12 13
		(a)	the landholder of the land on which the levee bank has been constructed, or	14 15
		(b)	the landholder of any land adjacent to the levee bank, or	16
		(c)	the landholder of the whole or any part of the land lying between the levee bank and the Hunter River,	17 18
		or lar stabil	se such conditions and restrictions on the use of such levee bank and as the Minister considers necessary or desirable to ensure the lity of the levee bank and in the interests of flood prevention or action within the Hunter Valley generally.	19 20 21 22
	(2)		Minister may in the like manner revoke or vary any notice given rsuance of this section.	23 24
	(3)		y landholder to whom such a notice has been given contravenes condition or restriction imposed by the notice, the Minister:	25 26
		(a)	may authorise any person to enter the land to which such notice relates, and to take such remedial measures on that land as the Minister considers necessary, and	27 28 29
		(b)	may recover any cost incurred in so doing from the landholder in any court of competent jurisdiction as a debt due and owing to the Crown	30 31 32

Chapter 5 Part 2 Division 2		Hunter Valley flood mitigation works		
	(4) If under this section conditions or restrictions on the use of any levee bank or land are imposed by the Minister, the landholder of the levee bank or land is entitled to claim and be paid compensation for any loss sustained by reason of those conditions or restrictions.		1 2 3 4	
253	Cor	struction of	fences, structures and flood works	5
	(1)	A person m	ust not:	6
		• •	truct any building, fence or structure in, on, or adjacent to, ee bank, or	7 8
		(b) cons	truct a flood work on a floodplain,	9
		except with	the consent of the Minister.	10
			penalty: 2,500 penalty units (in the case of a corporation) nalty units (in any other case).	11 12
	(2)	The Minist conditions.	er's consent may be given unconditionally or subject to	13 14
	(3)	An authoris	ed officer:	15
		has	enter any lands on which any building, fence or flood work been constructed otherwise than in accordance with the ister's consent, and	16 17 18
		the b	take such measures as are necessary to demolish or remove building, fence or flood work or to render the flood work fective.	19 20 21
	(4)		ncurred by an authorised officer under this section are from the landholder as a debt in a court of competent	22 23 24
	(5)		er may, by notice published in the Gazette, exclude any a floodplain.	25 26
	(6)	floodplain section 16	on, <i>floodplain</i> means any lands declared to be within the of the Hunter River by a proclamation in force under of the former <i>Hunter Valley Flood Mitigation Act 1956</i> , ands excluded from the floodplain by a notice published ection.	27 28 29 30 31

Power	s of Mi	inister	Division 2	
254	Lop	ping (of trees	
			the opinion of the Minister any tree on the bank of the Hunter or or on land adjoining any such bank:	
		(a)	constitutes a threat to any scheme for flood prevention or mitigation that the Minister has carried out, or has planned or is planning to carry out, or to any part of any such scheme, or	
		(b)	interferes with or impedes or tends to interfere with or impede the free flow of the Hunter River or otherwise detrimentally affects or is likely to detrimentally affect such flow or the stability of the banks of the Hunter River,	
			Minister may authorise any person to enter the land on which such is and to lop, cut down or remove such tree.	
Divis	sion 3	3	Finance	
255	Acc	ounts	s to be kept in Special Deposits Account	
		Treas	re is to be established in the Special Deposits Account in the sury an account to be called the "Hunter Valley Flood Mitigation punt".	
256	Allo	cation	n of money	
	(1)	acco	in the Hunter Valley Flood Mitigation Account is to be kept an unt to be called the "Hunter Valley Flood Mitigation Working bunt", in this Part referred to as the <i>Working Account</i> .	
	(2)	Mana whice deter	amount paid to the Minister by the Hunter Catchment agement Trust is to be paid into the Working Account, from the account may be made, in such manner as the Treasurer mines, any payments that the Minister is required to make under or the purposes of this Part.	
257	Cos	t of w	vorks generally	
	(1)	main Catcl	section applies to all flood works that are constructed or trained by the Minister, other than those for which the Hunter hment Management Trust is liable to make contribution under the other provision of this Division.	

Hunter Valley flood mitigation works

Public works

Clause 254

Chapter 5 Part 2

Clause 257 Chapter 5 Part 2 Division 3		Water Management Bill 2000 Public works Hunter Valley flood mitigation works Finance		
	(2)	quart	Hunter Catchment Management Trust is liable to contribute one- ter of the cost to the Minister of constructing and maintaining any I work to which this section applies.	1 2 3
258	Cos	st of w	orks to protect towns and villages	4
	(1)	the N flood the c	section applies to all flood works constructed or maintained by Minister for the purpose of protecting a town or village from ling, being works that have been constructed at the request or with oncurrence of the local council within whose area the town or ge is situated.	5 6 7 8 9
	(2)		ne cost to the Minister of constructing and maintaining any flood at to which this section applies:	10 11
		(a)	the local council is liable to contribute such percentage (not exceeding 12%) as the Minister may determine, and	12 13
		(b)	the Hunter Catchment Management Trust is liable to contribute one-quarter of the balance.	14 15
259	Cos	st of p	rotective works	16
	(1)		he cost to the Minister of constructing and maintaining any active works:	17 18
		(a)	the public authority whose works are protected by the protective works is liable to contribute such percentage as the Minister may determine, and	19 20 21
		(b)	the Hunter Catchment Management Trust is liable to contribute one-quarter of the balance.	22 23
	(2)	main	Minister may pay to any public authority that constructs or tains any protective works such amount as the Minister considers opriate towards the cost to the public authority of so doing.	24 25 26
	(3)	Mini	Hunter Catchment Management Trust is liable to pay to the ster one-quarter of any amount paid to a public authority under ection (2).	27 28 29

Water Management Bill 2000	Clause 260
Public works	Chapter 5
Hunter Valley flood mitigation works	Part 2
Finance	Division 3

260			contribution to works program by Hunter Catchment ent Trust	1
	(1)		Hunter Catchment Management Trust is liable to pay to the ster in each financial year:	3
		(a)	one-quarter of the estimated cost of constructing any flood works that the Minister proposes to construct during that financial year, and	5 6
		(b)	one-quarter of the estimated cost of maintaining any flood works that the Minister proposes to maintain during that financial year.	8 9 10
	(2)	the p	following amounts are to be deducted from the estimated costs for surpose of calculating the Hunter Catchment Management Trust's ity under this section:	11 12 13
		(a)	any contribution to those costs for which a local council will be liable under this Division,	14 15
		(b)	any amount that the Trust has paid to the Minister under this section during the previous financial year but that remained unexpended at the end of that year.	16 17 18
	(3)	Any	estimates are to be made by the Minister whose decision is final.	19
261		nsultat gram	ion with Hunter Catchment Management Trust as to works	20 21
	(1)	Before the beginning of each financial year or as soon as practicable thereafter, the Minister must submit to the Hunter Catchment Management Trust a program of the flood works that the Minister proposes to construct or maintain during that year.		22 23 24 25
	(2)		a program must be accompanied by a statement setting out the wing particulars:	26 27
		(a)	the estimated cost of constructing any work,	28
		(b)	the estimated annual cost of maintaining any work,	29
		(c)	the amount for which the Hunter Catchment Management Trust will be liable,	30 31
		(d)	such other particulars as may be prescribed by the regulations.	32

Clause 262 Water Management Bill 200

Chapter 5 Public works

Part 2 Hunter Valley flood mitigation works

Division 3 Finance

262 Consultation with local council as to works program

(1) This section applies to all flood works constructed or maintained by the Minister for the purpose of protecting a town or village from flooding, being works that have been constructed at the request or with the concurrence of the local council within whose area the town or village is situated. 1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

29

30

31

- (2) Before the beginning of each financial year or as soon as practicable thereafter, the Minister must submit to the local council concerned a program of the flood works that the Minister proposes to construct or maintain during that year.
- (3) Such a program must be accompanied by a statement setting out the following particulars:
 - (a) the estimated cost of constructing any work,
 - (b) the estimated annual cost of maintaining any work,
 - (c) the percentage that the Minister has determined that the council should contribute to the cost of the construction and maintenance of the work,
 - (d) the amount for which the council will be liable,
 - (e) such other particulars as may be prescribed by the regulations.
- (4) Any estimates are to be made by the Minister whose decision is final.
- (5) When the work has been completed, the Minister must notify the local council of the actual cost of the work.
- (6) If the amount paid by the local council in respect of the work is more than the amount for which the council is liable, the Minister must repay to the council the amount of such excess.
- (7) If the amount paid by the local council in respect of the work is less than the amount for which the council is liable, the council must pay to the Minister the amount of the shortfall.
- (8) Any amount paid to the Minister by a local council under this section is to be paid into the Hunter Valley Flood Mitigation Working Account.

Water Management Bill 2000	Clause 263
Public works	Chapter 5
Hunter Valley flood mitigation works	Part 2
Finance	Division 3

263	Payment of contribution towards maintenance by a local council				
	(1)	A local council that is liable to contribute to the cost of maintenance of any flood work must pay to the Minister in each financial year the same percentage of the estimated cost of maintenance as the council	3		
		is liable to contribute to the actual cost of maintenance of that work.	5		
	(2)	Z ,	6		
		respect of any financial year there is to be deducted any amount that the council has paid to the Minister in respect of any flood work during	7		
		any previous financial year and that remained unexpended at the end	9		
		of the immediately preceding financial year.	10		
	(3)	Any estimate is to be made by the Minister whose decision is final.	11		
264	Cor	sultation with local council as to maintenance program	12		
	(1)	Before the beginning of each financial year, the Minister must furnish the local council with a statement setting out the following particulars:	13 14		
		(a) the amount that the council will be liable to pay under this Division for that financial year,	15 16		
		(b) such other particulars as may be prescribed by the regulations.	17		
	(2)	The local council must pay to the Minister the amount referred to in subsection (1) (a).	18 19		
	(3)	Any such amount is to be paid into the Hunter Valley Flood Mitigation Working Account.	20 21		
265	Urg	ent maintenance works	22		
	(1)	If in the opinion of the Minister it becomes necessary to carry out	23		
		urgent maintenance on any flood work, the Minister may carry out the	24		
		maintenance, despite maintenance not having been included in any program of works or any statement furnished to a local council.	25 26		
	(2)				
	(2)	When the maintenance has been completed, the Minister must determine the total amount paid in respect of such maintenance and	27 28		
		notify any local council affected by the work and the Hunter	29		
		Catchment Management Trust of the respective amounts that they are	30		
		required to contribute in accordance with this Division.	31		
	(3)		32		
		Hunter Valley Flood Mitigation Working Account	33		

Part 2 Division 3			Hunter Valley flood mitigation works	
266	Pay	ments	s to be duly made	1
		Any	amount that is payable to the Minister under this Division:	2
		(a)	by the Hunter Catchment Management Trust, or	3
		(b)	by a local council, or	4
		(c)	by any other public authority,	5
			yable at such times, and in such instalments, as the Minister may mine.	6 7
Divis	sion 4	4	Miscellaneous	8
267		rks to ister	protect public and local government works to be approved by	9 10
	(1)		blic authority must not construct any protective works unless the	11
		Mini work	ster has approved of the site, nature, dimensions and design of the	12 13
	(2)			
	(2)		ving approval, the Minister may require such amendment of the nature, dimensions or design of the protective works as the	14 15
		Mini	ster considers necessary, and the public authority must comply	16
		with	that requirement accordingly.	17
	(3)		Minister may by notice in writing require the public authority by	18
			th any protective works have been constructed to take specified sures for the proper upkeep, preservation and maintenance of the	19 20
			ective works.	21
	(4)	The	public authority must comply with the requirements specified in	22
	` ′		otice in such manner and within such time as is so specified.	23
	(5)	If a p Mini	public authority fails to comply with the terms of the notice, the ster:	24 25
		(a)	may authorise any person to enter the land on which the	26
			protective works are located and carry out the measures specified in such notice, and	27
		(b)	may recover any cost incurred in so doing from the public	28
		(b)	authority in any court of competent jurisdiction as a debt due to	29 30
			the Crown.	31

Chapter 5

Water Management Bill 2000

Public works

Water Management Bill 2000	Clause 268
Public works	Chapter 5
Hunter Valley flood mitigation works	Part 2
Miscellaneous	Division 4

268	Мај	or floc	odgates to be maintained and operated by local councils	1
	(1)	prote small	the opinion of the Minister any floodgate operates for the ction of lands other than lands owned by one landholder or a localised group of landholders, the Minister may by notice in ng to the local council declare the floodgate to be a major gate.	2 3 4 5
	(2)	It is t	he duty of the local council:	7
		(a)	to keep any major floodgate at all times in a good and proper state of repair and working order, and	9
		(b)	to maintain and operate it in accordance with any directions that may from time to time be given by the Minister.	10 11
	(3)	If the	e local council fails to perform that duty, the Minister:	12
		(a)	may authorise any person to enter the land on which the major floodgate is located and to take such measures as to the maintenance, operation or repair of the floodgate as the Minister considers necessary, and	13 14 15
		(b)	may recover any cost incurred in so doing from the council in any court of competent jurisdiction as a debt due to the Crown.	17 18
269	Mai	ntenaı	nce of minor floodgates	19
	(1)		is section, <i>minor floodgate</i> means any floodgate that is not a r floodgate.	20 21
	(2)	mino	the duty of each landholder for the protection of whose land a r floodgate operates to keep the floodgate at all times in a good proper state of repair and working order.	22 23 24
	(3)	notic main	y such landholder fails to perform that duty, the Minister may by e in writing direct the landholder to take such measures as to the tenance, operation or repair of the floodgate as may be specified e notice.	25 26 27 28
	(4)	If any	y landholder fails to comply with the terms of the notice, the ster:	29 30
		(a)	may authorise any person to enter the land on which the minor floodgate is located and to take such measures as to the maintenance, operation or repair of the floodgate as the Minister considers necessary and	31 32 33

Clause 2 Chapter Part 2 Division	5	Water Management Bill 2000 Public works Hunter Valley flood mitigation works Miscellaneous	
		(b) may recover any cost incurred in so doing from the landholder in any court of competent jurisdiction as a debt due to the Crown.	1 2 3
((5)	In the case of multiple landholders, the liability imposed on them under this section is joint and several but, as between themselves, each is liable only for such share of the liability as bears the same proportion to the total liability as the area of the person's land for the protection of which the floodgate operates bears to the total area of land for the protection of which the floodgate operates.	4 5 6 7 8 9
((6)	If any landholder pays to the Minister more than the person's share of the total liability the landholder may recover the excess by way of contribution from the others.	10 11 12
270	Арр	lication of Public Works Act 1912	13
		Sections 91–95 of the <i>Public Works Act 1912</i> do not apply to any works constructed under this Part.	14 15
271 E	Evic	dence of cost of works	16
		If the Minister is empowered to recover any cost incurred in carrying out any work or measure, a certificate by the Minister of the amount of such cost is evidence of that cost.	17 18 19

Part	3 I	Lowbidgee flood control and irrigation works	1
272	App	plication	2
		This Part applies to and in respect of the Lowbidgee flood control and irrigation district, as constituted under Part 7 of the former <i>Water Act</i>	3 4
		1912 immediately before the repeal of that Part.	5
273	Def	initions	6
		In this Part:	7
		<i>flood irrigation</i> means the irrigation of lands by flooding with water diverted by gravitation or overflowing from a river or lake.	8 9
		Lowbidgee district means the Lowbidgee flood control and irrigation	10
		district, as constituted under Part 7 of the former <i>Water Act 1912</i> immediately before the repeal of that Part.	11 12
		Lowbidgee works mean the works situated on the Murrumbidgee	13
		river, being works constructed, operated and maintained by the Minister for the purpose of providing flood control and flood irrigation	14 15
		to land within the Lowbidgee district.	16
274	Sup	oply of water	17
	(1)	The Minister is authorised to operate the Lowbidgee works for the	18
		purpose of providing flood control and flood irrigation to land within the Lowbidgee district.	19 20
	(2)		
	(2)	The control of works within the Lowbidgee district, the arrangements for distribution of water for flood irrigation, and the order of supply or	21 22
		diversion of water for flood irrigation, and the order of supply of	23
		the Minister.	24
	(3)		25
		under any restraint nor is compensation payable to any person as a consequence of the flooding of the person's land.	26 27
	(4)	Nothing in this Part requires the Minister to provide flood irrigation to any land or person in the Lowbidgee district.	28 29

Clause 275 Water Management Bill 2000

Chapter 5 Public works

Part 3 Lowbidgee flood control and irrigation works

275	Fixi	ng of	rates	1		
	(1)		Minister must, in respect of each year commencing 1 July, fix the per hectare to be paid in respect of the lands within the Lowbidgee ct:	2 3 4		
		(a)	for which provision for flood control and irrigation is made by the Lowbidgee works, or	5		
		(b)	for which provision for flood control and irrigation is made by the Lowbidgee works and by works of the landholder,	7 8		
		and rate.	must in respect of each holding assess the total amount of such	9 10		
	(2)	The r	rate so fixed and assessed must be levied and paid as prescribed.	11		
276	Rates					
	(1)		s for a particular holding commence to be payable on a date to be mined by the Minister, being a date that is not earlier than the date hich:	13 14 15		
		(a)	provision for flood control or irrigation, as the case may be, in respect of that holding is made by the Lowbidgee works, or	16 17		
		(b)	provision for flood control and irrigation is made by the Lowbidgee works and by works of the landholder.	18 19		
	(2)	•	y land within the Lowbidgee district reverts to the Crown during year in which rates are payable:	20 21		
		(a)	the landholder, as regards that land, is liable only for payment of such part of the rates as are proportionate to the portion of the year for which the land was held by the landholder, and	22 23 24		
		(b)	any excess payment by the landholder must be refunded to the landholder.	25 26		
	(3)	asses lack of of the	the event of the Minister at any time finding an error in the assent of a rate in respect of any holding or landholder through of knowledge of the name of the landholder, or the area or extent e holding, or miscalculation, the Minister may at any time rest the rate in respect of any holding or landholder affected.	27 28 29 30 31		

277	Ber	nefit to	be taken into account	1
	(1)	The 1	rates in the Lowbidgee district may be determined or varied	2
			ng regard to the benefit which, in the opinion of the Minister, is	3
			ed by the landholder of the land:	4
		(a)	in respect of which works for flood control and irrigation have been provided, or	5 6
		(b)	in respect of which provision for flood control and irrigation has been made by the Lowbidgee works and by works of the landholder.	7 8 9
	(2)	provi	Minister may exempt from rating the lands which, without the sion of works for flood control and irrigation, would not be ed at times of maximum floods.	10 11 12
	(3)		letermination of the Minister as to whether and to what extent any would be so flooded is final.	13 14
278	Reg	gulatio	ns	15
			regulations may make provision for or with respect to the wing matters:	16 17
		(a)	the prevention of pollution of water conserved or distributed in the Lowbidgee works constructed or used under this Part, or flowing in rivers or within lakes affected by the Lowbidgee works constructed or used under this Part,	18 19 20 21
		(b)	the prevention of injury to such the Lowbidgee works or any works used in connection with those works,	22 23
		(c)	the diversion of water from rivers and lakes for flood irrigation and for the control of works provided for regulating flood irrigation,	24 25 26
		(d)	the fixing, assessing and levying of rates,	27
		(e)	the prevention of the waste of water,	28
		(f)	the forms of notices to be given under this Part and the manner of and periods for giving such notices.	29 30

Part 1 Division 1			Major utilities Preliminary	
Cha	pte	r 6	Public utilities	1
Part	:1 I	Majo	or utilities	2
Divis	ion '	1	Preliminary	3
279	Maj	or uti	lities	4
	(1)		Part applies to each body referred to in Schedule 2 (referred to as <i>ujor utility</i>).	5 6
	(2)	The	Governor may by proclamation amend Schedule 2 so as:	7
		(a)	to add the corporate name of any body that is to become a major utility for the purposes of this Part, or	8 9
		(b)	to vary the name of any major utility as a consequence of any change in its corporate name, or	10 11
		(c)	to omit the corporate name of any body that has ceased to exist.	12
280	Rec	quiren	nents for access licences and approvals	13
	(1)	than	ning in this Part authorises a major utility to take water otherwise in accordance with an access licence and water supply work oval or water authorisation held by the utility.	14 15 16
	(2)	than	ning in this Part authorises a major utility to use water otherwise in accordance with a water use approval or water authorisation by the utility.	17 18 19
Divis	ion 2	2	Water authorisations	20
281	Wat	ter au	thorisation	21
	(1)	The	Minister:	22
		(a)	on the application of a major utility, and	23
		(b)	on payment of the appropriate authorisation fee,	24
		may	grant to the major utility a water authorisation.	25

Chapter 6

Water Management Bill 2000

Public utilities

Water	autho	risation	Division 2	
	(2)	havin	appropriate authorisation fee is to be determined by the Minister, ag regard to the costs incurred by the Minister in dealing with the cation for the authorisation.	1 2 3
282	Cor	ndition	s of water authorisation	4
	(1)	A wa	ater authorisation is subject to:	5
		(a)	such conditions as are required to be imposed on the authorisation by the relevant management plan or a Minister's plan (<i>mandatory conditions</i>), and	6 7 8
		(b)	such other conditions as the Minister may from time to time impose on the authorisation (<i>discretionary conditions</i>).	9 10
	(2)		andatory condition prevails over a discretionary condition to the at of any inconsistency between them.	11 12
		term	If a management plan or Minister's plan is replaced or amended during the of a water authorisation, the mandatory conditions applying to the risation may vary.	13 14 15
283	lmp	ositio	n of conditions after water authorisation is granted	16
	(1)		retionary conditions may not be imposed on a water authorisation it has been granted unless the Minister:	17 18
		(a)	has given written notice to the holder of the authorisation that the Minister proposes to impose such conditions, and	19 20
		(b)	has given the holder of the authorisation a reasonable opportunity to make submissions to the Minister with respect to the proposed conditions, and	21 22 23
		(c)	has taken any such submissions into consideration.	24
	(2)		ection (1) does not apply to conditions imposed on a water orisation at the request of the holder of the authorisation.	25 26
	(3)	impo	Minister must cause written notice of any discretionary conditions used on a water authorisation after the time it is granted to be and on the holder of the authorisation.	27 28 29
	(4)	the date	scretionary condition referred to in subsection (3) takes effect on ay on which the notice referred to in that subsection is served on older of the water authorisation or on such later day as may be fied in the notice in that regard.	30 31 32 33

Public utilities

Major utilities

Clause 281

Chapter 6 Part 1

Part 1 Divisio	n 2	Major utilities Water authorisations		
284	Rev	ocatio	on of conditions	1
		whic	Minister may at any time revoke any discretionary conditions to the a water authorisation is subject, either on the Minister's own on or on the application of the holder of the authorisation.	2 3 4
285	Dur	ation	of water authorisation	5
	(1)	A wa	ater authorisation has effect for:	6
		(a)	such period (not exceeding 20 years) as is specified in the authorisation, or	7 8
		(b)	if that period is extended under this section, for that extended period.	9 10
	(2)	curre the N	the end of each 5 year period following the date on which its ent water authorisation was granted, a major utility may apply to Minister for an extension of the period of the authorisation for a per 5 years.	11 12 13 14
	(3)	must	etermining whether or not to grant such an extension, the Minister thave regard to the review that the Minister has conducted under Part in relation to that 5 year period.	15 16 17
286	Effe	ect of	water authorisation	18
		A wa	ater authorisation authorises its holder:	19
		(a)	to take and use water from any water source, and	20
		(b)	to construct or use a water management work,	21
		subje Part.	ect to the conditions of the authorisation and the provisions of this	22 23
287	Rev	view o	f water authorisation	24
	(1)	The	activities of the holder of a water authorisation:	25
		(a)	may be reviewed at any time, and	26
		(b)	must be reviewed:	27
			(i) before the end of the 6 month period following the date	28
			on which its first water authorisation was granted, and (ii) before the end of each 5 year period following the date	29 30

on which its current water authorisation was granted.

31

Clause 284

Chapter 6

Water Management Bill 2000

Public utilities

Public Major Water	utilitie		Chapter 6 Part 1 Division 2	
	(2)		the purpose of conducting such a review, the Minister must cause to be published:	1 2
		(a)	in the Gazette, and	3
		(b)	in a newspaper circulating throughout New South Wales,	4
			ing written submissions from interested persons in relation to the rities of the holder of the authorisation during the period under ew.	5 6 7
288	Wat	ter ma	anagement charges	8
	(1)	A m Mini	najor utility is liable to pay water management charges to the sister.	9 10
	(2)		charges so payable are to be fixed by the Minister, having regard e Minister's costs in relation to:	11 12
		(a)	the conduct of reviews under this Part, and	13
		(b)	the administration of this Part with respect to the major utility, and	14 15
		(c)	the monitoring of the extraction of water from, and the flow of water along, any river, estuary or lake that is affected by the work and activities of the major utility, and	16 17 18
		(d)	the exercise of any other water resource management function with respect to the major utility.	19 20
Divis	sion (3	Directions to major utilities	21
289	Dire	ection	s for remedial work	22
			Minister may, by order in writing, direct a major utility to take measures as are specified in the order:	23 24
		(a)	to protect a water source from any adverse effects of the utility's activities, or	25 26
		(b)	to restore or enhance any water source that has been adversely affected by the utility's activities, or	27 28
		(c)	to prevent the wastage or pollution of water from the utility's water supply works, or	29 30
		(d)	to ensure that the utility's activities are in compliance with the conditions of its water authorisation.	31 32

Division 3		Directions to major utilities	
290	Ren	nedial measures may be taken by Minister	1
	(1)	If a major utility fails to take the measures specified in a direction under this Division, the Minister may authorise any other person to take those measures.	2 3 4
	(2)	A person so authorised has and may exercise such of the functions of the major utility as are necessary to enable the person to take those measures.	5 6 7
	(3)	The amount of any costs and expenses incurred by the Minister as a result of the taking of those measures is recoverable in a court of competent jurisdiction as a debt due to the Crown from the major utility.	8 9 10 11
291	Lan	d and Environment Court may grant injunctions	12
		On the application of the Minister, the Land and Environment Court may grant an injunction directing a major utility to comply with a direction under this Division.	13 14 15
292	Cor	npensation for loss of water	16
	(1)	This section applies to a holder of an access licence in respect of which water that would otherwise be available ceases to be available as a result of a major utility's failure to comply with a direction under this Division in respect of which the Land and Environment Court has granted an injunction.	17 18 19 20 21
	(2)	A person to whom this section applies may commence proceedings in any court of competent jurisdiction for the recovery from the major utility of the amount of any loss or damage suffered by the person as a result of the major utility's failure to comply with the direction.	22 23 24 25
	(3)	Proceedings under this section may be commenced at any time up to 3 years after the date on which the Land and Environment Court's injunction took effect.	26 27 28
293	Offe	ences with respect to water authorisations	29
	(1)	A major utility that uses a water supply work that is not the subject of a water authorisation is guilty of an offence.	30 31
		Maximum penalty: 2,500 penalty units.	32

Chapter 6

Part 1

Water Management Bill 2000

Public utilities

Major utilities

Major Directi			utilities Part 1 Division 3	
	(2)		ajor utility does not commit an offence under this section with ect to a water supply work the subject of a water supply work oval.	1 2 3
294	Civ	il pena	alties	4
	(1)	If sat	tisfied that a major utility:	5
		(a)	has contravened the conditions of its water authorisation, or	6
		(b)	has failed to comply with a direction under this Division,	7
		exce pena	Minister may order the utility to pay to the Minister a penalty not eding \$500,000 and, in the case of a continuing offence, a further lity not exceeding \$20,000 for each day for which the ravention or failure continues.	8 9 10 11
	(2)	Befo	ore making an order under this section, the Minister:	12
		(a)	must cause written notice of its intention to make the order to be given to the major utility, and	13 14
		(b)	must give the major utility a reasonable opportunity to make written submissions to the Minister in relation to the proposed order, and	15 16 17
		(c)	must have regard to any submission that is duly made.	18
Divis	ion 4	4	Appeals	19
295	Арр	oeals t	to Land and Environment Court	20
	(1)		appeal lies to the Land and Environment Court against any of the wing decisions made by the Minister:	21 22
		(a)	a decision refusing to grant a water authorisation,	23
		(b)	a decision imposing a discretionary condition on a water authorisation,	24 25
		(c)	a decision fixing the term of a water authorisation,	26
		(d)	a decision refusing to extend the term of a water authorisation,	27
		(e)	a decision to give a direction to a major utility under Division 3,	28 29
		(f)	a decision to make an order requiring the major utility to pay a penalty.	30 31

Public utilities

Clause 293

Chapter 6

Chapter 6 Public utilities Part 1 Major utilities Division 4 Appeals (2) An appeal is to be made in accordance with rules of court, but may not 1 be made more than 28 days after the date on which the decision was 2 made. 3 (3) The lodging of an appeal does not operate to stay action on the 4 decision appealed against, except to the extent that the Land and 5 Environment Court otherwise directs.

6

Water Management Bill 2000

Preliminary	Division 1
Water supply authorities	Part 2
Public utilities	Chapter 6
Water Management Bill 2000	Clause 296

Part	t 2 Wa	ater supply authorities	1
Divis	sion 1	Preliminary	2
296	Definit	tions	3
	In	n this Part:	4
		rea of operations of a water supply authority means the area of perations prescribed by the regulations in relation to that authority.	5
		uthorised person means an employee or other person acting on ehalf of a water supply authority.	7 8
	cl	harging year, in relation to a water supply authority, means:	9
	(a	the period of 12 months declared by an order in force under Division 6 to be the charging year for the water supply authority, or	10 11 12
	(b	if the charging year is changed under that Division by a further order, the period between the end of one charging year and the beginning of the next.	13 14 15
		evelopment area means an area of land declared by an order in force nder Division 6 to be a development area.	1 <i>6</i>
		rainage area means an area of land declared by an order in force nder Division 6 to be a drainage area.	18 19
		<i>loodplain</i> means an area of land declared by an order in force under Division 6 to be a floodplain.	20 21
	m	neter includes any measuring device.	22
		iver management area means an area of land declared by an order in orce under Division 6 to be a river management area.	23 24
	se	ervice charge means any of the following:	25
	(a	a) a water service charge,	26
	(b	a sewerage service charge,	27
	(c	e) a drainage service charge,	28
	(d	d) a loan service charge,	29
	(e	e) a developmental works service charge,	30
	(f	a flood mitigation service charge.	31

Chapter 6 Part 2 Division 1		Public utilities Water supply authorities Preliminary			
		(g)	a river management service charge, or	1	
		(h)	a special industry service charge,	2	
		that 1	may be levied under Division 6.	3	
		special area means an area of land declared under Division 4 to be a special area.special industry means an industry declared by an order in force under Division 6 to be a special industry.			
			<i>tory body</i> means a corporation that is incorporated by or under an other than:	9	
		(a)	a company within the meaning of the Corporations Law, and	10	
		(b)	a society within the meaning of the Co-operation Act 1923 or	11	
			a co-operative within the meaning of the <i>Co-operatives Act</i> 1992, and	12 13	
		(c)	an incorporated association within the meaning of the <i>Associations Incorporation Act 1984</i> .	14 15	
		water	r management work includes a sewage work.	16	
297	Rec	uirem	nents for access licences and approvals	17	
		other	ing in this Part authorises a water supply authority to take water wise than in accordance with an access licence and water supply approval or water authorisation held by the authority.	18 19 20	
Divis	ion 2	2	Water supply authorities	21	
298	Wat	er su	pply authorities	22	
	(1)	The o	corporations listed in Schedule 3 are water supply authorities.	23	
	(2)		rporation listed in Part 1 of Schedule 3 is, for the purposes of any a statutory body representing the Crown.	24 25	
299	Constitution of water supply authorities			26	
	(1)	The	Governor may, by proclamation published in the Gazette:	27	
		(a)	name a water supply authority proposed to be constituted under this section, and	28 29	

Public utilities Chapter 6 Water supply authorities Part 2						
	er supply authorities Division 2					
		(b)	specify the number of members, the qualification for each member and whether or not the member is a full-time member or a part-time member, and	1 2 3		
		(c)	amend Schedule 3 by adding the name of the water supply authority to Part 1 of that Schedule.	4 5		
	(2)		members are to be appointed by the Governor and, on their bintment, the water supply authority is constituted as a corporation.	6 7		
	(3)		corporate name for the water supply authority is the name of the er supply authority stated in the proclamation.	8		
	(4)	author proc	a change of area of operations or functions for a water supply ority listed in Part 1 of Schedule 3, the Governor may, by lamation published in the Gazette, do either or both of the owing:	10 11 12 13		
		(a)	change the number of part-time members of the water supply authority,	14 15		
		(b)	change the qualifications for part-time members of the water supply authority.	16 17		
	(5)		nange referred to in subsection (4) does not affect the identity of water supply authority.	18 19		
300	Sta	Statutory body may be water supply authority				
	(1)	cons	Governor may, by proclamation published in the Gazette with the tent of the Minister administering the Act by or under which a story body is constituted, amend Schedule 3 by adding the name of statutory body to Part 2 of that Schedule.	21 22 23 24		
	(2)		he addition of the name of a statutory body to Part 2 of Schedule becomes a water supply authority but still has its other functions.	25 26		
301	Provisions relating to constitution and procedure of water supply authorities			27 28		
	(1)		1 of Schedule 5 has effect with respect to the members of the er supply authorities listed in Part 1 of Schedule 3.	29 30		
	(2)	Part	2 of Schedule 5 has effect with respect to the Broken Hill Water	31		

(3) Part 3 of Schedule 5 has effect with respect to the Cobar Water Board.

Water Management Bill 2000

Board.

32

33

Chapter 6 Part 2 Division 2		Water supply authorities		
	(4)	prov	s 2 and 3 of Schedule 5 do not affect the operation of the other isions of this Part in relation to the water supply authorities to the those Parts relate.	1 2 3
302	Are	a of o	perations	4
	(1)		area of operations of a water supply authority is such as may be cribed by the regulations for the water supply authority.	5 6
	(2)	If a regulation so provides, a water supply authority has, and may exercise, in its area of operations or a specified part of its area of operations:		7 8 9
		(a)	only such of the functions of a water supply authority as are specified in the regulation, or	10 11
		(b)	all functions of a water supply authority other than functions specified in the regulation.	12 13
	(3)		egulation made for the purposes of subsection (2) may make ision for or with respect to:	14 15
		(a)	matters affecting revenue, assets, rights, liabilities and employees of a water supply authority, and	16 17
		(b)	procedures to be adopted for the purposes of paragraph (a).	18
	(4)		in the area of operations of a water supply authority, a function or this Part may be exercised only by the water supply authority ss:	19 20 21
		(a)	the function is exercisable by the Minister, or	22
		(b)	the function is exercised by a catchment management trust under the <i>Catchment Management Act 1989</i> , or	23 24
		(c)	with the consent of the Minister, the function is exercised by another Minister, by the delegate of another Minister or by a statutory body, or	25 26 27
		(d)	this Part provides otherwise.	28
	(5)		the consent of the Minister, a water supply authority may exercise action outside its area of operations.	29 30
	(6)	of o	l a regulation is made for the purposes of subsection (1), the area perations of a water supply authority is the same as it was ediately before the commencement of this Part.	31 32 33

Water Management Bill 2000

Water supply authorities Water supply authorities		-			
303	Em	Employees			
			ater supply authority may employ such persons as are necessary to le it to exercise its functions.	2 3	
304	Delegation				
			ater supply authority may delegate to a person the exercise of any s functions, other than this power of delegation.	5 6	
Divis	Division 3 Functions of water supply authorities		7		
305	Functions of water supply authority				
	(1)	A wa	ater supply authority has the following functions:	9	
		(a)	to construct, maintain and operate water management works,	10	
		(b)	to conduct research, collect information and develop technology in relation to water management,	11 12	
		(c)	to do anything for the purpose of enabling the objects of this Act to be attained.	13 14	
	(2)		nter supply authority may exercise its functions within and beyond rea of operations.	15 16	
	(3)	cons	the duty of a water supply authority to exercise its functions sistently with the principles of ecologically sustainable elopment.	17 18 19	
306	Exercise of functions by a water supply authority				
	(1)	A wa	ater supply authority:	21	
		(a)	has and may exercise the functions conferred or imposed on it by or under this or any other Act, and	22 23	
		(b)	in the exercise of its functions (except in relation to the contents of a report or recommendation made by it) is subject to the control and direction of the Minister.	24 25 26	
	(2)	Part	function conferred or imposed on a water supply authority by this is inconsistent with a function conferred or imposed on the water oly authority by or under another Act, this Part prevails.	27 28 29	

Public utilities

Clause 303

Chapter 6

Division 3		F	functions of water supply authorities	
307	Cor	nmerc	cial operations	
		With	the approval of the Governor, a water supply authority may enter commercial operations with respect to:	
		(a)	any services developed in connection with the exercise of its functions, or	
		(b)	any products or by-products (including intellectual property) resulting from the exercise of its functions, or	
		(c)	such other matters as may be prescribed by the regulations.	
	(2)	or jo	the approval of the Governor, a water supply authority may form, in in forming, a company, partnership or trust for the purpose of cising its functions under this Part.	
308	Assistance to statutory body			
		with wate whic for th	ater supply authority may, at the request of a statutory body and the consent of the Minister, exercise any of the functions of the r supply authority in a part of its area of operations in respect of the statutory body may exercise different functions, even if, but he request, the water supply authority would not have exercised function.	
309	Entry on land to read meters or carry out works			
	(1)	A wa	ater supply authority may, by its employees and agents:	
		(a)	enter and occupy land and there exercise any of its functions, including the carrying out of any work on, below or above the surface of the land, and	
		(b)	divert water from, or alter the course of, a stream, and	
		(c)	impound or take water on, in or under the surface of any land.	
	(2)	empl (exce	power conferred by this section is sufficient authority for an loyee or agent to enter and occupy land or any part of a building ept an enclosed part occupied as a separate dwelling) in the cise of a water supply authority's functions during daylight for the ose of:	
		(a)	reading a meter, or	
		(b)	ascertaining whether trade waste is present or is being (or has recently been) discharged to a work of the authority, or	

Chapter 6

Part 2

Water Management Bill 2000

Water supply authorities

Public utilities

Water Management Bill 2000	Clause 309
Public utilities	Chapter 6
Water supply authorities	Part 2
Functions of water supply authorities	Division 3

	(c)	making a valuation,	1		
	unless the employee or agent is refused access by the lawful occupier of the land.				
(3)	A water supply authority may, in accordance with the regulations, attach a ventilating shaft, pipe or tube for a sewer to the wall of a building.				
(4)		ter supply authority may remove or use anything dug up or ed in the exercise of its powers under this section.	7 8		
(5)	A wat	er supply authority:	9		
	(a)	must do as little damage as practicable in exercising its powers under this section, and	10 11		
	(b)	must compensate all persons who suffer damage by the exercise of the powers,	12 13		
	in rela	tion to the land entered.	14		
(6)		Compensation may be made by reinstatement or repair, by construction of works or by payment.			
(7)	If a water supply authority provides a sewer, compensation is required only if:				
	(a)	the sewer causes injury to, or interference with, a building or other structure, or	19 20		
	(b)	a manhole or main ventilator is constructed on the land.	21		
(8)	A claim for compensation:		22		
	(a)	is ineffective unless made in writing not later than 6 months after the damage was suffered, and	23 24		
	(b)	in the absence of agreement on the compensation, must be dealt with as if it were a claim for compensation for the acquisition of land for public purposes under the <i>Public Works Act 1912</i> .	25 26 27		
Enti	ry on la	and for inspections	28		
(1)	A water supply authority may at any time, by its employees or agents, enter any land (except an enclosed part occupied as a separate dwelling) for the purpose of finding out:				
	(a)	the character and condition of the land or building, or	32		
	(b)	the condition or location of any water management work used in connection with the land or building, or	33 34		

310

no defective or improper work is discovered, and

35

(a)

Clause 310

Water Management Bill 2000

Functi	ons of	water	supply authorities	Division 3	
		(b)	no waste, misuse or undue consum the water supply authority is occur		1 2
		(c)	no offence against this Act is disco	vered,	3
			vater supply authority must make gooded by it for the purposes of the inspec	•	4 5
	(9)		do as little damage as practicable.	n, a water supply authority	6 7
311	Pov	ver to	break up roads		8
	(1)		ater supply authority may, on giving re y to be affected, open and break up:	easonable notice to persons	9 10
		(a)	the soil and pavement of a public r	oad or public reserve, and	11
		(b)	any sewer, drain or tunnel in or un reserve,	der a public road or public	12 13
		for th	ne purpose of exercising its functions		14
	(2)	road water power	statutory body having the control an or public reserve may, as prescribed by supply authority to comply with cers under subsection (1), including conce and removal of rubbish.	by the regulations, require a conditions in exercising its	15 16 17 18 19
	(3)	burst statu or pu	oublic road or public reserve is damaging of, a water supply authority's watery body having the control and manublic reserve may require the water supamage without delay.	ter main or sewer main, the agement of the public road	20 21 22 23 24
	(4)	If a v	vater supply authority fails:		25
	, ,	(a)	to comply with a condition in force	e under subsection (2), or	26
		(b)	to comply with a requirement under		27
		the c	tatutory body affected by the failure ost of doing so as a debt owed to the ly authority.		28 29 30

Water supply authorities

Public utilities

Clause 310

Chapter 6 Part 2

Division 3		Functions of water supply authorities		
312	Alte	Altering position of conduits		1
	(1)	If:		2
		(a)	a water supply authority, in order to exercise its functions, needs an alteration to be made in the position of a conduit owned by another person, and	3 4 5
		(b)	the alteration would not permanently damage the conduit or adversely affect its operation,	6 7
			vater supply authority may serve on the person a written notice complies with subsection (2).	8
	(2)	A no	tice must:	10
		(a)	specify the alteration needed, and	11
		(b)	require the alteration to be made within a reasonable time stated in the notice, and	12 13
		(c)	include an undertaking by the water supply authority to pay the reasonable cost of the alteration.	14 15
	(3)	autho the co	e alteration is not made as required by the notice, the water supply ority may make the alteration in such a manner as not to damage onduit permanently or adversely affect its operation on completion e alteration.	16 17 18 19
	(4)		is section, <i>conduit</i> means anything that is in or under a public and is used for the conveyance of a substance, energy or signals.	20 21
313	Obs	structi	on of works	22
	(1)	autho	person places a structure or other thing in or near a water supply prity's water management work in such a manner as to interfere the work, the water supply authority may:	23 24 25
		(a)	demolish and remove the structure or other thing, and	26
		(b)	repair the work, and	27
		(c)	recover the cost of doing so as a debt owing to the water supply authority by the person who placed the structure or other thing there.	28 29 30
	(2)	preve	ater supply authority may apply for and obtain an injunction to ent a structure or other thing being placed as referred to in ection (1).	31 32 33

Clause 312

Chapter 6

Part 2

Water Management Bill 2000

Water supply authorities

Public utilities

Functi	functions of water supply authorities Division 3					
314	Fine	ding s	ource of pollution of water sup	ply	1	
		If wa water likely	ater supplied by a water supply r supply authority may, after giving to be affected by its action, dig use of the pollution.	authority is being polluted, the ng reasonable notice to persons	2 3 4 5	
	(2)	respo	is found that a person given nonsible for the pollution, the waternses of:		6 7 8	
		(a)	the digging, examination, repair ground, and	and reinstatement of the broken	9 10	
		(b)	the repair of any damage cause to the property of the person.	d by the water supply authority	11 12	
	(3)	debt	nter supply authority may recover due to the water supply authority ollution.		13 14 15	
Divis	sion 4	4	Special areas		16	
315	Spe	ecial a	reas		17	
	(1)	publi	Governor may, on the recommen ished in the Gazette, declare an area special area for a water supply	ea of land described in the order	18 19 20	
	(2)	exerc	Minister may recommend an orderise of the State's water rights conder is made.		21 22 23	
316	Cro	wn lai	nd in special areas		24	
	(1)		on may not be taken under the <i>Cr</i> and within a special area for a wat	-	25 26	
		(a)	a water supply authority has gi	ven its approval in writing, and	27	
		(b)	any conditions to which the apwith.	pproval is subject are complied	28 29	
	(2)	of a	nter supply authority may, in a spettrustee appointed under Part 5 out being appointed as such.		30 31 32	

Water supply authorities

Public utilities

Clause 314

Chapter 6 Part 2

Division 4			Special areas	
317	Exe	rcise	of certain functions in special areas	1
	(1)	a wa this	ablic agency may not, in relation to land within a special area for ter supply authority, exercise functions other than functions under Part unless notice is first given, as prescribed by the regulations, e water supply authority.	2 3 4
	(2)		eceipt of such a notice, the water supply authority may make such esentations to the public agency as it thinks fit.	6
	(3)		ublic agency may not exercise functions contrary to any such esentations made by a water supply authority unless:	9
		(a)	at least 14 days' notice has been given to the water supply authority of the functions intended to be exercised, and	1(11
		(b)	that period has expired without the matter being referred to the Minister as a dispute for determination.	12 13
	(4)	In th	is section, <i>public agency</i> means:	14
		(a)	the Governor, or	15
		(b)	a Minister of the Crown, or	16
		(c)	a statutory body.	17
Divis	ion (5	Developer contributions to the construction of works	18 19
318	App	olicatio	on for certificate of compliance	20
	(1)	com	erson may apply to a water supply authority for a certificate of pliance for development carried out, or proposed to be carried out, in the water supply authority's area.	21 22 23
	(2)		application must be accompanied by such information as the lations may prescribe.	24 25
319		hority ompl	may impose certain requirements before granting certificate iance	26 27
	(1)		section applies to such kinds of development as are prescribed by egulations for the purposes of this section.	28 29

Water supply authorities

Public utilities

Clause 317 Chapter 6

Part 2

(1) A water supply authority must grant a certificate of compliance for

within 60 days after an application for the granting of such a

Water Management Bill 2000

authority.

development:

(a)

Granting of certificates of compliance

certificate is made, or

320

Clause 319

1

2

3

4

5

6

7

8

9

30

31

32

33

34

35

Clause 320		Water Management Bill 2000				
Chapte	er 6	P	ublic utilities			
Part 2		Water supply authorities				
Divisio	n 5	D	eveloper contributions to the construction of works			
		(b)	if, within that period, the water supply authority imposes a requirement on the applicant under this Division, as soon as it is satisfied that the requirement has been complied with.	1 2 3		
	(2)	Divis suppl	ter supply authority may be satisfied that a requirement under this ion has been complied with if the applicant lodges with the water y authority such security for compliance with the requirement as rater supply authority may approve.	4 5 6 7		
	(3)	certifi made Envir	water supply authority fails or refuses to give a compliance icate within the period of 60 days after an application is duly in that regard, the applicant may appeal to the Land and conment Court, within 12 months after the expiration of that d, against the failure or refusal.	8 9 10 11 12		
Divis	ion (ô	Finance	13		
321	Ord	ers foi	r purpose of service charges	14		
	(1)		Governor may, by order published in the Gazette, declare any land ibed in the order to be a development area for the purposes of this	15 16 17		
	(2)	descr	Minister may, by order published in the Gazette, declare any land ibed in the order to be a drainage area, floodplain or river gement area for the purposes of this Part.	18 19 20		
	(3)	indus	Minister may, by order published in the Gazette, declare any try specified in the order to be a special industry for the purposes s Part.	21 22 23		
	(4)		Minister may, by order published in the Gazette, declare a period months to be the charging year for a water supply authority.	24 25		
322	Cor	nectio	ons	26		
	(1)		water supply authority's water main or a sewer main becoming able for connection, the water supply authority may:	27 28		
		(a)	publish in the Gazette notice of its availability, and	29		
		(b)	give such other notices as may be prescribed.	30		

(2)		dholder of land to which a notice under subsection (1) (a) relates nes liable:	1 2
	(a)	in the case of a water main, to payment of water service charges after the expiration of 21 days from publication of the notice, or	3 4 5
	(b)	in the case of a sewer main, to payment of sewerage service charges after the expiration of 21 days from publication of the notice.	6 7 8
(3)	author subsec	andholder does not make a connection to the water supply rity's water main or sewer main before becoming liable under ction (2), the water supply authority may make the connection at pense of the landholder.	9 10 11 12
(4)	under recove	amount due to a water supply authority for a connection made subsection (3) is a charge on the land connected and may be ered as a debt owed to the water supply authority by the older of the land.	13 14 15 16
(5)	If a la	ndholder of land fails:	17
	(a)	to connect the land to a water supply authority's water main or sewer main before becoming liable under subsection (2), or	18 19
	(b)	to do any work that by this Part is required to be done by the landholder,	20 21
	in acc	cupier of the land may make the connection or do the work and, ordance with subsection (6), recover the cost from the landholder interest at the prescribed rate.	22 23 24
(6)		nount (including interest) recoverable under subsection (5) by an ier from a landholder may be recovered:	25 26
	(a)	by deducting it from any rent from time to time payable to the landholder by the occupier, or	27 28
	(b)	as a debt owed to the occupier by the landholder.	29
(7)	author as ma or wa	person desiring to connect premises with a water supply rity's water main or sewer main may (subject to such conditions y be imposed by law) open up the surface and soil of any road y, public or private, or any footpath or public reserve to the required to make the connection.	30 31 32 33 34

Clause 323	Water Management Bill 2000
Chapter 6	Public utilities
Part 2	Water supply authorities
Division 6	Finance

323	Aut	hority	may levy service charges and impose fees and other charges	1
	(1)		ater supply authority may, in accordance with this Part, levy the wing service charges on land within its area of operations:	2 3
		(a)	water service charges,	4
		(b)	sewerage service charges,	5
		(c)	drainage service charges,	6
		(d)	loan service charges,	7
		(e)	developmental works service charges,	8
		(f)	flood mitigation service charges,	9
		(g)	river management service charges,	10
		(h)	special industry service charges.	11
	(2)	A w	ater supply authority may, in accordance with the regulations,	12
	. ,		ose fees and charges for any service or thing supplied or provided	13
		by it	in the exercise of its functions under this Part.	14
324		d in r rges	respect of which a water supply authority may levy service	15 16
	(1)	A wa	ater supply authority may only levy water service charges on land:	17
		(a)	to which water is supplied, or	18
		(b)	to which, in the opinion of the water supply authority, it is reasonably practicable for water to be supplied,	19 20
		from	one of the water supply authority's water mains.	21
	(2)	A wa land:	ater supply authority may only levy sewerage service charges on	22 23
		(a)	from which sewage is discharged, or	24
		(b)	from which, in the opinion of the water supply authority, it is reasonably practicable for sewage to be discharged,	25 26
		into	one of the water supply authority's sewer mains.	27
	(3)		ater supply authority may only levy drainage service charges on that is within a drainage area.	28 29
	(4)		ater supply authority may only levy developmental works service ges on land within a development area.	30 31
	(5)		rater supply authority may only levy flood mitigation service ges on land within a floodplain.	32 33

Water Finance		y autho	Division 6	
	(6)		ater supply authority may only levy river management service ges on land within a river management area.	1 2
	(7)		rater supply authority may only levy special industry service ges on land on which a special industry is conducted.	3 4
325	Cer	tain la	and exempt from service charges	5
	(1)		ater supply authority may not levy service charges in respect of the s described in Schedule 4.	6 7
	(2)	Subs	section (1) does not apply to the following land:	8
		(a)	land that is within a public reserve or park that is the subject of a lease, licence or other authority under which a person carries on a trade or business,	9 10 11
		(b)	land that is vested in the Crown or a public body leased to any person for private purposes,	12 13
		(c)	land that is used or occupied by the Crown in connection with an undertaking declared by the Governor by proclamation published in the Gazette to be an industrial undertaking for the purposes of this section.	14 15 16 17
326	Cla	ssifica	ation of lands	18
	(1)	a wa	regulations must specify the factor or factors according to which ter supply authority may classify land for the purpose of levying ice charges.	19 20 21
	(2)	speci	nout limiting the generality of subsection (1), the regulations may ify that a water supply authority may classify land according to any or more of the following factors:	22 23 24
		(a)	the purpose for which the land is actually being used,	25
		(b)	the intensity with which the land is being used for that purpose,	26
		(c)	the purposes for which the land is capable of being used,	27
		(d)	the nature and extent of the water or sewerage services connected to the land.	28 29
	(3)	used	the purposes of subsection (2) (c), land is not capable of being for a purpose if the use of the land for that purpose would be in ravention of:	30 31 32
		(a)	the Environmental Planning and Assessment Act 1979, or	33

Public utilities

Clause 324

Chapter 6

Clause 326			•			
Chapt			ublic utilities			
Part 2 Division 6		Water supply authorities Finance				
DIVISIO)II 0	- ' '	mance			
		(b)	any environmental planning instrument in force under that Act, or	1 2		
		(c)	any other Act or law relating to the use of land.	3		
327	Bas	sis of le	evying service charges	4		
	(1)		regulations must specify the basis or bases according to which a supply authority may levy service charges.	5 6		
	(2)	specif	out limiting the generality of subsection (1), the regulations may fy that a water supply authority may levy service charges ding to any one or more of the following bases:	7 8 9		
		(a)	the land value of land within the meaning of the <i>Valuation of Land Act 1916</i> ,	10 11		
		(b)	the assessed annual value of land within the meaning of the <i>Valuation of Land Act 1916</i> ,	12 13		
		(c)	the nominal size (as determined in accordance with the regulations) of the water service pipe supplying water to the land,	14 15 16		
		(d)	the nominal size (as determined in accordance with the regulations) of the sewerage service pipe discharging sewage from the land,	17 18 19		
		(e)	the area of the land.	20		
	(3)	service value the va	regulation provides that a water supply authority may levy a ce charge on the basis of the value of land (whether on the land of the land, the assessed annual value of the land or otherwise) alue of the land is to be that calculated in accordance with the sions of the <i>Valuation of Land Act 1916</i> .	21 22 23 24 25		
328	Det	ermina	ations by water supply authority	26		
	(1)	water	ater than 1 month before the beginning of each charging year a supply authority must, in accordance with the regulations, mine the following matters for that charging year:	27 28 29		
		(a)	the amount of money that it proposes to raise by way of service charges levied on land within its area of operations,	30 31		
		(b)	the factor or factors according to which land within its area of operations is to be classified for the purposes of levying service charges,	32 33 34		

	(c)	the basis or bases according to which service charges are to be levied,	1 2
	(d)	the rate at which service charges are to be levied on, or the maximum or minimum service charges that are to be applicable to, land within its area of operations.	3 4 5
(2)		king a determination under this section, a water supply authority have regard to the following:	6 7
	(a)	its estimation of the degree of benefit that accrues to land in respect of each service for which a charge is to be levied,	8 9
	(b)	its estimation of the degree to which each service is or may be used in relation to land in respect of which a charge is to be levied,	10 11 12
	(c)	its estimation of the cost involved in the construction, maintenance and operation of each service for which a charge is to be levied,	13 14 15
	(d)	such other matters as it considers relevant to the levying of charges for each such service.	16 17
(3)	A det	ermination under this section:	18
	(a)	is required to be approved by the Minister and does not have effect unless it is so approved, and	19 20
	(b)	must be published in the Gazette before the commencement of the charging year to which it relates, and	21 22
	(c)	takes effect on the commencement of the charging year to which it relates.	23 24
(4)	Gazet relates charge	ermination does not fail merely because it is not published in the te before the commencement of the charging year to which it is but, in that event, no person is liable for payment of the service es to which the determination relates until the determination is shed in the Gazette.	25 26 27 28 29
(5)		vice charge determined under this section is levied on publication determination in the Gazette.	30 31
(6)	If, for	any reason:	32
	(a)	a determination under this section is not made before the charging year to which it relates, or	33 34

Clause 328 Chapter 6 Part 2 Division 6		Water Management Bill 2000 Public utilities Water supply authorities Finance			
		(b)	there is any irregularity or alleged irregularity in the making of any such determination,		
		(whe	Governor may extend the time for making the determination ther or not that time has expired) and may authorise the water ly authority concerned to do anything necessary to cure any ularity and to make a valid determination.		
329	Ass	essme	ent of service charges	,	
	(1)	A wa	tter supply authority:	:	
		(a)	must classify each parcel of land within its area of operations in respect of which a service charge is to be levied, and	10	
		(b)	must assess the service charges payable for each such parcel of land,	1	
		in acc	cordance with its determination under this Division.	1	
	(2)	autho serve	rit makes an assessment under this section, a water supply ority must, in accordance with the regulations, cause a notice to be ad on the landholder of each parcel of land in respect of which a ce charge has been levied.	1. 1: 1: 1	
	(3)	Such	a notice must include information as to the following:	13	
		(a)	the factor or factors according to which the parcel of land has been classified for the purposes of the levying of service charges,	19 20 2	
		(b)	the basis or bases according to which service charges have been levied in respect of the parcel of land,	2:	
		(c)	if service charges have been levied on the basis of the value of the parcel of land, the value of the parcel on which those service charges have been levied,	24 25 20	
		(d)	the rate at which service charges have been levied on, or the maximum or minimum service charges that are applicable to, the parcel of land,	2° 2° 2°	
		(e)	the amounts payable in respect of each service charge that has been levied in respect of the parcel of land.	30	
	(4)	notic	ervice of such a notice, the landholder of the land to which the e relates becomes liable for payment of the service charges fied in the notice.	32 32 34	

Water Finance		y autho	prities	Part 2 Division 6		
330	Re-	asses	sment of service charges		1	
	(1)		section applies to any land in respect of which a been levied on the basis of the value of the land.	service charge	2 3	
	(2)	If a v	water supply authority becomes aware that the value	e (as calculated	4	
			ecordance with this Division) of any parcel of lan		5	
			ch it has levied a service charge for any charging ye		6	
			value (as so calculated) as at the date on which the originally assessed, the water supply authority may	•	7 8	
			ice charge on the basis of the different value.	ty 1c-assess the	9	
	(3)	A se	rvice charge may only be re-assessed as from the d	ate from which	10	
	` '		evised calculation of the value of the land has effe		11	
	(4)	After it makes a re-assessment under this section, a water supply			12	
		authority must, in accordance with the regulations, cause a notice to be served on the landholder of the parcel of land in respect of which the				
	(5)	Such	a notice must specify:		16	
		(a)	the revised calculation of the value of the land,	and	17	
		(b)	the date from which the revised calculation of t land has effect, and	he value of the	18 19	
		(c)	the revised assessment of the amounts payable each service charge that has been levied on that		20 21	
	(6)	landholder of the land to which the notice relates is liable are varied in			22	
					23	
		acco	rdance with the terms of the notice.		24	
Divis	sion 7	7	Miscellaneous		25	
331	Acquisition and divesting of land					
	(1)	A w	ater supply authority may acquire land (or an inter	est in land) for	27	
	- *	the p	ourposes of this Part by agreement or by compuls	sory process in	28	
		acco	rdance with the Land Acquisition (Just Terms Con	ipensation) Act	29	

1991.

Public utilities

30

Clause 330

Chapter 6

Chapter 6 Part 2 Division 7		Public utilities Water supply authorities Miscellaneous		
	(2)	of land is taken to be	e Public Works Act 1912, any such acquisition for an authorised work and the water supply on to that authorised work, taken to be the	1 2 3 2
	(3)	Part 3 of the <i>Public W</i> works constructed under	<i>Vorks Act 1912</i> does not apply in respect of er this Part.	5
	(4)	proclamation published or interest of a water su	local council concerned, the Governor may, by I in the Gazette, vest in a local council the estate apply authority in any land on which is situated supply authority (whether wholly or partly	7 8 9 10 11
332	Cor	pensation for interfere	ence or damage	12
	(1)	If:		13
			or causes to be done, any work that interferes s, any property of a water supply authority, and	14 15
		()	nable cause to believe that the interference or result from the doing of the work,	16 17
		the person is liable to cointerference or damage.	compensate the water supply authority for the .	18 19
	(2)		ity is not entitled to compensation both under provision of this Part for the same interference	20 21 22
333	Effi	iency review		23
	(1)		point a person to investigate and review the apply authority in exercising its functions.	24 25
	(2)		ity must bear the costs of an investigation and on that is requested by it.	26 27
334	App	ointment of administra	ator	28
	(1)	or requirement given or	rity refuses or fails to comply with a direction r made by the Minister in relation to this Part, rder published in the Gazette:	29 30 31

Clause 331

Water Management Bill 2000

Water	utilitie suppl	y autho	Chapter 6 prities Part 2 Division 7	
		(a)	in the case of a water supply authority listed in Part 1 of Schedule 3, appoint an administrator to exercise all the functions, or specified functions, of the water supply authority, or	1 2 3 4
		(b)	in the case of a statutory body listed in Part 2 of Schedule 3, with the consent of the Minister administering the Act by which the statutory body is constituted, appoint an administrator to exercise all the functions, or specified functions, of the water supply authority under this Part.	5 6 7 8 9
	(2)	exclu	administrator has, during his or her term of office and to the usion of the water supply authority, the functions the administrator appointed to exercise.	10 11 12
	(3)	Regu	ulations may be made for or with respect to:	13
		(a)	the accommodation (if any) to be provided at the offices of a water supply authority for the administrator and his or her assistants, and	14 15 16
		(b)	requiring the employees of a water supply authority to assist, and to refrain from obstructing, the administrator in the exercise of his or her functions.	17 18 19
335	Reg	gulatio	ons	20
	(1)		regulations may make provision for or with respect to the wing matters:	21 22
		(a)	fees, service charges and other charges, including charges for the abstraction or use of water,	23 24
		(b)	development of land,	25
		(c)	the supply and use of water,	26
		(d)	special areas,	27
		(e)	water, sewerage and drainage systems,	28
		(f)	trade wastes,	29
		(g)	drainage areas,	30

the opening of the surface of roads,

the procedure at meetings of a water supply authority.

public recreation,

(h)

(i)

(j)

Water Management Bill 2000

31

32

33

Clause 334

another Act, including an environmental planning instrument made

under the Environmental Planning and Assessment Act 1979.

1

4

5

6

7

8

9

34

35

Clause 335

Chapter 6

Water Management Bill 2000

Public utilities

1

Chapter 7 Enforcement

336	Directions concerning the production of information					
	(1)	the la	Minister may, by order in writing served on a landholder, direct andholder to furnish the Minister with specified information in on to:	4 5 6		
		(a)	the construction, use or ownership of any water management works situated on the land, or	7 8		
		(b)	the taking or using of water from a water source by the landholder.	9 10		
	(2)	contr withi	Minister may, by order in writing served on a person by whom a colled activity is being carried out on waterfront land or on land in a water source protection zone, direct the person to furnish the ster with specified information in relation to that activity.	11 12 13 14		
	(3)		rson is not excused from giving information on the ground that aformation may tend to incriminate the person.	15 16		
	(4)	again an of	mation obtained from a person under this section is not admissible ast the person in criminal proceedings, other than proceedings for affence under this Act with respect to the furnishing of false or eading information.	17 18 19 20		
337	Dire	ections	s concerning waste of water	21		
			Minister may, by order in writing served on a landholder, direct andholder to take specified measures to ensure that:	22 23		
		(a)	water used under the authority of a water use approval applying to the land is beneficially used, and is not wasted or improperly used, or	24 25 26		
		(b)	water taken by means of a water supply work situated on the land is beneficially used, and is not wasted or improperly used.	27 28		

338	Dire	ections	s to protect water sources	1	
			Minister may, by order in writing served on a landholder, direct andholder to take specified measures to ensure that:	2 3	
		(a)	the use of a water supply work situated on the land does not impair any water source, or	4 5	
		(b)	the use of a drainage work situated on the land does not impair any water source into which water is discharged by the work, or	6 7 8	
		(c)	the use of a flood work situated on the land does not impair any water source into or from which water is diverted by the work, or	9 10 11	
		(d)	the carrying out of a controlled activity on the land does not impair any water source in the vicinity of the work, or	12 13	
		(e)	interference with an aquifer does not impair the aquifer.	14	
339	Dire	ections	s to stop work where unlawful activity occurring	15	
	(1)	The Minister may, by order in writing served on:			
		(a)	a landholder on whose land is situated a water management work:	17 18	
			 (i) that is being constructed or used, or is about to be constructed or used, in contravention of the provisions of this Act, or (ii) that is being constructed or used in a manner that threatens to impair a water source, or 	19 20 21 22 23	
		(b)	any person having the control and management of such a work,	24	
			t the landholder or person to take specified measures to prevent onstruction or use.	25 26	
	(2)	The N	Minister may, by order in writing served on any person:	27	
		(a)	who is carrying out a controlled activity in contravention of the provisions of this Act, or	28 29	
		(b)	who is carrying out a controlled activity in a manner that threatens to impair a water source,	30 31	
		direct	t the person to cease carrying out that activity.	32	

Water Management Bill 2000
Enforcement
Directions to landholders and other persons

Clause 340 Chapter 7 Part 1

340	Directions to holders of basic landholder rights				
		The I	Minister may, by order in writing served on:	2	
		(a)	 a landholder on whose land is situated a water supply work: (i) that is being used to take water from a water source pursuant to the landholder's domestic and stock rights, 	3 4 5	
			or (ii) that is being used to capture rainwater run-off pursuant to the landholder's harvestable rights, or	6 7 8	
		(b)	any person having the control and management of such a work,	9	
			t the landholder or person to take specified measures to protect the conment or the allocations to water of downstream users.	10 11	
341	Ten	nporar	ry stop work order	12	
		The I	Minister may, by order in writing served on:	13	
		(a)	any landholder on whose land is situated a water supply work or drainage work, or	14 15	
		(b)	any person having the control or management of such a work, or	16 17	
		(c)	any person by whom a controlled activity is being carried out,	18	
	direct the landholder or person to cease using the work, or to cease the activity, in any circumstances in which the Minister considers that the public interest requires such a direction to be given.		19 20 21		
342	Dire	ections	s concerning unusable water management works	22	
	(1)	The I	Minister may, by order in writing served on:	23	
		(a)	a landholder on whose land is situated a water management work whose construction or use is not authorised by this Act, or	24 25 26	
		(b)	any person having the control or management of that work,	27	
			t the landholder or person to take specified measures to remove render it inoperable.	28 29	
	(2)		a direction may be given even if the water management work is	30 31	

Part 1 Directions to landholders and other persons

343	Dire	ections	s concerning damage caused by straying stock	1
			Minister may, by order in writing served on any landholder, direct andholder to take specified measures:	2
		(a)	to prevent stock from straying from the landholder's land into or onto a water management work that is owned by, or is under the control or management of, the Ministerial Corporation, or	4 5 6
		(b)	to repair any damage caused to any such water management work as a consequence of stock having strayed from the landholder's land.	7 8 9
344	Mea	asures	that may be specified in direction	10
	(1)		measures that may be specified in a direction under this Part are llows:	11 12
		(a)	measures to clean, alter, reconstruct or add to a water management work or otherwise render it effective,	13 14
		(b)	measures to demolish, remove, dismantle or block a water management work or otherwise render it ineffective,	15 16
		(c)	measures to restore or enhance the condition of any water source that has been impaired by the use, misuse or lack of use of a water management work or by the carrying out of any controlled activity,	17 18 19 20
		(d)	measures to repair any damage caused by a water management work or by the carrying out of any controlled activity (including any damage caused to any specified land, structure or vegetation, or to the environment),	21 22 23 24
		(e)	measures to ensure that any specified land, structure or vegetation, or the environment, will not be damaged or adversely affected or further damaged or adversely affected by a water management work or by the carrying out of any controlled activity,	25 26 27 28 29
		(f)	measures to correct or restore any alteration caused by a water management work, or by the carrying out of any controlled activity, to the flow of water in, to or from, or the quantity of water contained in, any specified water source,	30 31 32 33
		(g)	any ancillary measures that the Minister considers to be necessary or expedient.	34 35

Directi	ons to	landholders and other persons Part 1				
	(2)	In this section, <i>controlled activity</i> includes the activity of penetrating an aquifer, or interfering with water in an aquifer or obstructing its flow.	1 2 3			
345	Pric	or notice of direction not required	4			
		The Minister is not required to notify any person who may be affected by a direction under this Part before giving the direction.	5 6			
346	Remedial measures may be taken by Minister					
	(1)	If a person fails to take the measures specified in a direction under this Part, the Minister may authorise any other person to take those measures.	8 9 10			
	(2)	The amount of any costs and expenses incurred by the authorised person as a result of the taking of those measures is recoverable in a court of competent jurisdiction as a debt due to the Ministerial Corporation from the person on whom the direction was served.	11 12 13 14			
347	Lan	d and Environment Court may grant injunctions	15			
		On the application of the Minister, the Land and Environment Court may grant an injunction directing a landholder to comply with a direction under this Part.	16 17 18			

Enforcement

Restraint of breaches of this Act

regulations.

own behalf or on behalf of another person (with their consent), or of
a body corporate or unincorporate (with the consent of its committee
or other controlling body), having like or common interests in those
proceedings.

(4) Any such proceedings may be brought by a person on the person's

(1) Any person may bring proceedings in the Land and Environment Court

(2) Any such proceedings may be brought whether or not proceedings

(3) Any such proceedings may be brought whether or not any right of the

for an order to remedy or restrain a breach of this Act or the

have been instituted for an offence against this Act or the regulations.

person has been or may be infringed by or as a consequence of the

Clause 344 Chapter 7

Part 1

Chapter 7 Enforcement
Part 1 Directions to landholders and other persons

(5) Any person on whose behalf proceedings are brought is entitled to contribute to or provide for the payment of the legal costs and expenses incurred by the person bringing the proceedings.

Water Management Bill 2000

- (6) If the Court is satisfied that a breach has been committed or that a breach will, unless restrained by the order of the Court, be committed, it may make such orders as it thinks fit to remedy or restrain the breach.
- (7) In this section, *breach* includes a threatened or apprehended breach.

1

2

3

4

5

6

7

8

Clause 348

Part 2 Powers of entry

349	Pov	vers of	f entry generally	2
	(1)	An au	uthorised officer may enter any premises:	3
		(a)	for the purpose of inspecting any water management works situated on the premises, or	5
		(b)	for the purpose of monitoring the use of water on the premises, or	6
		(c)	for the purpose of monitoring any controlled activity or aquifer interference occurring on the premises, or	9
		(d)	for the purpose of carrying out any surveys for the purposes of this Act, or	10 11
		(e)	for the purpose of taking measurements of any matter, or for reading any meter, for the purposes of this Act, or	12 13
		(f)	for the purpose of investigating any alleged contravention of this Act or the regulations.	14 15
	(2)	Reaso	onable force may be used for the purpose of effecting entry.	16
	(3)	Excep hours	pt in emergencies, entry may be effected only during daylight	17 18
	(4)	section	I times while on premises under the power conferred by this on, an authorised officer must carry, and produce on demand, nce of his or her authority to be on the premises.	19 20 21
	(5)		section does not apply to any part of premises that are used sively for residential purposes.	22 23
50	Sea	rch wa	arrants	24
	(1)	of a s	athorised officer may apply to an authorised justice for the issue search warrant if the authorised officer believes on reasonable ads that a provision of this Act or the regulations is being or has contravened at any premises.	25 26 27 28
	(2)	satisf warra	uthorised justice to whom such an application is made may, if ied that there are reasonable grounds for doing so, issue a search ant authorising a named authorised officer to enter the premises of exercise any of the authorised officer's functions under this Part	29 30 31

1

2

3

4

5

6

7

8

9

30

of entry.

Clause 350

Chapter 7

Water Management Bill 2000

Enforcement

Water Management Bill 2000	Clause 353
Enforcement	Chapter 7
Offences	Part 3
Major offences	Division 1

Part	3 (Offences	1
Divis	sion '	1 Major offences	2
353	Unla	awful taking of water	3
	(1)	A person must not take an amount of water from any water source otherwise than in accordance with an access licence and any relevant available water determination.	4 5 6
	(2)	This section does not prevent a landholder from exercising a basic landholder right in accordance with this Act.	7 8
	(3)	In this section, a reference to an access licence includes a reference to any licence of a similar nature (however described):	9 10
		(a) that is granted under the law of another State or Territory, and	11
		(b) that is declared by the regulations to have the same effect as an access licence for the purposes of this section.	12 13
	(4)	A person is not guilty of an offence under this section as a consequence of having taken water in contravention of an available water determination if the person establishes that he or she took all reasonable steps to ascertain the terms of the determination but was unable to do so.	14 15 16 17 18
354	Usi	ng water without a water use approval	19
	(1)	A person must not use any amount of water for any purpose on any land otherwise than in accordance with a water use approval in force in respect of that land.	20 21 22
	(2)	This section does not prevent a landholder from exercising a basic landholder right in accordance with this Act.	23 24
355		nstructing or using water management work without a water nagement work approval	25 26
	(1)	A person must not:	27
		(a) construct or use a water supply work otherwise than in accordance with a water supply work approval, or	28 29
		(b) construct or use a drainage work otherwise than in accordance with a drainage work approval, or	30 31

Clause 355		V	Vater Management Bill 2000	
Chapter 7		Е	inforcement	
Part 3		Offences		
Division	on 1	N	Najor offences	
		(c)	construct or use a flood work in or in the vicinity of a river,	1
			estuary or lake, or within a floodplain, otherwise than in	2
			accordance with a flood work approval.	3
	(2)		ection (1) (a) does not prevent a landholder from exercising a	4
			landholder right (otherwise than by means of a dam or water	5
		bore)) in accordance with this Act.	6
	(3)	Subs	ection (1) (c) does not prevent a landholder from constructing or	7
		using	g a water supply work or drainage work in accordance with a	8
		drain	age work approval or flood work approval.	9
356	Unl	awful	carrying out of certain activities	10
	(1)	A pe	rson must not:	11
		(a)	carry out a controlled activity on waterfront land or in a water	12
			source protection zone otherwise than in accordance with a	13
			controlled activity approval, or	14
		(b)	penetrate an aquifer, or interfere with water in an aquifer or	15
			obstruct its flow, otherwise than in accordance with an aquifer	16
			interference approval.	17
	(2)	Subs	ection (1) does not prevent a person from constructing and using	18
	` /		ter management work in accordance with a water management	19
		work	approval.	20
	(3)	Subs	ection (1) (a) does not prevent a person:	21
		(a)	from carrying out a controlled activity in accordance with a	22
			controlled activity approval, or	23
		(b)	from using a building or work that has been erected or carried	24
			out in accordance with a controlled activity approval.	25
	(4)		ection (1) does not apply to or in respect of any person or activity	26
			npt from the need for an activity approval by an order in force	27
		unde	r section 83.	28
357	Cor	ntravei	ntion of certain directions	29
		A per	rson on whom a direction under Part 1 is served must not fail to	30
		comp	oly with the direction.	31

Water Management Bill 2000	Clause 358
Enforcement	Chapter 7
Offences	Part 3
Major offences	Division 1

358	Des	structio	on, damage and interference with certain works	1
	(1)	A per	rson must not destroy, damage or interfere with:	2
	(-)	(a)	any work that is owned by, or is under the control and management of, the Ministerial Corporation, an irrigation corporation, a private irrigation board, a private drainage board or a private water trust, or	3 4 5
		(b)	any meter (regardless of who owns it or has the control and management of it) that is used for measuring the quantity or quality of water in, or passing through, a water supply work or drainage work, or	7 8 9 10
		(c)	any mark, peg, stake or level fixed for the purposes of this Act.	11
	(2)	estab	rson is not guilty of an offence against this section if the person lishes that the act giving rise to the alleged offence was done with all authority.	12 13 14
359	Tak	ing wa	ater from public or private works	15
	(1)	owne Corpo privat	rson must not take water from any water supply work that is ad by, or is under the control and management of, the Ministerial coration, an irrigation corporation, a private irrigation board, a te drainage board or a private water trust, except with the crity of that corporation, board or trust.	16 17 18 19 20
	(2)	estab	rson is not guilty of an offence against this section if the person lishes that the act giving rise to the alleged offence was done with all authority.	21 22 23
360	Max	kimum	penalty for offences under this Division	24
			rson who is guilty of an offence under this Division is liable, on iction:	25 26
		(a)	in the case of a corporation, to a penalty not exceeding 2,500 penalty units and, in the case of a continuing offence, to a further penalty not exceeding 1,200 penalty units for each day the offence continues, or	27 28 29 30
		(b)	in the case of an individual, to a penalty not exceeding 1,200 penalty units and, in the case of a continuing offence, to a further penalty not exceeding 600 penalty units for each day the offence continues.	31 32 33 34

Part 3 Division	n 2	Offences Other offences	
Divis	ion 2	on 2 Other offences	
361	Expo	sure of underground pipes	2
		A person must not, by opening any ground, expose any pipe or other work of a water supply authority:	3 4
	(a) without lawful excuse, or	5
	(b) without having given the water supply authority at least 2 days' written notice of intention to open the ground.	6 7
		Maximum penalty: 100 penalty units in the case of a corporation or 20 penalty units in any other case.	8 9
362	Work	done by unqualified person	10
	Ċ	A person must not do any kind of work of water supply, sewerage or lrainage intended for direct or indirect connection with the pipes, ewers or drains of a water supply authority unless the person:	11 12 13
	(a) holds an endorsed licence or a supervisor certificate in force under the <i>Home Building Act 1989</i> authorising the holder to do that kind of work, or	14 15 16
	(b) does the work under the immediate supervision of the holder of such a licence or certificate, or	17 18
	(holds a registration certificate in force under the <i>Home Building Act 1989</i> authorising the holder to do that work under supervision, and does that work under the general supervision of the holder of a licence or certificate referred to in paragraph (a).	19 20 21 22 23
	N	Maximum penalty: 100 penalty units.	24
363	Obstr	uction	25
		A person must not obstruct or hinder an authorised officer in the exercise of the authorised officer's functions under this Act.	26 27

28

Clause 361

Chapter 7

Water Management Bill 2000

Maximum penalty: 20 penalty units.

Enforcement

Water Management Bill 2000	Clause 364
Enforcement	Chapter 7
Offences	Part 3
Other offences	Division 2

364	False or misleading information	1
	A person must not, in or in connection with any application under this	2
	Act, make a statement that the person knows to be false or misleading	3
	in a material particular.	4
	Maximum penalty: 10 penalty units.	5

Part 4 Recovery of rates and charges

Part	4 F	Recovery of rates and charges	1
365	Def	inition	2
		In this Part, <i>charging authority</i> means the Minister, an irrigation corporation, a private irrigation board, a private drainage board, a private water trust or a water supply authority.	3 4 5
366	Cer	tain rates and charges to be a charge on land	6
	(1)	A rate or charge imposed on the owner of any land is a charge on the land to which the charge relates.	7 8
	(2)	A charge imposed for a service or thing supplied or provided in connection with the occupation of land is a charge on the land.	9 10
367	Inte	rest on rates and charges	11
		A charging authority may charge interest on overdue rates and charges at a rate not exceeding the rate of interest payable for the time being on an unpaid judgment of the Supreme Court.	12 13 14
368	Rec	covery of rates, charges and other money	15
	(1)	Any rate or charge or other money due to a charging authority under this Act may be recovered in any court of competent jurisdiction as if it were a debt due to the charging authority.	16 17 18
	(2)	An unsatisfied judgment or order of any court for the recovery of any rate or charge from any person is not a bar to its recovery from any other person who is liable under this Act for its payment.	19 20 21
369	Joir	nt owners	22
	(1)	If land is owned or held jointly by two or more persons, such persons are jointly and severally liable to the charging authority for the rate, but as between themselves each are only be liable for such part of the rate as is proportionate to his or her interest in the land.	23 24 25 26
	(2)	If any of such persons pays to the charging authority more than his or her proportionate part, he or she may recover the excess by way of contribution from the others.	27 28 29

Water Management Bill 2000
Enforcement
Recovery of rates and charges

Clause 370 Chapter 7 Part 4

370	Sale	e of la	nd for unpaid rates and charges	1
	(1)	This	section applies to rates and charges imposed under this Act by the ster or by a water supply authority.	2 3
	(2) The provisions of Division 5 of Part 2 of Chapter 17 of the <i>Local Government Act 1993</i> apply to land in respect of which a rate or charge (being a rate or charge that is, by virtue of this Part, a charge on land) remains unpaid in the same way as they apply to land in respect		rnment Act 1993 apply to land in respect of which a rate or ge (being a rate or charge that is, by virtue of this Part, a charge on	4 5 6 7 8
	(3)	Chap	the purpose of applying the provisions of Division 5 of Part 2 of the 17 of the <i>Local Government Act 1993</i> to land in respect of the a rate or charge remains unpaid:	9 10 11
		(a)	a reference in those provisions to a council is to be read as a reference to the Minister or to a water supply authority, as the case requires, and	12 13 14
		(b)	a reference to a general manager or public officer is to be read as a reference to the Director-General or to the principal officer of a water supply authority, as the case requires, and	15 16 17
		(c)	a reference to a member of staff of a council is to be read as a reference to a member of staff of the Department or to an employee of a water supply authority, as the case requires, and	18 19 20
		(d)	a reference to a rating authority is to be read as including a reference to a council.	21 22
371	Cer	tificate	e as to amount due	23
	(1)		arging authority must, on written application being made to it and an antion of the fee determined by it, issue to the applicant a certificate:	24 25
		(a)	containing particulars of any amounts payable to the authority in respect of a parcel of separately assessed land, or	26 27
		(b)	to the effect that there are no such amounts.	28
	(2)	An a _l	pplication for a certificate must:	29
		(a)	specify the name and address of the applicant, and	30
		(b)	identify the land to which the application relates.	31

Chapte Part 4	er 7	Enforcement Recovery of rates and charges	
	(3)	Such a certificate is conclusive proof, in favour of a purchaser in good faith and for value of the land to which the certificate relates that, at the date of its issue, no amounts were payable to the charging authority in respect of that land other than such amounts as are specified in the certificate.	1 2 3 4 5
372	Lial	pility where an estate or interest is transferred	6
	(1)	A person who disposes of his or her estate or interest in any land in respect of which any rates or charges have been or may be levied remains liable for rates or charges to the same extent as if the person had not disposed of his or her estate or interest in the land, if the rates or charges are levied either:	7 8 9 10 11
		(a) before the person disposed of his or her estate or interest in the land, or	12 13
		(b) before the person has given to the charging authority the prescribed notice of disposal.	14 15
	(2)	If any person who disposes of land to another person pays any amount to the charging authority in respect of rates or charges levied after the land disposed of but before the prescribed notice is given to the authority, the person by whom the amount was paid may recover the amount from the person to whom the land was disposed.	16 17 18 19 20
	(3)	As between an owner of land and any other person from or to whom the owner derives or disposes of his or her estate or interest in the land, rates or charges under this Act are to be considered as accruing from day to day and are apportionable accordingly.	21 22 23 24
373	Lial	pility where a person becomes entitled to an estate or interest	25
	(1)	An owner of land is liable for all arrears of rates and charges owing by any previous owner of the land, despite the fact that the new owner acquired the land after the rates or charges were levied.	26 27 28
	(2)	If any person who becomes an owner of land pays to the charging authority any rates or charges in respect of that land that were levied before the person became the owner, the person may recover from the previous owner such part of the rates or charges as was levied in	29 30 31 32

respect of the period during which the previous owner was the owner of the land.

33 34

Clause 371

Water Management Bill 2000

Water Management Bill 2000	Clause 374
Enforcement	Chapter 7
Legal proceedings and appeals	Part 5
Legal proceedings	Division 1

Part 5 Legal proceedings and appeals

Divis	ion '	I	Legal proceedings	2	
374	Offences by corporations				
	(1)	If a co	orporation commits an offence against this Act or the regulations:	4	
		(a)	each person who is a director of the corporation, and	5	
		(b)	each person who is concerned in the management of the corporation,	6 7	
			ten to have committed the same offence if the person knowingly orised or permitted the act or omission constituting the offence.	8	
	(2)		rson may be proceeded against and convicted under this section her or not the corporation has been proceeded against or icted.	10 11 12	
	(3)		ing in this section affects any liability imposed on a corporation offence committed by the corporation.	13 14	
375	Proceedings for offences				
	(1)	Proceedings for an offence against this Act or the regulations are to be disposed of summarily:		16 17	
		(a)	by a Local Court constituted by a Magistrate sitting alone, or	18	
		(b)	by the Land and Environment Court in its summary jurisdiction.	19 20	
	(2)	2) Proceedings for an offence against this Act may be commenced:		21	
		(a)	in the case of an offence prescribed by the regulations, within (but not later than) 3 years after the date of the alleged commission of the offence, and	22 23 24	
		(b)	in the case of any other offence, within (but not later than) 12 months after the date of the alleged commission of the offence.	25 26	
	(3)		maximum monetary penalty that may be imposed by a Local t in proceedings for an offence against this Act or the regulations	27 28 29	
		(a)	100 penalty units for an offence committed by a corporation, or	30	

Chapter 7 Part 5 Division 1		Enforcement Legal proceedings and appeals Legal proceedings			
		(b) 50 penalty units in any other case,			
		or the maximum monetary penalty specified in respect of the offence, whichever is the lesser.			
	(4)	The maximum penalty that may be imposed by the Land and Environment Court in proceedings for an offence against this Act or the regulations is the maximum penalty specified in respect of the offence.			
376	Penalty notices for certain offences				
	(1)	In this section:			
		<i>penalty notice</i> means a notice to the effect that, if the person served with the notice does not wish to have an alleged offence dealt with by a court, the person may pay, in accordance with the notice, the penalty specified in the notice.			
		penalty notice offence means an offence against this Act or the regulations that is declared by the regulations to be a penalty notice offence.			
	(2)	An authorised officer may serve a penalty notice on a person who appears to the authorised officer to have committed a penalty notice offence.			
	(3)	The amount of the penalty to be specified in a penalty notice is the amount prescribed by the regulations for the alleged offence, being an amount not exceeding the maximum penalty which could be imposed for the offence by a court.			
	(4)	A penalty notice may be served personally or by post.	;		
	(5)	If the amount of the penalty prescribed by the regulations for an alleged offence is paid under this section, no person is liable to any further proceedings for the alleged offence.	: :		
	(6)	Payment of a penalty under this section is not to be regarded as an admission of liability for the purposes of, nor is in any way to affect or prejudice, any civil claim, action or proceeding arising out of the same occurrence.	: :		
	(7)	within 28 days after the date on which it was served and, in that event:			
		(a) the amount payable under the notice ceases to be payable, and	3		

Clause 375

Water Management Bill 2000

		edings edings	and appeals	Part 5 Division 1	
		(b)	any amount that has been paid under the person by whom it was paid, and	he notice is repayable to	1
		(c)	further proceedings for the offence i notice was served may be taken agains the person on whom the notice was ser never been served.	st any person (including	3 2 4
	(8)	This section does not limit the operation of any other provision of this or any other Act or law in relation to proceedings that may be taken in respect of offences.			? 9
377		gal proceedings do not affect, and are unaffected by, other action der this Act		10 11	
		regu offer	prosecution of a person for an offence lations, or the issue of a penalty notice ace, does not affect, and is unaffected by er this Act in relation to the act or omi- ace.	e in respect of such an r, any other action taken	12 13 14 15
378	Evi	dentia	ry certificates		17
	(1)		ertificate that is issued by the Ministeria s that, on a date or during a period specif		18 19
		(a)	a specified person was or was not th access licence or specified approval, o		20 21
		(b)	specified land was or was not the approval, or	subject of a specified	22 23
		(c)	a specified water management work wa of a specified water management work		24 25
		(d)	the conditions of a specified access lice were not as so specified, or	ence or approval were or	26 27
		(e)	the terms of an available water determine as so specified,	ination were or were not	28 29
			lmissible in any legal proceedings and is so stated.	evidence of the fact or	30 31

Enforcement

Clause 376

Chapter 7

Part 5 Divisio	on 1	Legal proceedings and appeals Legal proceedings		
	(2)	In pr	roceedings for an offence against this Act or the regulations:	1
		(a)	evidence that a water management work was constructed or used on specified land is evidence that the work was constructed or used by the person who was the landholder of the land at the time the work was constructed or used, and	2 3 4 5
		(b)	evidence that water was taken from a water source by means of a water supply work situated on specified land is evidence that the water was taken from the water source by the person who was the landholder of the land at the time the water was taken, and	6 7 8 9 10
		(c)	evidence that water was discharged into a water source by means of a drainage work situated on specified land is evidence that the water was discharged into the water source by the person who was the landholder of the land at the time the water was discharged, and	11 12 13 14 15
		(d)	evidence that water was used on specified land is evidence that the water was used by the person who was the landholder of the land at the time the water was used.	16 17 18
Divis	ion 2	2	Appeals	19
379	App	eals t	to Land and Environment Court	20
	(1)		appeal lies to the Land and Environment Court against any of the wing decisions made by the Minister:	21 22
		(a)	a decision refusing to grant an access licence,	23
		(b)	a decision granting a designated access licence, if the appellant was an objector to the granting of the licence,	24 25
		(c)	a decision imposing a discretionary condition on an access licence,	26 27
		(d)	a decision fixing the term of an access licence,	28
		(e)	a decision refusing consent to the transfer of an access licence,	29
		(f)	a decision suspending or cancelling an access licence,	30
		(g)	a decision refusing to grant an approval, other than a decision refusing to accept an application for an approval,	31 32
		(h)	a decision granting a designated water use approval, if the appellant was an objector to the granting of the approval,	33 34

Enforcement

Clause 378 Chapter 7

Appeals		Division 2	
	(i)	a decision imposing a discretionary condition on an approval,	
	(j)	a decision fixing the term of an approval,	;
	(k)	a decision refusing to amend an approval in accordance with an application made by its holder,	,
	(1)	a decision suspending or cancelling an approval,	
	(m)	a decision to give a direction to a landholder under Part 1,	
	(n)	a decision as to a person's entitlement to compensation for	
		damage arising from the exercise of a power of entry under Part 2.	!
(2)	Desp	oite subsection (1):	1
	(a)	no appeal lies against any decision made by the Minister on an	1
		application to which an objection has been made if:	1:
		(i) in the case of the applicant, the Minister has dismissed	1.
		the application as a consequence of the applicant having	14
		failed to participate in mediation or neutral evaluation	1:
		proceedings, or	10
		(ii) in the case of the objector, the Minister has dismissed	1
		the objection as a consequence of the objector having	1

Legal proceedings and appeals

(b)

Enforcement

(3) An appeal is to be made in accordance with rules of court, but may not be made more than 28 days after the date on which the decision was made.

proceedings, and

failed to participate in mediation or neutral evaluation

no appeal lies against any decision made by the Minister

pursuant to a report from a Committee of Inquiry under section

119 of the Environmental Planning and Assessment Act 1979.

- (4) In addition to the appellant and the Minister, the parties to an appeal against a decision to grant a designated access licence or designated approval include any objector to the granting of the licence or approval who, in accordance with rules of court, gives notice to the Land and Environment Court of the objector's wish to be a party to the appeal.
- (5) The lodging of an appeal does not operate to stay action on the decision appealed against, except to the extent that the Land and Environment Court otherwise directs.

Clause 379

19

20

21

22

23

24

25

26

27

28

29

30

31

32

33

34

Chapter 7

Part 5

Part 5 Division 2	Legal proceedings and appeals Appeals	
(6)	In this section:	1
	designated access licence means an access licence to which a person has a right of objection under section 52.	2 3
	designated approval means a water use approval to which a person has a right of objection under section 85.	4 5

Enforcement

Clause 379 Chapter 7

1

Chapter 8 Administration

Part	1 \	Nate	r Advisory Council	2
380	Esta	ablishı	ment of Water Advisory Council	3
	(1)	There	e is to be a Water Advisory Council.	4
	(2)		Water Advisory Council is to have at least 12, but not more than numbers appointed by the Minister, of whom:	5 6
		(a)	at least two are to be persons appointed to represent the interests of environmental protection groups, and	7 8
		(b)	at least two are to be persons appointed to represent the interests of water user groups, and	9 10
		(c)	at least two are to be persons appointed to represent the interests of local councils, and	11 12
		(d)	at least one is to be a person having such technical qualifications in connection with environmental protection as the Minister considers appropriate for the functions to be exercised by the Council, and	13 14 15 16
		(e)	at least one is to be a person having such qualifications in ecology as the Minister considers appropriate for the functions to be exercised by the Council, and	17 18 19
		(f)	at least two are to be persons appointed to represent the interests of catchment management boards and trusts, and	20 21
		(g)	at least one is to be an Aboriginal person appointed to represent the interests of Aboriginal persons, and	22 23
		(h)	one is to be a person appointed as an independent chairperson for the Council.	24 25
	(3)		regulations may make provision with respect to qualifications for intment as a member of the Water Advisory Council.	26 27
	(4)		dule 6 has effect with respect to the constitution and procedure of Vater Advisory Council.	28 29

Part 1 Water Advisory Council

381 I	Fun	nctions of Water Advisory Council	1	l
((1)	The principal functions of the Water Advisory Council are as	follows:	2
		(a) to review such draft management plans and implement programs as the Minister refers to it for review,	nentation 3	
		(b) to investigate matters affecting the management of t sources throughout the State, including such matter Minister refers to it for investigation,		6
		(c) to report to the Minister on matters affecting the man of the water sources throughout the State, including matters as the Minister refers to it for report,)
		(d) to advise the Minister on matters affecting the manag the water sources throughout the State, including such as the Minister refers to it for advice.		2
((2)	The Water Advisory Council has such other functions as are cor imposed on it by or under this or any other Act or law.	conferred 14	
	(3)	For the purpose of exercising its functions, the Water A Council may consult with, and receive submissions from, othe and bodies.	•	7
((4)	It is the duty of the Water Advisory Council to exercise its fit consistently with the principles of ecologically sus development.)

functions, and

Part	2 \	Water Administration Ministerial Corporation	1
382	Cor	nstitution of Water Administration Ministerial Corporation	2
	(1)	There is constituted by this Act a corporation with the corporate name "Water Administration Ministerial Corporation".	3 4
	(2)	The affairs of the Ministerial Corporation are to be managed by the Minister.	5 6
	(3)	Any act, matter or thing done in the name of, or on behalf of, the Ministerial Corporation by the Minister, or with the authority of the Minister, is taken to have been done by the Ministerial Corporation.	7 8 9
	(4)	The Ministerial Corporation is, for the purposes of any Act, a statutory body representing the Crown.	10 11
383	Fun	actions of Ministerial Corporation	12
	(1)	The Ministerial Corporation has the following functions:	13
		(a) to construct, maintain and operate water management works,	14
		(b) to conduct research, collect information and develop technology in relation to water management,	15 16
		(c) to do anything for the purpose of enabling the objects of this Act to be attained.	17 18
	(2)	The Ministerial Corporation may exercise its functions within and beyond New South Wales.	19 20
	(3)	The Ministerial Corporation may exercise any of its functions, and may otherwise act, in the name of the Department.	21 22
	(4)	It is the duty of the Ministerial Corporation to exercise its functions consistently with the principles of ecologically sustainable development.	23 24 25
384	Cor	nmercial operations	26
	(1)	With the approval of the Governor, the Ministerial Corporation may enter into commercial operations with respect to:	27 28
		(a) any services developed in connection with the exercise of its	29

30

1

6

8

Clause 384

Chapter 8

Water Management Bill 2000

Administration

Part	3 (General administration	1
389	Cor	nmittees	2
		The Minister may establish advisory and other committees for the purposes of this Act.	3 4
390	Dele	egation	5
	(1)	The Minister may delegate to any person any of the Minister's functions under this Act, other than this power of delegation.	6 7
	(2)	Despite subsection (1), the Director-General may sub-delegate to any person any such function that has been delegated to the Director-General, but only if the terms of the delegation authorise the Director-General to sub-delegate that function.	8 9 10 11
391	Aut	horised officers	12
		The Minister may appoint authorised officers, either for the purposes of this Act generally or for the purposes of any specified provisions of this Act.	13 14 15
392	Res	olution of disputes between public authorities	16
	(1)	Any dispute arising between two or more public authorities as to the exercise of their functions under this Act is to be resolved:	17 18
		(a) by agreement between the Ministers responsible for them, or	19
		(b) if the dispute is between Ministers, by the Premier.	20
	(2)	Each public authority must comply with any directions arising out of the resolution of the dispute under this section.	21 22

Cha	pte	r 9	Miscellaneous	1
393	Stat	te's w	ater rights	2
	(1)	For t	he purposes of this Act, the rights to the control, use and flow of:	3
		(a)	all water in rivers, lakes and aquifers, and	4
		(b)	all water conserved by any works that are under the control or management of the Minister, and	5 6
		(c)	all water occurring naturally on or below the surface of the ground,	7 8
		are th	he State's water rights.	9
	(2)	to w	State's water rights are vested in the Crown, except to the extent hich they are divested from the Crown by or under this or any Act.	10 11 12
	(3)	unde	State's water rights prevail over any authority conferred by or any other Act or law, except to the extent to which this or any Act expressly so provides.	13 14 15
	(4)	The	State's water rights are not to be exercised in contravention of:	16
		(a)	the powers conferred by section 15 of the <i>Fire Brigades Act</i> 1989 and section 26 of the <i>Rural Fires Act</i> 1997 on an officer in charge at a fire or hazardous material incident, or	17 18 19
		(b)	the rights and obligations of the Snowy Hydro Company under a Snowy water licence within the meaning of the <i>Snowy Hydro Corporatisation Act 1997</i> , or	20 21 22
		(c)	the rights conferred by the <i>Rural Lands Protection Act 1989</i> with respect to the watering of stock at stock watering places within the meaning of that Act.	23 24 25
394	Abo	olition	of common law riparian rights	26
		have lake	right that the owner of riparian land would, but for this section, at common law with respect to the flow of any river, estuary or through or past the land, or to the taking or using of water from such river, estuary or lake, is hereby abolished.	27 28 29 30

395	Service of documents					
	(1)	A do	cument may be served on a person:	2		
		(a)	personally, or	3		
		(b)	by leaving it at, or by sending it by post to, the person's residential address or, in the case of a corporation, the address of the corporation's registered office, or	5 6		
		(c)	if it is impracticable for the document to be served on a landholder of land in the manner referred to in paragraph (a) or (b), by affixing the document in a conspicuous position on the land.	5 8 9		
	(2)	or of document them	e case of land that is owned or occupied by more than one person an approval that is held by more than one person, service of a ment on any one of them, or on any person duly appointed by for the purpose of accepting service of documents, is taken to be ce on all of them.	11 12 13 14		
	(3)	This s	section does not limit any other manner in which a document may rved.	16 17		
396	Unp	aid fe	es and charges	18		
		unpai	fee or charge imposed by the Minister under this Act that remains id is recoverable in any court of competent jurisdiction as a debt o the Crown.	19 20 21		
397	Act	binds	Crown	22		
			Act binds the Crown in right of New South Wales and, in so far e legislative power of Parliament permits, the Crown in all its cities.	23 24 25		
398	Exc	lusion	of personal liability	26		
	(1)	An a	ct or omission of:	27		
		(a)	the Minister or the Director-General, or	28		
		(b)	a prescribed authority, or	29		
		(c)	a member of staff of the Department or of a prescribed authority, or	30 31		

		(d)	a person acting under the direction of a person referred to in paragraph (a), (b) or (c),	1 2
		does	not subject the Minister, Director-General, member of staff or	3
			n so acting personally to any action, liability, claim or demand if	4
			et or omission was done, or omitted to be done, in good faith for	5
		the p	urpose of executing this Act.	6
	(2)	In thi	s section, <i>prescribed authority</i> means:	7
		(a)	the Ministerial Corporation, or	8
		(b)	the Water Advisory Council, or	9
		(c)	a management committee, or	10
		(d)	a water supply authority, or	11
		(e)	a private irrigation board, or	12
		(f)	a private drainage board, or	13
		(g)	a private water trust.	14
399	Exc	lusion	of Crown liability	15
	(1)		er the Crown nor any other person is subject to any action, ity, claim or demand arising:	16 17
		(a)	from the unavailability of water, or	18
		(b)	from any failure in the quantity or quality of water,	19
		as a c	onsequence of anything done or omitted to be done in good faith	20
			e Minister, by a prescribed authority or by any person acting on	21
			f of the Minister or a prescribed authority, in the exercise any	22
		funct	ions under this Act.	23
	(2)		er the Crown nor any other person is subject to any action, ity, claim or demand arising as a consequence of:	24 25
		(a)	the use in good faith of any water management work, or	26
		(b)	the release in good faith of water from any water management	27
			work,	28
		by the	e Minister, by a prescribed authority or by any person acting on	29
		behal	f of the Minister or a prescribed authority, in the exercise any	30
		funct	ions under this Act.	31

	(3)	In th	is section, <i>prescribed authority</i> means:	1
		(a)	the Ministerial Corporation, or	2
		(b)	a water supply authority.	3
400	Anr	nual re	eport of Department	4
		The	annual report prepared for the Department under the Annual	5
		Repo	orts (Departments) Act 1985 is to include a report on the	6
			ster's work and activities under this Act for the period to which	7
			report relates, and may also include the annual report prepared for	8
			Ministerial Corporation under the <i>Annual Reports (Statutory es) Act 1984</i> .	9 10
401	Reg	julatio	ons	11
	(1)	The (Governor may make regulations, not inconsistent with this Act, for	12
	(-)		ith respect to any matter that by this Act is required or permitted	13
			prescribed or that is necessary or convenient to be prescribed for	14
		carry	ring out or giving effect to this Act and, in particular, for or with	15
		respe	ect to:	16
		(a)	the forms to be used in connection with the administration of	17
			this Act, and	18
		(b)	the fees and charges to be imposed in connection with the	19
			administration of this Act, and the circumstances under which	20
			fees and charges may be waived, reduced and remitted, and	21
		(c)	the requirement for security deposits, the circumstances in	22
			which security deposits are to be forfeited and the application	23
			of money arising from the forfeiture of security deposits, and	24
		(d)	the regulation and control of bore drilling and the licensing of	25
			bore drillers, and	26
		(e)	the information to be provided to the Minister by the holder of	27
			any approval under this Act, including the circumstances in	28
			which any such information must be verified by statutory declaration.	29
				30
	(2)		egulation may exempt any person, matter or thing from the	31
			ation of this Act or any specified provision of this Act, either	32
			anditionally or subject to conditions.	33
	(3)		gulation may create an offence punishable by a penalty not	34
		exce	eding 20 penalty units.	35

	(4) A regulation may apply, adopt or incorporate any force from time to time.	publication as in 1 2
402	Repeals	3
	The Acts listed in Schedule 7 are repealed.	4
403	Amendment of other Acts	5
	Each Act listed in Schedule 8 is amended as set our	t in that Schedule.
404	Savings, transitional and other provisions	7
	Schedule 9 has effect.	8
405	Review of Act	9
	(1) The Minister is to review this Act to determine w	whether the policy 10
	objectives of the Act remain valid and whether the	
	remain appropriate for securing those objectives.	12
	(2) The review is to be undertaken as soon as possible	after the period of
	5 years from the date of assent to this Act.	14
	(3) A report on the outcome of the review is to be tabled	l in each House of
	Parliament within 12 months after the end of the pe	

Sc	hedule 1 Irrigation corporations	1
	(Section 110)	2
Col Are	leambally Irrigation Limited , in respect of the former Coleambally Irrigation ea.	3 4
Jen	nalong Wyldes Plains Irrigation Limited, in respect of:	5
(a)	the former Jemalong Domestic and Stock Water Supply and Irrigation District, and	6 7
(b)	the former Wyldes Plains Domestic and Stock Water Supply and Irrigation District.	8
Lo	wer Murray Irrigation Areas Limited, in respect of:	10
(a)	the former Buronga Irrigation Area, and	11
(b)	the former Coomealla Irrigation Area, and	12
(c)	the former Curlwaa Irrigation Area.	13
Mu	urray Irrigation Limited, in respect of:	14
(a)	the former Tullakool Irrigation Area, and	15
(b)	the former Berriquin Provisional Domestic and Stock Water Supply and Irrigation District, and	16 17
(c)	the former Denimein Provisional Domestic and Stock Water Supply and Irrigation District, and	18 19
(d)	the former Deniboota Provisional Domestic and Stock Water Supply and Irrigation District, and	20 21
(e)	the former Wakool Domestic and Stock Water Supply and Irrigation District.	22
Mu	urrumbidgee Irrigation Limited, in respect of:	23
(a)	the former Yanco No 1 Irrigation Area, and	24
(b)	the former Mirrool No 1 Irrigation Area, and	25
(c)	the former Benerembah Domestic and Stock Water Supply and Irrigation District, and	26 27
(d)	the former Tabbita Domestic and Stock Water Supply and Irrigation District, and	28 29
(e)	the former Wah Wah Domestic and Stock Water Supply and Irrigation District.	30

Schedule 2 Major utilities

Schedule 2 Major utilities		1
	(Section 279)	2
Delta Electricity		3
Hunter Water Corporation		4
Sydney Catchment Authority		5
Sydney Water Corporation		6

Water supply au	ıthorities
-----------------	------------

Schedule 3 Water supply authorities	1
(Section 298)) 2
Part 1 Bodies established by this Act	3
Benerembah Irrigation District Environment Protection Trust	4
Broken Hill Water Board	5
Cobar Water Board	6
Upper Parramatta River Catchment Trust	7
Part 2 Statutory bodies established by other Acts	8
Gosford City Council	9
Olympic Co-ordination Authority	10
Wyong Council	11

Schedul	e 4	Land exempt from water supply authority service charges	1 2
		(Section 325)	3
(1)		that is vested in the Crown, or in a public body, or in trustees, used for a public cemetery.	4 5
(2)		that is vested in the Crown, or in a public body, or in trustees, used for a common.	6 7
(3)		that is vested in the Crown, or in a public body, or in trustees, used for a public reserve or park.	8
(4)	or pub	that belongs to any public hospital, public benevolent institution blic charity, and is used or occupied by the hospital, institution or y for its purposes.	10 11 12
(5)	with,	that is used or occupied solely for the purposes of, or connected a baby health centre, day nursery, kindergarten or amenities for ed not conducted for private gain.	13 14 15
(6)	Health	that is vested in an area health service constituted under the <i>h Services Act 1997</i> and is used or occupied by the area health e for its purposes.	16 17 18
(7)		that is vested in the Crown, or in a public body, or in trustees, used solely for the purposes of a free public library.	19 20
(8)	careta	owned by the Crown that (except in the case of land leased to a ker at a nominal rental) is not leased by the Crown to any person ivate purposes.	21 22 23
(9)		that belongs to a religious body and that is occupied and used in ction with:	24 25
	(a)	any church or other building used or occupied for public worship, or	26 27
	(b)	any building used or occupied solely as the residence of a minister of religion in connection with any such church or building, or	28 29 30
	(c)	any building used or occupied for the purposes of religious teaching or training, or	31 32

	(d)	any building used or occupied solely as the residence of the official head or the assistant official head, or both, of any religious body in the State or in any diocese in the State.	1 2 3
(10)		that is a public place within the meaning of the <i>Local</i> rnment Act 1993.	4 5
(11)	Land	that:	6
	(a)	is unoccupied, and	7
	(b)	is not supplied with water from any water-pipe of a water supply authority and is not connected to any sewer of a water supply authority, and	8 9 10
	(c)	has been determined, by the council of the local government area in which the land is situated, to be unsuitable for the erection of a building because of flooding or landslip.	11 12 13
(12)	Land water	that is unoccupied and that is below highwater mark of any tidal .	14 15
(13)		that belongs to and that is occupied and used in connection with egistered non-government school under the <i>Education Act 1990</i> , ding:	16 17 18
	(a)	any playground that belongs to and is used in connection with any such school, and	19 20
	(b)	any building that is occupied as a residence by any caretaker, employee or teacher of any such school and that belongs to and is used in connection with the school.	21 22 23
(14)		that comprises any sports ground, garden or children's round provided by a council under the <i>Local Government Act</i>	24 25 26
(15)	the U	that is vested in the University of Newcastle, or in a college of niversity, and is used or occupied by the University or college of for its purposes.	27 28 29

Schedule 4	Land exempt from w	ater supply authority	service charges
------------	--------------------	-----------------------	-----------------

(16)	Land that is vested in:		
	(a) the New South Wales Aboriginal Land Council, or	2	
	(b) a Regional Aboriginal Land Council, or	3	
	(c) a Local Aboriginal Land Council,	4	
	constituted under the <i>Aboriginal Land Rights Act 1983</i> , being land that is declared under Division 5 of Part 6 of that Act to be exempt from the payment of rates under this Schedule.	5 6 7	
(17)	Land that is vested in the Mines Rescue Board and used for the purposes of a mine rescue station controlled by the Board.	8	

Schedule 5		le 5	Constitution and procedure of water supply authorities	
			(Section 301)	3
Part			sions relating to the members of a water ly authority	4 5
1	Em	ployme	ent of full-time member	6
		subjec	mployment of a full-time member of a water supply authority is ct to Part 2A of the <i>Public Sector Management Act 1988</i> , but is abject to Part 2 of that Act.	7 8 9
2	Elec	cted m	ember	10
	(1)	In this	s clause:	11
		or mo	d member, in the case of a water supply authority of which one or of the members is appointed after being elected by employees water supply authority, means a member so elected.	12 13 14
	(2)		lations may be made for or with respect to the election of a n to hold office as an elected member.	15 16
	(3)	emplo Comr and m	Electoral Commissioner for New South Wales, or a person byed in the office of and nominated by the Electoral missioner, is to be the returning officer for an election, and has may exercise the functions conferred or imposed on the returning or by the regulations made under this clause in relation to the on.	17 18 19 20 21 22
	(4)		oyees of a water supply authority are entitled to vote at an on in accordance with the regulations made under this clause.	23 24
	(5)		son's nomination as a candidate for election as elected member alid if:	25 26
		(a)	the nomination is not made by at least 2 persons who are employees of the water supply authority, or	27 28
		(b)	the person is not, at the time of nomination, an employee of the water supply authority, or	29 30

	(c)	the person is not, at the time of nomination, a member of an	1			
		industrial organisation of employees within the meaning of the	2			
		Industrial Relations Act 1996 or an association of employees registered as an organisation under the Workplace Relations Act	3 4			
		1996 of the Commonwealth, being such an organisation or	5			
		association that is prescribed by the regulations for the	6			
		purposes of this paragraph as being an organisation or	7			
		association representing employees of the water supply	8			
		authority, or	9			
	(d)	the instrument of nomination is not accompanied by a statutory	10			
		declaration to the effect that the person is, at the time of	11			
		nomination, such a member, and the returning officer is entitled	12			
		to rely on the information contained in the statutory declaration.	13			
(6)		son may be, at the same time, both the elected member and an	14			
	emplo	byee of the water supply authority.	15			
(7)	Nothi	ng in any law, rule, direction or other requirement that:	16			
	(a)	is applicable to the elected member in his or her capacity as an	17			
		employee of the water supply authority, and	18			
	(b)	would not be so applicable if the elected member were not such	19			
		an employee,	20			
		tes so as to prevent or restrict the exercise by the elected member	21			
	of any	y of the functions of an elected member.	22			
(8)		person is nominated at an election, or if for any other reason an	23			
	election fails, the Governor may appoint a person eligible for election					
	to be a part-time member, and the person, on being appointed, is taken					
	to be a person elected in the manner prescribed by the regulations					
	made under this clause.					
(9)		acancy occurs in the office of elected member otherwise than by	28			
	reason of the expiration of the period for which an elected member was appointed:					
	(a)	the Governor may appoint a person eligible for election to hold,	31			
		subject to this Schedule, the office of elected member for a term	32			
		commencing on the date of appointment or a later date	33			
		specified in the relevant instrument of appointment and ending	34			
		on the commencement of the term of office of the next elected	35			
		member, and	36			

		(b)	any such person, on being so appointed, is taken to be a person elected in the manner prescribed by the regulations made under this clause.	1 2 3
3	Cha	irpers	on of a water supply authority	4
	(1)	as suc	of the members, in and by the relevant instrument of appointment of a member, or by another instrument executed by the Governor, be appointed as chairperson of a water supply authority.	5 6
	(2)	The C	Governor may remove a member from the office of chairperson.	8
	(3)		son who is a member and chairperson is taken to have vacated as chairperson if the person:	9 10
		(a)	is removed from that office by the Governor under subclause (2), or	11 12
		(b)	resigns that office by instrument in writing addressed to the Minister, or	13 14
		(c)	ceases to be a member.	15
	(4)	during chairp	Minister may appoint a member to act in the office of chairperson g a vacancy in the office or during the illness or absence of the person and the person, while so acting, has and may exercise all unctions of the chairperson and is taken to be the chairperson.	16 17 18 19
	(5)		clause does not apply to the Broken Hill Water Board or the r Water Board constituted as provided by Parts 2 and 3 of this dule.	20 21 22
4	Acti	ing me	embers	23
	(1)	office or abs	Minister may, from time to time, appoint a person to act in the of a member during a vacancy in the office or during the illness sence of the member, and the person, while so acting, has and exercise all the functions of the member and is taken to be a per.	24 25 26 27 28
	(2)		Minister may remove any person from any office to which the n was appointed under this clause.	29 30
	(3)	to be	rson while acting in the office of a part-time member is entitled paid such remuneration (including travelling and subsistence ances) as the Minister may from time to time determine in respect e person.	31 32 33 34

5	Ter	ms of	office	1				
		Subje	ect to this Part, a member is to hold office:	2				
		(a)	in the case of a part-time member other than an elected member—for such period not exceeding 5 years, or	3				
		(b)	in the case of an elected member—for such period not exceeding 5 years but not less than 3 years,	5				
			by be specified in the instrument of appointment of the member, eligible (if otherwise qualified) for re-appointment.	8				
6	Rer	nunera	ation	Ģ				
		(inclu	art-time member is entitled to be paid such remuneration uding travelling and subsistence allowances) as the Minister may time to time determine in respect of the member.	10 11 12				
7	Cas	sual va	cancies	13				
	(1)	A member (other than a full-time member) is taken to have vacated office if the member:						
		(a)	dies, resigns or is removed from office, or	16				
		(b)	absents himself or herself from 4 consecutive meetings of the water supply authority of which reasonable notice has been given to the member personally or in the ordinary course of post, unless:	17 18 19 20				
			(i) the water supply authority has granted the member leave to be absent from those meetings, or(ii) within 4 weeks after the last of those meetings, the	21 22 23				
			member is excused by the water supply authority for having been absent from those meetings, or	24 25				
		(c)	becomes bankrupt, applies to take the benefit of any law for the relief of bankrupt or insolvent debtors, compounds with his or her creditors or makes an assignment of his or her remuneration for their benefit, or	26 27 28 29				
		(d)	becomes a mentally incapacitated person, or	30				
		(e)	is convicted in New South Wales of an offence that is punishable by imprisonment for 12 months or more or is convicted elsewhere than in New South Wales of an offence that, if committed in New South Wales, would be an offence so punishable, or	31 32 33 34 35				

		(f) being an elected member, ceases to be an employee of the water supply authority for which he or she is an elected member.	1 2 3
	(2)	The Governor may remove a part-time member from office.	4
	(3)	If the office of any member becomes vacant, a person is, subject to this Part, to be appointed to fill the vacancy.	5
8	Disc	closure of pecuniary interests	7
	(1)	A member who has a direct or indirect pecuniary interest:	8
		(a) in a matter that is being considered, or is about to be considered, at a meeting of the water supply authority, or	9 10
		(b) in a thing being done or about to be done by the water supply authority,	11 12
		must, as soon as possible after the relevant facts have come to the member's knowledge, disclose the nature of the interest at a meeting of the water supply authority.	13 14 15
	(2)	A disclosure by a member at a meeting of the water supply authority that the member:	1 <i>6</i> 17
		(a) is a member, or is in the employment, of a specified company or other body, or	18 19
		(b) is a partner, or is in the employment, of a specified person, or	20
		(c) has some other specified interest relating to a specified company or other body or a specified person,	21 22
		is taken to be a sufficient disclosure of the nature of the interest in any matter or thing relating to that company or other body or to that person that may arise after the date of the disclosure.	23 24 25
	(3)	The water supply authority must cause particulars of any disclosure made under subclause (1) or (2) to be recorded in a book kept for the purpose and that book must be open at all reasonable hours to the inspection of any person on payment of such fee as may be determined by the water supply authority from time to time.	26 27 28 29 30

	(4)	After a member has, or is deemed to have, disclosed the nature of an interest in any matter or thing under subclause (1) or (2), the member must not, unless the Minister otherwise determines:	1 2 3		
		(a) be present during any deliberation of the water supply authority, or take part in any decision of the water supply authority, with respect to that matter, or	4 5		
		(b) exercise any functions under this Act with respect to that thing,	7		
		as the case requires.	8		
	(5)	A contravention of this clause does not invalidate any decision of the water supply authority or the exercise of any function under this Act.	9 10		
	(6)	A reference in this clause to a meeting of the water supply authority includes a reference to a meeting of a committee of the water supply authority.	11 12 13		
9	Effe	ect of certain other Acts	14		
	(1)	Part 2 of the <i>Public Sector Management Act 1988</i> does not apply to or in respect of the appointment of a member (other than a full-time member) and the office of such a member is not, for the purposes of any Act, an office or place of profit under the Crown.			
	(2)	If by or under any other Act provision is made:	19		
		(a) requiring a person who is the holder of a specified office to devote the whole of his or her time to the duties of that office, or	20 21 22		
		(b) prohibiting the person from engaging in employment outside the duties of that office,	23 24		
		that provision does not operate to disqualify the person from holding that office and also the office of a part-time member or from accepting and retaining any remuneration payable to the person under this Part as a part-time member.	25 26 27 28		

Part	2 E	Broke	en Hill Water Board	1
Divis	ion 1		Constitution of the Board	2
10	Mer	nbers	of the Broken Hill Water Board	3
	(1)		Broken Hill Water Board is to consist of 6 members appointed by dovernor.	4 5
	(2)	Of th	e members:	6
		(a)	one, in and by the instrument by which the member is appointed or by another instrument, is to be appointed as President of the Board, and	7 8 9
		(b)	five, in and by the instruments by which the members are appointed, are to be appointed as part-time members.	10 11
	(3)	Of th	e part-time members:	12
		(a)	one is to be a member of the Public Service who, in and by the instrument by which the member is appointed or by another instrument, is to be appointed as Vice-President of the Board, and	13 14 15 16
		(b)	two are to be nominated in accordance with subclause (4) by the Council of the City of Broken Hill or, in the absence of a nomination, by the Minister, and	17 18 19
		(c)	two are to be selected by the Minister from a panel of 4 persons jointly nominated by the companies listed in clause 11.	20 21
	(4)	purpo	inations (other than a nomination by the Minister) for the oses of subclause (3) are to be made as prescribed by the ations.	22 23 24
	(5)	The F	President of the Board is a full-time member.	25
	(6)	and c	Vice-President of the Board ceases to be a member of the Board, eases to hold office as Vice-President, if he or she ceases to be a ber of the Public Service.	26 27 28

11	Cor	npanies nominating panel of candidates for appointment	1
	(1)	The companies entitled to nominate a panel for the purposes of clause 10 (3) (c) are:	2
		(a) Pasminco Australia Limited, and	4
		(b) Minerals Mining and Metallurgy Limited, and	5
		(c) any other company carrying on mining operations at Broken Hill that is considered by the Minister to be entitled to join in nominating the panel.	6 7 8
	(2)	If, in the opinion of the Minister, a company referred to in subclause (1) has ceased to carry on mining operations at Broken Hill, the company ceases to be entitled to join in nominating a panel.	9 10 11
12	Pre	sident and Vice-President	12
	(1)	The President, if present, is to preside at a meeting of the Broken Hill Water Board.	13 14
	(2)	In the absence of the President, the Vice-President may act as President and, while so acting, has and may exercise the functions of the President.	15 16 17
	(3)	In the absence of the President and the Vice-President from a meeting of the Board, a member elected from among those present is to preside at the meeting.	18 19 20
Divis	ion 2	The Treasurer and the mining companies	21
13	Def	initions	22
	(1)	In this Division:	23
		council means the Council of the City of Broken Hill.	24
		<i>financial year</i> means the financial year for the Broken Hill Water Board.	25 26
		<i>mining company</i> means a company for the time being prescribed as a mining company for the purposes of this Division.	27 28
	(2)	Until the regulations otherwise provide, the following companies are mining companies for those purposes:	29 30
		Pasminco Australia Limited	31
		Minerals Mining and Metallurgy Limited	32

	(3)		ompany ceases to be a mining company for the purposes of this ion, it remains responsible for any undischarged liability to which	1 2			
			s subject under this Division while it was a mining company.	3			
14	Exe	mptio	n from service charges	4			
	(1)	Land	at Broken Hill that is held by a mining company under a lease	5			
			the Crown, and is used for mining or incidental purposes, is not ct to a water service charge.	6 7			
	(2)		referred to in subclause (1) is not subject to a sewerage service ge because of the construction of any sewerage works.	8 9			
15	Sup	ply of	water to mining companies	10			
	(1)	The I	Broken Hill Water Board must supply each mining company, at	11			
			s as near as possible to the mines or works of the company, with	12			
			rater required by the company to carry on its business.	13			
	(2)	Subcl	lause (1) has effect only in so far as is permitted by:	14			
		(a)	the capacity of the works of the Board, and	15			
		(b)	the requirements of the landholders of land and of other	16			
			persons using water for domestic purposes.	17			
16	Min	ing co	mpanies to take water	18			
	(1)		ning company is to draw from the Broken Hill Water Board all	19			
			rater that it requires for the purposes of its business and that the	20			
			d is able to supply.	21			
	(2)	Subclause (1) does not prevent a mining company from drawing water					
		from:		23			
		(a)	supplies conserved in its own tanks and dams, or	24			
		(b)	sources other than Stephens Creek, by means lawfully used by it on and after 21 December 1915 and before 1 January 1939.	25 26			
17	Wat	ter cha	arges	27			
	(1)		charge for water supplied by the Broken Hill Water Board to a	28			
		minir	ng company is to be:	29			
		(a)	11 cents per kilolitre unless it is water referred to in paragraph (b), or	30 31			

		(b)	5.5 cents per kilolitre if it is used only for a dwelling or plantation or any other purpose determined by the Board to be a social purpose.	1 2 3
	(2)	it to a	ermination by the Board of the proportion of water supplied by mining company that is water referred to in subclause (1) (b) is ng on the company.	4 5 6
	(3)		harge imposed by subclause (1) is taken to be a charge imposed e Board for water supplied to the land referred to in clause 14.	7 8
	(4)	No lia	ability for a service charge arises in relation to land that:	9
		(a)	adjoins the rising main between Menindee and the pumping station at Stephens Creek, and	10 11
		(b)	is part of a holding under the Western Lands Act 1901 or a part of land held under freehold title,	12 13
			s the whole of the holding or freehold land of which it forms part hin the area of operations of the Board.	14 15
18	Add	litional	payments to Water Fund	16
	(1)	water	er excluding income received from the mining companies for supplied to them during a financial year, there is a deficiency for ear in the Water Fund of the Broken Hill Water Board:	17 18 19
		(a)	the Treasurer, out of money to be provided by Parliament, may pay to the Board towards the deficiency thirteen fifty-ninths of its amount, and	20 21 22
		(b)	each mining company must, on demand, pay to the Board the prescribed contribution for the company.	23 24
	(2)		prescribed contribution for a mining company towards the ency for a financial year is the amount ascertained by:	25 26
		(a)	calculating the amount that bears to forty-six fifty-ninths of the deficiency the same proportion as is borne by the amount payable by the company under clause 17 in that year to the total amount so payable by all the mining companies, and	27 28 29 30
		(b)	deducting from the amount so calculated the amount paid by the company under clause 17 in that year.	31 32
	(3)		mount payable by a mining company under this clause is taken a charge imposed by the Board for water supplied by the Board.	33 34

19	Esti	mated deficit an	d advance payments	1
	(1)	commencement	epared by the Broken Hill Water Board before the of a financial year disclose the possibility of a deficit the Water Fund of the Board:	2 3 4
		` '	rd is to provide the Treasurer and each mining with particulars of the estimates and the possible and	5 6 7
			surer and each mining company must, in that year, yments to the Board as required by subclause (2).	8
	(2)		rging the respective liabilities of the Treasurer and each y for a financial year:	10 11
		provided	surer must pay to the Board out of money to be by Parliament the amount estimated by the Treasurer ficient to discharge the liability of the Treasurer, and	12 13 14
		estimated	ning company must pay to the Board the amount d by the Board to be sufficient to discharge the liability ining company,	15 16 17
		by equal instaln that year.	nents on 1 January, 1 April, 1 July and 1 October in	18 19
	(3)	On the auditing of the Board's accounts for a financial year, the Board must pay to, or receive from, the Treasurer and each mining company an amount in adjustment of any difference between the estimated and actual liabilities under clause 18 for that financial year.		20 21 22 23
20	Sev	erage charges		24
	(1)	Broken Hill Wa \$6,400 the same the council for o	mpany must, in respect of each financial year of the ster Board, pay to the Board an amount that bears to e proportion as the amount payable by the company to ordinary rates bears to the total amount payable by all panies to the council for ordinary rates for that year.	25 26 27 28 29
	(2)	to be a charge i	able by a mining company under subclause (1) is taken mposed by the Board for sewerage services for that of the land referred to in clause 14.	30 31 32

Schedule 5	Constitution	and	procedure	of	water	supply	authoritie	es.

	(3)		a financial year of the Board, there is a deficiency in its Sewerage	1
			, the Treasurer, out of money to be provided by Parliament, may of the Board for application towards the deficiency:	2
		(a)	•	3
		(a) (b)	the amount of the deficiency, or \$6,000,	4
		` /		5
			hever is the lesser amount.	6
	(4)		ey at credit in the Broken Hill Water Administration Account in	7
			pecial Deposits Account at the Treasury is, at the request of the d, to be paid to the Board for the renewal or replacement of	8 9
			al assets.	10
21	Def	ault by	/ mining company	11
		-	Broken Hill Water Board:	12
		(a)	has obtained judgment against a mining company for an	13
		(4)	amount payable under this Division, and	14
		(b)	the judgment remains unsatisfied even though the Board has	15
			taken all reasonable steps to enforce it,	16
			oard may recover the unpaid amount from each company not in	17
		defau Divis	ilt rateably in proportion to each amount paid by it under this	18
		DIVIS	IOII.	19
Dart	3 (` oha	r Water Board	20
ıarı		JUDA	Water Board	20
22	Mer	nbers	of the Cobar Water Board	21
	(1)		Cobar Water Board is to consist of 5 part-time members	22
		appoi	inted by the Governor.	23
	(2)	Of th	e members:	24
		(a)	one is to be a member of the Public Service who, in and by the	25
			instrument by which the member is appointed, is to be	26
		<i>a</i> >	appointed as President of the Board, and	27
		(b)	one is to be selected by the Minister from a panel of 2 persons nominated jointly by:	28 29
			(i) Peak Gold Mines Pty Limited, and	30
			(ii) such other mining companies as the Minister	31
			determines, and	32

	(3)		nominat (i) I (ii) s 2 are to	ted jointly by Pasminco Ausuch other determines, a be nominated under subc	y: ustralia Li mining and ed by the 0 lause (2) 0	companies Cobar Shire (b), (c) or (d)	as Counc	the cil.	Minister de within	2 3 4 5 6	
				•	ster, a pers	son nominate	ed by	the N	Iinister is		
		to be	appointed	1.						9	
		_		D							
23	Vac	ation o	of office k	oy Presiden	t					10)
23	Vac			•		of the Coba	ır Wat	ter B	oard if he		
23	Vac	The F	resident o	ceases to be	a member	of the Coba Public Service		ter B	oard if he		
23		The For she	resident o	ceases to be be a memb	a member			ter B	oard if he	11	
		The For she	resident of ceases to	ceases to be to be a member a	a member per of the		ce.			11 12	:
	Mee	The For she	resident of ceases to ceases to resident,	ceases to be to be a member a	a member per of the	Public Servic	ce.			11 12	;

Sch	edu	le 6	Water Advisory Council and management committees	1 2
			(Sections 10 and 380)	3
Part	1 (Const	titution	4
1	App	licatio	n of Schedule	5
			Schedule applies to the Water Advisory Council and all gement committees.	<i>6</i>
2	Act	ng cha	airperson	8
	(1)	office the pe	Minister may, from time to time, appoint a person to act in the of chairperson during the illness or absence of a chairperson, and erson, while so acting, has all the functions of the chairperson and en to be the chairperson.	9 10 11 12
	(2)		Minister may, at any time, remove a person from an office to a the person has been appointed under this clause.	13 14
	(3)		the purposes of this clause, a vacancy in the office of chairperson ten to be an absence from office of the chairperson.	15 16
3	Teri	n of of	ifice	17
		Subje	ct to this Schedule:	18
		(a)	a member of the Water Advisory Council holds office for 3 years, but is eligible (if otherwise qualified) for re-appointment, and	19 20 21
		(b)	a member of a management committee holds office during the term for which the committee is established.	22 23
4	Ren	nunera	ition	24
		comm	ember of the Water Advisory Council or a management nittee is entitled to be paid such remuneration (including ling and subsistence allowances) as the Minister may from time the determine in respect of the member.	25 26 27

5	Vac	ancy i	n office of appointed member	1
	(1)	The office of a member of the Water Advisory Council or a management committee becomes vacant if the member:		
		(a)	dies, resigns or is removed from office, or	3 4
		(b)	absents himself or herself from 4 consecutive meetings of the Council or committee of which reasonable notice has been	5 6
			given to the member personally or in the ordinary course of	7
			post, unless:	8
			(i) the Council or committee has granted the member leave	9
			to be absent from those meetings, or	10
			(ii) within 4 weeks after the last of those meetings, the	11
			member is excused by the Council or committee for having been absent from those meetings, or	12 13
		()		
		(c)	becomes bankrupt, applies to take the benefit of any law for the relief of bankrupt or insolvent debtors, compounds with his or	14 15
			her creditors or makes an assignment of his or her	16
			remuneration for their benefit, or	17
		(d)	becomes a mentally incapacitated person, or	18
		(e)	is convicted in New South Wales of an offence that is	19
		(0)	punishable by imprisonment for 12 months or more or is	20
			convicted elsewhere than in New South Wales of an offence	21
			that, if committed in New South Wales, would be an offence so	22
			punishable.	23
	(2)	The N	Minister may remove a member of the Water Advisory Council	24
	` '	or a n	nanagement committee from office at any time.	25
6	Filli	ng of v	vacancy in office of member	26
		If the	e office of a member of the Water Advisory Council or a	27
		management committee becomes vacant, a person is, subject to this		
		Act, t	o be appointed to fill the vacancy.	29
7	Disc	closure	e of pecuniary interests	30
	(1)	A m	ember of the Water Advisory Council or a management	31
	` /		nittee who has a direct or indirect pecuniary interest:	32
		(a)	in a matter that is being considered, or is about to be	33
			considered, at a meeting of the Council or committee, or	34

	(b)	in a thing being done or about to be done by the Council or committee,	1 2	
	memb	as soon as possible after the relevant facts have come to the per's knowledge, disclose the nature of the interest at a meeting Council or committee.	3 4 5	
(2)	A dis	closure at such a meeting that the member:	6	
	(a)	is a member, or is in the employment, of a specified company or other body, or	7 8	
	(b)	is a partner, or is in the employment, of a specified person, or	9	
	(c)	has some other specified interest relating to a specified company or other body or a specified person,	10 11	
	matte	en to be a sufficient disclosure of the nature of the interest in any r or thing relating to that company or other body or to that person hay arise after the date of the disclosure.	12 13 14	
(3)	partic record all rea	Vater Advisory Council or management committee must cause ulars of any disclosure made under subclause (1) or (2) to be led in a book kept for the purpose and that book must be open at asonable hours to the inspection of any person on payment of fee as may be determined by it from time to time.	15 16 17 18 19	
(4)	After a member has, or is deemed to have, disclosed the nature of an interest in any matter or thing under subclause (1) or (2), the member must not, unless the Minister otherwise determines:			
	(a)	be present during any deliberation, or take part in any decision, of the Water Advisory Council or management committee with respect to that matter, or	23 24 25	
	(b)	exercise any functions under this Act with respect to that thing,	26	
	as the	case requires.	27	
(5)	Water	atravention of this clause does not invalidate any decision of the Advisory Council or management committee or the exercise of unction under this Act.	28 29 30	
(6)	or a n	erence in this clause to a meeting of the Water Advisory Council nanagement committee includes a reference to a meeting of any committees.	31 32 33	

8	Effe	ect of c	certain other Acts	1
	(1)	Part 2 of the <i>Public Sector Management Act 1988</i> does not apply to or in respect of the appointment of a member of the Water Advisory		2 3
			cil or a management committee and the office of such a member	4
		is not, for the purposes of any Act, an office or place of profit under		5
		the Crown.		
	(2)	If by	or under any other Act provision is made:	7
		(a)	requiring a person who is the holder of a specified office to	8
			devote the whole of his or her time to the duties of that office,	9
			or	10
		(b)	prohibiting the person from engaging in employment outside	11
			the duties of that office,	12
		that p	provision does not operate to disqualify the person from holding	13
			office and also the office of a member of the Water Advisory	14
		Council or a management committee, or from accepting and retaining any remuneration payable to the person under this Part as such a		15
				16
		meml	per.	17
Part	2 I	Proce	edure	18
9	Ger			
		neral p	rocedure	19
		-	rocedure pt as otherwise provided by this Act or the regulations:	19 20
		Excep	pt as otherwise provided by this Act or the regulations:	20
		-	pt as otherwise provided by this Act or the regulations: meetings of the Water Advisory Council or a management	
		Excep	pt as otherwise provided by this Act or the regulations:	20 21
		Except (a)	pt as otherwise provided by this Act or the regulations: meetings of the Water Advisory Council or a management committee are to be held at such times and places as are fixed by the chairperson, and	20 21 22
		Excep	pt as otherwise provided by this Act or the regulations: meetings of the Water Advisory Council or a management committee are to be held at such times and places as are fixed	20 21 22 23
		Except (a)	pt as otherwise provided by this Act or the regulations: meetings of the Water Advisory Council or a management committee are to be held at such times and places as are fixed by the chairperson, and the procedure for the convening of meetings and for the	20 21 22 23 24
10		Except (a)	pt as otherwise provided by this Act or the regulations: meetings of the Water Advisory Council or a management committee are to be held at such times and places as are fixed by the chairperson, and the procedure for the convening of meetings and for the conduct of business at those meetings is to be as determined by	20 21 22 23 24 25
10		Excep (a) (b)	pt as otherwise provided by this Act or the regulations: meetings of the Water Advisory Council or a management committee are to be held at such times and places as are fixed by the chairperson, and the procedure for the convening of meetings and for the conduct of business at those meetings is to be as determined by the chairperson.	20 21 22 23 24 25 26
10		Excep (a) (b) orum A ma	pt as otherwise provided by this Act or the regulations: meetings of the Water Advisory Council or a management committee are to be held at such times and places as are fixed by the chairperson, and the procedure for the convening of meetings and for the conduct of business at those meetings is to be as determined by	20 21 22 23 24 25 26
10		Excep (a) (b) orum A ma	pt as otherwise provided by this Act or the regulations: meetings of the Water Advisory Council or a management committee are to be held at such times and places as are fixed by the chairperson, and the procedure for the convening of meetings and for the conduct of business at those meetings is to be as determined by the chairperson. ajority of the members of the Water Advisory Council or a gement committee constitute a quorum, so long as of those	20 21 22 23 24 25 26 27 28
10		Excep (a) (b) orum A ma mana	pt as otherwise provided by this Act or the regulations: meetings of the Water Advisory Council or a management committee are to be held at such times and places as are fixed by the chairperson, and the procedure for the convening of meetings and for the conduct of business at those meetings is to be as determined by the chairperson. ajority of the members of the Water Advisory Council or a gement committee constitute a quorum, so long as of those	20 21 22 23 24 25 26 27 28 29

Schedule 6	Water A	dvisorv	Council	and i	management	committees
Ochiedale 0	vvalci A	tu visoi y	Council	anui	management	COMMITTEE

		(b)	at least one is a member appointed to represent the interests of water user groups, and	1 2
		(c)	at least one is a member appointed to represent the interests of local councils, and	3
		(d)	in the case of a management committee, at least one is a member of staff of the Department.	5 6
11	Pres	siding	members	7
		appoi	chairperson (or, in the absence of the chairperson, a member nted by the members then present) is to preside at a meeting of vater Advisory Council or a management committee.	8 9 10
12	Dec	isions		11
	(1)	Water	ision supported by a majority of the votes cast at a meeting of the Advisory Council or a management committee at which a m is present is the decision of the Council or committee.	12 13 14
	(2)		case of an equality of votes, the presiding member has a second sting vote.	15 16
13	Rec	ord of	proceedings	17
	(1)	a man	residing member at a meeting of the Water Advisory Council or agement committee must cause a record of the proceedings at the ng to be made.	18 19 20
	(2)		rds made for the purposes of this clause may be destroyed after apiry of the period prescribed by the regulations.	21 22
14	Aut	hentica	ation of documents	23
			locument requiring authentication by the Water Advisory Council nanagement committee is sufficiently authenticated if it is signed	24 25 26
		(a)	the member who presided at the meeting that dealt with the proceedings with respect to which the document was prepared, or	27 28 29
		(b)	in the absence of that member, any other member who was present at that meeting.	30 31

Mater A	Advisorv	Council	and	management	committees
vvalei <i>i</i>	4UVI5UIV	Council	anu	manauement	COMMITTEES

	ıed		

15	First meeting	1
	The first meeting of a management committee is to be called by the	2
	Minister in such manner as the Minister considers appropriate.	3

Schedule 7 Repeals

Schedule 7 Repeals		1
	(Section 402)	2
Balranald Irrigation Act 1902 No 78		3
Drainage Act 1939 No 29		4
Fish River Water Supply Administration Act 1945 No 16		5
Glennies Creek Dam Act 1979 No 126		6
Hunter Valley Flood Mitigation Act 1956 No 10		7
Irrigation Act 1912 No 73		8
Irrigation (Amendment) Act 1916 No 22		9
Irrigation (Amendment) Act 1941 No 54		10
Irrigation (Amendment) Act 1973 No 3		11
Irrigation (Amendment) Act 1979 No 160		12
Irrigation and Water (Amendment) Act 1943 No 2		13
Irrigation Corporations Act 1994 No 41		14
Irrigation, Water and Rivers and Foreshores Improvement (Ar. No 12		15 16
Miscellaneous Acts (Water Administration) Amendment Act 1	986 No 205	17
Private Irrigation Districts Act 1973 No 47		18
Rivers and Foreshores Improvement Act 1948 No 20		19
Water Act 1912 No 44		20
Water (Amendment) Act 1936 No 31		21
Water (Amendment) Act 1940 No 57		22
Water (Amendment) Act 1976 No 33		23
Water (Amendment) Act 1979 No 159		24
Water (Soil Conservation) Amendment Act 1986 No 143		25
Water Administration Act 1986 No 195		26
Water Administration (Transfer of Functions) Act 1986 No 20	04	27
Water Supply Authorities Act 1987 No 140		28

Sch	edule 8 Amendment of other Acts	1
		(Section 403)
8.1	Catchment Management Act 1989 No 235	3
[1]	Section 6 Responsible Minister	4
	Omit "Water Administration Act 1986" wherever occurring. Insert instead "Water Management Act 2000".	5
[2]	Section 9 Membership of Co-ordinating Committee	7
	Omit "Water Administration Act 1986" from section 9 (1) (g). Insert instead "Water Management Act 2000".	
[3]	Section 38 Definitions	10
	Omit "Water Supply Authorities Act 1987" from the definition Insert instead "Water Management Act 2000".	n of <i>owner</i> . 11
8.2	Chipping Norton Lake Authority Act 1977 No 38	13
	Section 25 Application of Rivers and Foreshores Improvements	ent Act 14
	Omit the section.	16
8.3	Contaminated Land Management Act 1997 No 140	17
	Section 18 Details of investigation order	18
	Omit "Water Act 1912" from section 18 (3) (c).	19
	Insert instead "Water Management Act 2000".	20

Insert instead "Water Management Act 2000".

8.4	Criminal Procedure Act 1986 No 209	1
	Schedule 1 Indictable offences triable summarily	2
	Omit item 25 of Table 1.	3
8.5	Crown Lands Act 1989 No 6	4
	Section 172 Land with boundaries to lakes, roads etc	5
	Omit "Water Act 1912" from section 172 (9). Insert instead "Water Management Act 2000".	6
8.6	Crown Lands (Continued Tenures) Act 1989 No 7	8
	Section 3 Definitions	Ģ
	Omit "Water Administration Act 1986" from the definition of Water Administration Ministerial Corporation in section 3 (1). Insert instead "Water Management Act 2000".	10 11 12
8.7	Dams Safety Act 1978 No 96	13
	Section 8 Members	14
	Omit "Water Administration Act 1986" from section 8 (2) (c).	15

8.8	Environmental Plann	ing and Assessme	nt Act 1979 No 203	1			
[1]	Section 91 What is "int	egrated developmen	t"?	2			
			provals appearing at the end	3 4 5			
	Insert in alphabetical ord	Insert in alphabetical order:					
	Water Management Act 2000	ss 80, 81, 82	water use approval, water management work approval or activity approval under Part 3 of Chapter 3	7 8 9			
[2]	Section 120A Additional is involved	procedural requirem	ents where water approval	10 11			
	Omit "Water Administra Insert instead "Minister	-	oration" wherever occurring.	12 13			
[3]	Section 120A (1)			14			
	require a water approval	". olve the need for an	r licence or a work that may approval under the <i>Water</i>	15 16 17 18			
[4]	Section 120A (3)			19			
	Omit "for a water licenc Insert instead "for an app	**	val".	20 21			

[5]	Section 12	ΛΔ (4)	and (5)	1
[2]			.,	1
	Omit the subsections. Insert instead:			
	(4)		Commission of Inquiry must defer concluding its inquiry afficient time to enable:	3 4
		(a)	the applicant or proponent to apply for an approval, and	5
		(b)	any objectors to object to the granting of an approval,	6
		under	the Water Management Act 2000.	7
	(5)	application the W must in the its intapplication application applications.	coon as practicable after the applicant's or proponent's cation for an approval is referred to it under section 86 of <i>Vater Management Act 2000</i> , the Commission of Inquiry give at least 28 days' notice, by advertisement published a Gazette and in such newspapers as it thinks necessary, of tention to hold a public hearing in connection with the cation concerned and of the time and place at which the ng is to be held.	8 9 10 11 12 13 14
[6]	Section 12	0A (7)	(a)	16
	Omit "for a	water	licence or for a water approval".	17
			an approval".	18
[7]	Section 12	0A (7)	(b)	19
	Omit the pa	aragrap	bh. Insert instead:	20
		(b)	any objection to the granting of an approval that has been referred to it under section 86 of the <i>Water Management Act 2000</i> .	21 22 23
[8]	Section 12	0A (8)	(a)	24
	Omit "a wa	ter lice	ence or a water approval". Insert instead "an approval".	25
[9]	Section 12	0A (8)	(b)	26
	Omit "licen	ice or"		27

[10]	Section 120A (11)	1
	Omit "a water licence or a water approval". Insert instead "an approval".	2
[11]	Section 120A (12)	3
	Omit the subsection.	4
8.9	Essential Services Act 1988 No 41	5
	Section 4 Essential services	6
	Insert after section 4 (2):	7
	(2A) To avoid doubt, the regulation of bulk water supply by the Water Administration Ministerial Corporation in the exercise of its rights to the control, use and flow of water is capable of being declared to be an essential service for the purposes of this Act.	8 9 10 11 12
8.10	Farm Water Supplies Act 1946 No 22	13
[1]	Section 16 Works to be approved	14
	Omit "No work to which Part 2, 5 or 8 of the Water Act 1912 extends".	15
	Insert instead "No work that is a water management work within the meaning of the <i>Water Management Act 2000</i> ".	16 17
[2]	Section 16	18
	Omit "licence, permit, authority or".	19
[3]	Section 16A Loan to Board	20
	Omit the definition of <i>Board</i> . Insert instead:	21
	Board means a private irrigation board within the meaning of the <i>Water Management Act 2000</i> .	22 23

Schedule 8	Amendment	of other	Acto
ochequie o	Amenament	or other	ACIS

[4]	Section 16A (5)	1
	Omit "group licence issued under Division 4A of Part 2 of the <i>Water Act</i> 1912".	2 3
	Insert instead "water supply work approval under the <i>Water Management Act 2000</i> ".	4 5
8.11	Hunter Water Act 1991 No 53	6
[1]	Section 3 Definitions	7
	Omit "Water Administration Act 1986" from the definition of Water Administration Ministerial Corporation in section 3 (2). Insert instead "Water Management Act 2000".	8 9 10
[2]	Section 4C Role of certain Ministers	11
	Omit "Water Administration Act 1986" from section 4C (1) (b). Insert instead "Water Management Act 2000".	12 13
[3]	Section 16 Area covered by operating licence	14
	Omit "referred to in section 11 (2) of the <i>Water Administration Act 1986</i> " from section 16 (2) (b). Insert instead "within the meaning of the <i>Water Management Act 2000</i> ".	15 16 17
[4]	Section 16 (2) (c)	18
	Omit the paragraph. Insert instead:	19
	(c) a water supply authority within the meaning of the Water Management Act 2000,	20 21
[5]	Section 53 Special areas	22
	Omit "powers conferred on the Water Administration Ministerial Corporation by section 12 of the <i>Water Administration Act 1986</i> in relation to the use, flow and control of water" from section 53 (2).	23 24 25
	Insert instead "State's water rights under the Water Management Act 2000".	26

8.12		
[1]		
	Insert after section 4 (5):	3
	(6) To avoid doubt, the services for which fees and charges are payable under Chapter 3 of the <i>Water Management Act 2000</i> are capable of being declared to be government monopoly services provided by the Water Administration Ministerial Corporation.	4 5 6 7 8
[2]	Schedule 1 Government agencies for which Tribunal has standing reference	9 10
	Omit "Water Supply Authorities Act 1987". Insert instead "Water Management Act 2000".	11 12
[3]	Schedule 1	13
	Omit the following: Administrator of the South-west Tablelands Water Supply Administrator of the Fish River Water Supply	14 15 16
8.13	Land and Environment Court Act 1979 No 204	17
[1]	Section 17 Class 1—environmental planning and protection appeals	18
	Omit section 17 (cb), (cc) and (da). Insert after section 17 (b):	19 20
	(c) appeals under section 295 or 379 of the <i>Water Management Act 2000</i> ,	21 22
[2]	Section 18 Class 2—local government and miscellaneous appeals and applications	23 24
	Omit section 18 (a2). Insert instead:	25

Schedule 8	Amendment o	fathar Aata
SCHEOLIE A	AMENOMEN O	i oinei acis

	(a2) appeals under section 320 of the <i>Water Management Act</i> 2000,	1 2
[3]	Section 19 Class 3—land tenure, valuation, rating and compensation matters	3 4
	Omit section 19 (e1).	5
[4]	Section 20 Class 4—environmental planning and protection and development contract civil enforcement	6 7
	Omit section 20 (1) (cj) and (df).	8
[5]	Section 20 (3) (a)	9
	Omit "Rivers and Foreshores Improvement Act 1948,".	10
[6]	Section 21 Class 5—environmental planning and protection summary enforcement	11 12
	Omit section 21 (ga) and (hb).	13
8.14	Land Tax Management Act 1956 No 26	14
[1]	Section 62B Expenditure for which allowance is to be made	15
	Omit "Division 4 or 4A of Part 2 of the <i>Water Act 1912</i> " from section 62B (1) (c). Insert instead "the <i>Water Management Act 2000</i> ".	16 17 18
[2]	Section 62G Apportionment of joint expenditure on water supply scheme	19 20
	Omit "Division 4 or 4A of Part 2 of the Water Act 1912".	21
	Insert instead "the Water Management Act 2000".	22

8.15	Local Government Act 1993 No 30		
[1]	Section 5	66 Application of Division	2
	Omit the	section. Insert instead:	3
	56 Ap	oplication of Division	4
	(1	The provisions of this Division relating to water supply and sewerage (but not stormwater drainage) do not apply to:	5 6
		(a) land within the area of operations of the Sydney Water Board under the <i>Sydney Water Act 1994</i> , or	7 8
		(b) land within the area of operations of the Hunter Water Board under the <i>Hunter Water Act 1991</i> .	9 10
	(2	The provisions of this Division (sections 57, 58 and 59 excepted) relating to water supply and sewerage (but not stormwater drainage) do not apply to land within the area of operations of a water supply authority constituted under the <i>Water Management Act 2000</i> .	11 12 13 14 15
[2]	Section 6	64 Construction of works for developers	16
	Omit the	section. Insert instead:	17
	64 Co	onstruction of works for developers	18
		Division 5 of Part 2 of Chapter 6 of the <i>Water Management Act 2000</i> applies to a council exercising functions under this Division in the same way as it applies to a water supply authority exercising functions under that Act.	19 20 21 22
[3]	Section council?	68 What activities, generally, require the approval of the	23 24
	Omit sect	ion 68 (2) and (3). Insert instead:	25
	(2	This section does not apply to the carrying out of an activity specified in Part B of the following Table:	26 27
		(a) on land within the area of operations of the Sydney Water Board under the <i>Sydney Water Act 1994</i> , or	28 29

		(b) on land within the area of operations of the Hunter Water Board under the <i>Hunter Water Act 1991</i> .	1 2
	(3)	This section does not apply to the carrying out of an activity	3
		specified in item 1, 2, 3, 4 or 6 of Part B of the following Table	4
		on land within the area of operations of a water supply authority constituted under the <i>Water Management Act 2000</i> .	5 6
[4]	Section 12	24 Orders	7
		tence granted under Part 2 of the <i>Water Act 1912</i> " from item 11 de to the section.	8 9
		ad "a water management work approval granted under the Water	10
	Manageme	ent Act 2000'.	11
[5]	Dictionary		12
	Omit the d	efinition of water supply authority. Insert instead:	13
		water supply authority has the same meaning as it has in the	14
		Water Management Act 2000.	15
8.16	Murray-D	Darling Basin Act 1992 No 65	16
[1]	Section 4	Definitions	17
	Omit "Wa	ter Administration Act 1986" from the definition of relevant	18
	water auth	<i>pority</i> in section 4 (1).	19
	Insert inste	ad "Water Management Act 2000".	20
[2]		Powers, liabilities and immunities under Water Management and other Acts	21 22
	Omit "Wat	ter Administration Act 1986".	23
	Insert inste	ad "Water Management Act 2000".	24
[3]	Section 23	Actions against relevant water authority	25
	Omit "Sect	tion 19 (Exclusion of liability) of the Water Administration Act	26
	1986".	•	27
	Insert inste	ad "Section 399 of the Water Management Act 2000".	28

8.17	National Parks and Wildlife Act 1974 No 80	1
[1]	Section 5 Definitions	2
	Omit "Irrigation Act 1912" from section 5 (2) (b). Insert instead "Water Management Act 2000".	3 4
[2]	Section 47J Provisions relating to mining	5
	Omit "Irrigation Act 1912" from section 47J (7) (b). Insert instead "Water Management Act 2000".	6 7
[3]	Section 47J (7) (c)	8
	Omit "Irrigation Corporations Act 1994". Insert instead "Water Management Act 2000".	9 10
8.18	Native Vegetation Conservation Act 1997 No 133	11
	Section 12 Clearing excluded from operation of Act	12
	Omit section 12 (o) and (p). Insert instead:	13
	(o) any clearing carried out in accordance with an approval under the <i>Water Management Act 2000</i> .	14 15
8.19	New South Wales—Queensland Border Rivers Act 1947 No 10	16
[1]	Section 4 Definitions	17
	Omit "Water Administration Act 1986" from the definition of Ministerial Corporation.	18 19
	Insert instead "Water Management Act 2000".	20

[2]	Section 24 Water Management Act 2000 to be read subject to the Agreement	1 2
	Omit "Water Act 1912, as amended by subsequent Acts,".	3
	Insert instead "Water Management Act 2000".	4
8.20	Plantations and Reafforestation Act 1999 No 97	5
[1]	Section 19 Related regulatory approvals for authorised plantations	6
	Omit "licences under the <i>Water Act 1912</i> " from the note to section 19 (1).	7
	Insert instead "licences and approvals under the Water Management Act 2000".	8
[2]	Section 52	10
	Omit the section. Insert instead:	11
	52 Plantation operations not subject to certain provisions of the Water Management Act 2000	12 13
	(1) A controlled activity approval is not required under the <i>Water Management Act 2000</i> for plantation operations on an authorised plantation.	14 15 16
	(2) A direction cannot be given under Part 1 of Chapter 7 of that Act so as to prevent or interfere with the carrying out of plantation operations on an authorised plantation.	17 18 19
8.21	Roads Act 1993 No 33	20
[1]	Section 99 Private bodies to maintain or repair certain water supply and drainage works	21 22
	Omit "a private body established under the <i>Water Act 1912</i> , the <i>Drainage Act 1939</i> or the <i>Private Irrigation Districts Act 1973</i> or the <i>Irrigation Corporations Act 1994</i> ".	23 24 25
	Insert instead "an irrigation corporation, a private irrigation board, a private drainage board or a private water trust (within the meaning of the <i>Water Management Act 2000</i>)".	26 27 28

[2]	Dictionary	1
	Omit ", the Irrigation Act 1912" from the definition of Crown Lands Acts.	2
8.22	Rural Assistance Act 1989 No 97	3
	Section 3 Definitions	4
	Omit paragraph (a) of the definition of <i>Crown Lands Acts</i> in section 3 (1). Insert instead:	5
	(a) the Crown Lands Acts within the meaning of the <i>Crown Lands Act 1989</i> , or	7 8
8.23	Snowy Hydro Corporatisation Act 1997 No 99	9
[1]	Section 20 Water inquiry to be held	10
	Omit "Water Administration Act 1986" from section 20 (6). Insert instead "Water Management Act 2000".	11 12
[2]	Section 23 Rights and obligations under licence	13
	Omit "and to section 7 of the <i>Water Act 1912</i> (Rights of occupiers of riparian land)" from section 23 (4). Insert instead "and to Part 1 of Chapter 3 of the <i>Water Management Act 2000</i> (Basic landholder rights)".	14 15 16 17
[3]	Section 32 Local water extraction	18
	Omit "Water Act 1912 a right to take and use water" from section 32 (1).	19
	Insert instead "Water Management Act 2000 an access licence, water use approval or water supply work approval".	20 21

8.24	Soil Conservation Act 1938 No 10	1
[1]	Section 14 Carrying out of works by Minister or Commissioner	2
	Omit "a licence or permit issued pursuant to Part 2 of the <i>Water Act 1912</i> " from section 14 (3).	3 4
	Insert instead "a water management work approval under the <i>Water Management Act 2000</i> ".	5 6
[2]	Section 14 (3)	7
	Omit "Part 2 of the Water Act 1912 relating to licences or permits".	8
	Insert instead "the Water Management Act 2000 relating to water management work approvals".	9 10
8.25	Subordinate Legislation Act 1989 No 146	11
	Schedule 4 Excluded instruments	12
	Omit item 11. Insert instead:	13
	11 By-laws under Part 2 of Chapter 4 of the <i>Water Management Act 2000</i> .	14 15
8.26	Sydney Water Act 1994 No 88	16
[1]	Section 3 Definitions	17
	Omit "Water Administration Act 1986" from the definition of Water	18
	Administration Ministerial Corporation in section 3 (2). Insert instead "Water Management Act 2000".	19 20
[2]	Section 6 Role of certain Ministers	21
[4]		
	Omit "Water Administration Act 1986" from section 6 (1) (b). Insert instead "Water Management Act 2000".	22 23
	most mode muci management for 2000.	23

[3]	Section 10 Area of operations	1
	Omit "referred to in section 11 (2) of the <i>Water Administration Act 1986</i> " from section 10 (3) (c).	2 3
	Insert instead "within the meaning of the Water Management Act 2000".	4
[4]	Section 10 (3) (d)	5
	Omit the paragraph. Insert instead:	6
	(d) a water supply authority within the meaning of the Water Management Act 2000,	7 8
[5]	Section 12 Grant of operating licences	9
	Omit section 12 (3)–(6).	10
[6]	Section 30 Sydney Water Corporation Licence Regulator	11
	Omit "Water Administration Act 1986" from section 30 (4) (e). Insert instead "Water Management Act 2000".	12 13
[7]	Section 90 Activities outside area of operations	14
	Omit the section.	15
[8]	Section 91 Contracting out	16
	Omit the section.	17
8.27	Sydney Water Catchment Management Act 1998 No 171	18
[1]	Section 3 Definitions	19
	Omit the definition of water supply authority. Insert instead:	20
	water supply authority has the same meaning as it has in the Water Management Act 2000.	21 22
[2]	Section 25 Grant of operating licence	23
	Omit section 25 (2)–(5).	24

Schedule 8	Amendment of other Acts
OCHEGUE O	AITIETIUTIETI VI VIITET ACIS

8.28	Valuation of Land Act 1916 No 2	1
[1]	Section 58 Allowances for profitable expenditure—land	2
	Omit "Division 4 or 4A of Part 2 of the <i>Water Act 1912</i> " from section 58 (2) (c).	3
	Insert instead "the Water Management Act 2000".	5
[2]	Section 58 (6) and (7)	ϵ
	Omit "Division 4 or 4A of Part 2 of the Water Act 1912" wherever	7
	occurring.	8
	Insert instead "the Water Management Act 2000".	ç

Schedule 9 Savings, transitional and other provisions				1
			(Section 404)	2
Part	1 F	Prelin	minary	3
1	Sav	ings a	and transitional regulations	4
	(1)		regulations may contain provisions of a savings or transitional reconsequent on the enactment of the following Acts:	5 6
		this A	Act	7
	(2)		a provision may, if the regulations so provide, take effect from ate of assent to the Act concerned or a later day.	8 9
	(3)	is ear	ne extent to which such a provision takes effect from a date that clier than the date of its publication in the Gazette, the provision not operate so as:	10 11 12
		(a)	to affect, in a manner prejudicial to any person (other than the State or an authority of the State), the rights of that person existing before the date of that publication, or	13 14 15
		(b)	to impose liabilities on any person (other than the State or an authority of the State) in respect of anything done or omitted to be done before the date of that publication.	16 17 18
Part	2 F	Provi	isions consequent on enactment of this Act	19
Divis	ion '	1	General	20
2	Def	inition	s	21
		In thi	is Part:	22
		amen	nded legislation means any Act amended by Schedule 8, as so nded.	23 24
		appo	inted day means:	25
		(a)	in relation to a provision of the old legislation that has been repealed or amended by this Act, the day on which the repeal or amendment commences, or	26 27 28

		(b)	in relation to a new provision inserted into the amended legislation by this Act, the day on which the new provision commences.	1 2 3
		old le	gislation means:	4
		(a)	any Act repealed by Schedule 7, as in force immediately before its repeal, and	5
		(b)	any Act amended by Schedule 8, as in force immediately before its amendment.	7 8
3	Exis	sting w	vater management works	ç
	(1)	Nothi	ing in this Act requires a person to hold an approval:	10
		(a)	to complete the construction of a water management work whose construction was lawfully commenced before the appointed day, or	11 12 13
		(b)	to use a water management work whose construction was lawfully commenced before the appointed day, whether or not its construction was completed before the appointed day.	14 15 16
	(2)	enfor	lause (1) does not prevent a direction from being given and ced under Chapter 7 in relation to a water management work red to in that subclause.	17 18 19
	(3)	imme entitle	clause does not apply to a water management work that, ediately before the appointed day, was the subject of an ement referred to in clause 9, a licence referred to in clause 12 or mit referred to in clause 14.	20 21 22 23
	(4)		clause ceases to have effect on a day to be appointed by amation.	24 25
4	Dele	egatio	ns	26
		under	delegation that was in force immediately before the appointed day a provision of the old legislation for which there is a sponding provision in the amended legislation is taken to be a ation in force under the corresponding provision of the amended ation.	27 28 29 30

5	Construct Corporation	ion of references to Water Administration Ministerial on	1 2
	Subje	ct to this Schedule and the regulations, in any Act or instrument,	3
	a refe	erence to the Water Administration Ministerial Corporation	4
		ever described), in relation to a function that under the old	5
	legisla	ation was exercisable by the Ministerial Corporation but under	6
	this A	act is exercisable by the Minister, extends to the Minister.	7
6	Construct	ion of other references	8
	Subje	ct to this Schedule and the regulations, in any Act or instrument:	9
	(a)	a reference to a provision of the old legislation for which there	10
		is a corresponding provision in the amended legislation extends	11
		to the corresponding provision of the amended legislation, and	12
	(b)	a reference to any act, matter or thing referred to in a provision	13
		of the old legislation for which there is a corresponding	14
		provision in the amended legislation extends to the	15
		corresponding act, matter or thing referred to in the	16
		corresponding provision of the amended legislation.	17
7	General sa	aving	18
	Subje	ct to this Schedule and the regulations:	19
	(a)	anything begun before the appointed day under a provision of	20
		the old legislation for which there is a corresponding provision	21
		in the amended legislation may be continued and completed	22
		under the old legislation as if this Act had not been enacted,	23
		and	24
	(b)	subject to paragraph (a), anything done under a provision of the	25
		old legislation for which there is a corresponding provision in	26
		the amended legislation (including anything arising under	27
		paragraph (a)) is taken to have been done under the	28
		corresponding provision of the amended legislation.	29
Divis	ion 2	Water Act 1912	30
8	Definition		31
		s Division, the 1912 Act means the Water Act 1912, as in force	32
	imme	diately before the appointed day.	33

9	Ent	itlements	1
	(1)	An entitlement that, immediately before the appointed day, was in force under the 1912 Act:	2 3
		(a) to the extent to which it entitles any person or body to take a specified quantity of water, is taken to be an access licence held by that person or body for the quantity of water so specified (subject to such of the conditions of the entitlement as are applicable to an access licence), and	4 5 6 7 8
		(b) to the extent to which it entitles any person or body to use a specified water management work, is taken to be a water management work approval held by that person or body in respect of that work (subject to such of the conditions of the entitlement as are applicable to an approval of that kind), and	9 10 11 12 13
		 (c) is taken to be a water use approval entitling that person or body to use water on the land to which the entitlement relates (subject to such of the conditions of the entitlement as are applicable to an approval of that kind): (i) except as provided by subparagraph (ii), for any purpose, or (ii) if the conditions of the entitlement restrict the use of water to specified purposes, for the purposes so specified. 	14 15 16 17 18 19 20 21 22
	(2)	The reference in subclause (1) (a) to the conditions of an entitlement include a reference to any water allocation determined in respect of that entitlement under a volumetric water allocations scheme in force under Division 4B of Part 2 of the 1912 Act.	23 24 25 26
	(3)	In the case of a licence under section 13A or an authority under section 20CA of the 1912 Act, the holder of the corresponding water management work approval referred to in subclause (1) (b) may continue to exercise the rights conferred by section 13A (7) or 20CA (8) of that Act, as the case requires, while that approval (excluding any renewal of that approval) remains in force.	27 28 29 30 31 32
		Note. After that approval expires, the continued exercise of those rights will need to be supported by an easement, whether negotiated by agreement between the respective landholders or imposed by means of an order under section 88K of the <i>Conveyancing Act 1919</i> .	33 34 35 36

	(4) An access licence arising under subclause (1) (a), a water management work approval arising under subclause (1) (b) and a water use approval arising under subclause (1) (c) each have effect for 10 years from the appointed day.					
	(5)	In th	is clause, <i>entitlement</i> means:	5		
		(a)	a licence, permit, authority, irrigation corporation licence or group licence referred to in Part 2 of the 1912 Act, or	6		
		(b)	a right to take and use water referred to in section 38B of the 1912 Act, or	9		
		(c)	a licence referred to in Part 5 of the 1912 Act, or	10		
		(d)	an approval referred to in Part 8 of the 1912 Act.	11		
10	Rat	es and	d charges	12		
		-	rates and charges fixed under Part 7 of the 1912 Act are taken to been fixed under Part 3 of Chapter 5 of this Act.	13 14		
11	Floo	odplai	ins	15		
		appo	ect to the regulations, any land that, immediately before the pinted day, was designated as a floodplain under Part 8 of the 1912 is taken to be a floodplain for the purposes of this Act.	16 17 18		
12	Wat	er ma	anagement licences	19		
		day, autho	was in force under Part 9 of the 1912 Act is taken to be a water orisation under Part 1 of Chapter 6 of this Act, and is taken to be ect to the same conditions as it was subject under that Act.	20 21 22 23		
Divis	ion (3	Rivers and Foreshores Improvement Act 1948	24		
13	Def	inition	1	25		
			nis Division, the 1948 Act means the Rivers and Foreshores covement Act 1948, as in force immediately before the appointed	26 27 28		

Schedule 9	Savinge	transitional	and	othor	nrovicione
Juliedale 3	Savirius,	uansilionai	anu	Othici	PIUVISIUIS

14	Por	mits	1
1-4	(1)	15	2 3 4 5 6
	(2)	A controlled activity approval arising under subclause (1) has effect for the balance of the term for which it was granted.	7 8
Divis	ion 4	Water Administration Act 1986	9
15	Def	inition	10
		In this Division, <i>the 1986 Act</i> means the <i>Water Administration Act</i> 1986, as in force immediately before the appointed day.	11 12
16	Cor	ntinuation of Ministerial Corporation	13
		The Water Administration Ministerial Corporation constituted by Part 2 of Chapter 8 of this Act is a continuation of, and the same legal entity as, the Water Administration Ministerial Corporation constituted by the 1986 Act.	14 15 16 17
17	Cor	ntinued operation of section 19	18
		Section 19 of the 1986 Act continues to apply to acts, matters and things that arose before the appointed day.	19 20
Divis	ion (Water Supply Authorities Act 1987	21
18	Def	inition	22
		In this Division, <i>the 1987 Act</i> means the <i>Water Supply Authorities Act</i> 1987, as in force immediately before the appointed day.	23 24
19	Cor	ntinuation of water supply authorities	25
	(1)	Each water supply authority constituted by Part 2 of Chapter 6 of this Act is a continuation of, and the same legal entity as, the water supply authority of the same name constituted by the 1987 Act.	26 27 28

	(2)	Subject to the regulations, the area of operations of a water supply authority constituted by Part 2 of Chapter 6 of this Act is the same as the area of operations for the corresponding water supply authority constituted by the 1987 Act.	1 2 3 4
	(3)	Subject to this Act, the members of a water supply authority constituted by the 1987 Act continue to hold office as members of the corresponding water supply authority constituted by Part 2 of Chapter 6 of this Act for the residue of their respective terms of office under the 1987 Act.	5 6 7 8 9
20	Spe	cial areas	10
		A special area for a water supply authority constituted by the 1987 Act is taken to be a special area for the corresponding water supply authority constituted by Part 2 of Chapter 6 of this Act.	11 12 13
21	Cer	tificates of compliance	14
		A certificate of compliance granted by a water supply authority under the 1987 Act is taken to be a certificate of compliance issued by the corresponding water supply authority under Part 2 of Chapter 6 of this Act.	15 16 17 18
22	Ser	vice charges	19
	(1)	An order declaring a development area, charging year, drainage area, floodplain, river management area or special industry under section 28 of the 1987 Act is taken to be the corresponding order under Part 2 of Chapter 6 of this Act.	20 21 22 23
	(2)	Service charges levied by a water supply authority under the 1987 Act are taken to be service charges levied by the corresponding water supply authority under Part 2 of Chapter 6 of this Act.	24 25 26
	(3)	Determinations made by a water supply authority under section 34 of the 1987 Act are taken to be determinations made by the corresponding water supply authority under Part 2 of Chapter 6 of this Act.	27 28 29 30
	(4)	Assessments made by a water supply authority under section 35 or 36 of the 1987 Act are taken to be assessments made by the corresponding water supply authority under Part 2 of Chapter 6 of this Act.	31 32 33 34

	(5) A certificate issued by a water supply authority under section 41 of the 1987 Act is taken to be a certificate issued by the corresponding water supply authority under Part 2 of Chapter 6 of this Act.	1 2 3
23	Penalty notices	4
	A penalty notice issued by an authorised employee of a water supply	5
	authority under section 51 of the 1987 Act is taken to have been issued	6
	by an authorised employee or agent of the corresponding water supply	7
	authority under Part 2 of Chapter 6 of this Act, and may be enforced accordingly.	8 9
24	Regulations	10
	Any regulations in force under the 1987 Act immediately before the	11
	appointed day are taken to be regulations under this Act, and may be	12
	amended and repealed accordingly.	13
Divis	ion 6 Fish River Water Supply Administration Act 1945	14
25	Definition	15
	In this Division, the 1945 Act means the Fish River Water Supply	16
	Administration Act 1945, as in force immediately before the appointed day.	17 18
26	Contributions by councils	19
	Any amount payable by a council under Part 4 of the 1945 Act is taken	20
	to be payable under Part 1 of Chapter 5 of this Act.	21
27	Regulations	22
	Any regulations in force under the 1945 Act immediately before the	23
	appointed day are taken to be regulations under this Act, and may be	24
	amended and repealed accordingly.	25
Divis	ion 7 Hunter Valley Flood Mitigation Act 1956	26
28	Definition	27
	In this Division, the 1956 Act means the Hunter Valley Flood	28
	Mitigation Act 1956, as in force immediately before the appointed day.	29

29	Permissions	1
	Any permission in force under section 12 of the 1956 Act immediately	2
	before the appointed day, and any approval in force under section 16	3
	of the 1956 Act immediately before the appointed day, are taken to be	4
	consents in force under Part 2 of Chapter 5 of this Act.	5
30	Control and management of works	6
	Any work that, immediately before the appointed day, was under the	7
	control and management of the Ministerial Corporation under the 1956	8
	Act is taken to be under the control and management of the Ministerial	9
	Corporation under Part 2 of Chapter 5 of this Act.	10
31	Finance	11
	Any amount for which the Hunter Catchment Management Trust or a	12
	local council was liable under the 1956 Act immediately before the	13
	appointed day is taken to be an amount for which the Trust or council	14
	is liable under Part 2 of Chapter 5 of this Act.	15
32	Regulations	16
	Any regulations in force under the 1956 Act immediately before the	17
	appointed day are taken to be regulations under this Act, and may be	18
	amended and repealed accordingly.	19
Divis	sion 8 Irrigation Corporations Act 1994	20
33	Definition	21
	In this Division, the 1994 Act means the Irrigation Corporations Act	22
	1994, as in force immediately before the appointed day.	23
34	Operating licences	24
	Any operating licence granted to an irrigation corporation under the	25
	1994 Act is taken to be an operating licence under Part 1 of Chapter	26
	4 of this Act.	27
35	Continuation of indemnity	28
	Section 75 of the 1994 Act continues to apply to matters arising under	29
	that Act hefore the appointed day as if this Act had not been enacted	30

Schedule 9	Savings,	transitional	and	other	provisions

36	Reg	ulations	1
		Any regulations in force under the 1994 Act immediately before the appointed day are taken to be regulations under this Act, and may be amended and repealed accordingly.	2 3 4
Divis	ion 9	Private Irrigation Districts Act 1973	5
37	Def	inition	6
		In this Division, <i>the 1973 Act</i> means the <i>Private Irrigation Districts Act 1973</i> , as in force immediately before the appointed day.	7 8
38	Priv	rate districts	9
	(1)	A provisional private domestic and stock water supply district constituted under the 1973 Act is taken to be a private domestic and stock water supply district constituted under Part 2 of Chapter 4 of this Act.	10 11 12 13
	(2)	A private domestic and stock water supply district constituted under the 1973 Act is taken to be a private domestic and stock water supply district constituted under Part 2 of Chapter 4 of this Act.	14 15 16
	(3)	A provisional private domestic and stock water supply and irrigation district constituted under the 1973 Act is taken to be a private domestic and stock water supply and irrigation district constituted under Part 2 of Chapter 4 of this Act.	17 18 19 20
	(4)	A private domestic and stock water supply and irrigation district constituted under the 1973 Act is taken to be a private domestic and stock water supply and irrigation district constituted under Part 2 of Chapter 4 of this Act.	21 22 23 24
39	Boa	ards of management	25
	(1)	The Board of management of a provisional private district or private district under the 1973 Act is taken to be a private irrigation board under Part 2 of Chapter 4 of this Act.	26 27 28
	(2)	Subject to this Act, the members of a Board of management of a provisional private district or private district constituted under the 1973 Act continue to hold office as members of the corresponding private irrigation board under this Act for the residue of their respective terms of office under the 1973 Act.	29 30 31 32 33

40	Control and management of works	1
	Any work that, immediately before the appointed day, was under the control and management of the Board of management of a provisional private district or private district under the 1973 Act is taken to be under the control and management of the corresponding private irrigation board under Part 2 of Chapter 4 of this Act.	2 3 4 5 6
41	Rates and charges	7
	(1) Any rates and charges fixed by the Board of management of a provisional private district or private district under the 1973 Act before the appointed day are taken to have been fixed by the corresponding private irrigation board under Part 2 of Chapter 4 of this Act.	8 9 10 11
	(2) A certificate issued by the Board of management of a provisional private district or private district under the 1973 Act before the appointed day is taken to be a certificate issued by the corresponding private irrigation board under Part 2 of Chapter 4 of this Act.	12 13 14 15
42	Water allocations	16
	Any determination made by the Board of management of a provisional private district or private district under the 1973 Act before the appointed day is taken to be a determination made by the corresponding private irrigation board under Part 2 of Chapter 4 of this Act.	17 18 19 20 21
43	By-laws	22
	Any by-laws made the Board of management of a provisional private district or private district under the 1973 Act before the appointed day are taken to be by-laws made by the corresponding private irrigation board under Part 2 of Chapter 4 of this Act, and may be amended and repealed accordingly.	23 24 25 26 27
44	Regulations	28
	Any regulations in force under the 1973 Act immediately before the appointed day are taken to be regulations under this Act, and may be amended and repealed accordingly.	29 30 31

Schedule 9	Savings,	transitional	and	other	provisions

Divis	on 10 Drainage Act 1939	1
45	Definition	2
	In this Division, <i>the 1939 Act</i> means the <i>Drainage Act 1939</i> , as in force immediately before the appointed day.	3 4
46	Valuations	5
	Any valuation in force in respect of a board under Division 1 of Part 5 of the 1939 Act immediately before the appointed day is taken to be a valuation in respect of a private drainage board under Division 4 of Part 3 of Chapter 4 of this Act.	6 7 8 9
47	Rates	10
	Any rate levied under Division 2 of Part 5 of the 1939 Act before the appointed day is taken to be a rate levied under Division 3 of Part 3 of Chapter 4 of this Act.	11 12 13
48	Regulations	14
	Any regulations in force under the 1939 Act immediately before the appointed day are taken to be regulations under this Act, and may be amended and repealed accordingly.	15 16 17 18
Divis	on 11 Glennies Creek Dam Act 1979	19
49	Definition	20
	In this Division, <i>the 1979 Act</i> means the <i>Glennies Creek Dam Act</i> 1979, as in force immediately before the appointed day.	21 22
50	Supply of water to Macquarie Generation	23
	(1) The obligation to supply Macquarie Generation with water that the Ministerial Corporation had immediately before the appointed day under section 6 of the 1979 Act is taken to continue as if that Act had not been repealed.	24 25 26 27
	Note. Under section 6 of the 1979 Act, the obligation is expressed to be to supply Pacific Power. By virtue of an order under clause 3 of Schedule 5 to the <i>Energy Services Corporations Act 1995</i> , the right to that supply has been transferred from Pacific Power to Macquarie Generation.	28 29 30 31

	ned		

(2)	This	clause ceases to have effect:	1
	(a)	on the expiry of 2 years after the appointed day, or	2
	(b)	on the day on which a proclamation adding Macquarie Generation's name to Schedule 2 takes effect,	3 4
	whicl	never occurs first.	5

Dictionary

Dictionary (Section 4)	1 2
access licence means an access licence referred to in section 47.	3
activity approval means a controlled activity approval or an aquifer interference approval.	4 5
<i>approval</i> means a water use approval, a water management work approval or an activity approval.	6 7
aquifer means a geological structure or formation, or an artificial landfill, that is permeated with water or is capable of being permeated with water.	8 9 10
aquifer interference approval means an approval referred to in section 82.	11 12
authorised officer, in relation to a provision of this Act, means a person authorised by the Minister to exercise the functions conferred on an authorised officer by that provision, whether generally or in a particular case.	13 14 15 16
available water, in relation to a water management area or water source, means the water that is available in that area or water source in accordance with an available water determination that is in force in respect of that area or water source.	17 18 19 20
available water determination means a determination referred to in section 50.	21 22
basic landholder rights means domestic and stock rights or harvestable rights.	23 24
bulk access regime means a bulk access regime established by a management plan, as referred to in section 17 (1) (e), or by a Minister's plan, and includes a bulk access regime as varied by the Minister under section 38.	25 26 27 28
coastal waters of the State has the same meaning as it has in Part 10 of the Interpretation Act 1987.	29 30
construct a work includes install, alter or extend the work.	31

controlled activity means:	1
(a) the erection of a building, or	2
(b) the carrying out of a work, or	3
(c) the removal of extractive material from land, or	4
(d) the carrying out of landfill operations.	5
controlled activity approval means an approval referresection 82.	ed to in
<i>Crown land</i> has the same meaning as it has in the <i>Crown I</i> 1989.	Lands Act 8
deemed regional environmental plan means a management the kind referred to in section 20 (3) (a).	nt plan of
Department means the Department of Land and Water Cons	servation. 12
development and carry out development have the same me they have in the Environmental Planning and Assessment A	
development consent has the same meaning as it has Environmental Planning and Assessment Act 1979.	s in the 15
Director-General means the Director-General of the Depart	tment. 17
domestic and stock rights means the rights conferred on a laby section 44.	undholder 18
drainage management principles means the principles refesection 22.	rred to in 20
drainage work means a work (such as a pump, pipe or char is constructed or used for the purpose of draining water fr including a reticulated system of such works, and inc associated pipes, sluices, valves and equipment, but does no	rom land, 23 ludes all 24
(a) any sewage work, or	26
(b) any work declared by the regulations not to be a draina	age work. 27
drainage work approval means an approval referred to in se	ection 81. 28
environment includes all aspects of the surroundings of huma whether affecting them as individuals or in their social grou	
environmental water principles means environmental water pestablished by a management plan or Minister's plan.	principles 31
estuary means the waters between the mouth of a river and the waters of the State.	ne coastal 33

Dictionary

exercise	a function includes perform a duty.	1
	we material means earth, sand, gravel, rock, mud, clay or any ch substance.	2 3
<i>flood wo</i> embankı	ork means a work (such as a barrage, causeway, cutting or ment):	4 5
(i	hat is situated: i) in or in the vicinity of a river, estuary or lake, or ii) within a floodplain, and	6 7 8
p a (i	hat is of such a size or configuration that, regardless of the surpose for which it is constructed or used, it is likely to have significant effect on: i) the flow of water to or from a river, estuary or lake, or the distribution or flow of floodwater in times of flood,	9 10 11 12 13
	udes all associated pipes, valves and equipment, but does not any work declared by the regulations not to be a flood work.	14 15
flood wo	ork approval means an approval referred to in section 81.	16
floodpla	in means land declared by the regulations to be a floodplain.	17
manual 1	ain Development Manual means the floodplain development referred to in section 733 of the Local Government Act 1993, recently notified under section 733 (5) (a) of that Act.	18 19 20
floodpla section 2	tin management principles mean the principles referred to in 26.	21 22
function	<i>i</i> includes a power, authority and duty.	23
	able rights mean the rights conferred on a landholder by a ble rights order.	24 25
	able rights area means an area of land that is constituted as a ble rights area by a harvestable rights order.	26 27
harvesta	able rights order means an order in force under section 46.	28
impleme	entation program means a program referred to in section 43.	29
	on corporation means an irrigation corporation referred to in f Chapter 4.	30 31

lake in	ncludes:	1
(a)	a wetland, a lagoon, a saltmarsh and any collection of still water, whether perennial or intermittent and whether natural or artificial, and	2 3 4
(b)	any water declared by the regulations to be a lake,	5
but do	bes not include any water declared by the regulations not to be a	6 7
	ncludes any water source, and also includes the land on or in any water source is situated.	8 9
landh	older, in relation to land, means:	10
(a)	the owner of the land or (if the owner is not in occupation of the land) the lawful occupier of the land, or	11 12
(b)	the local council, in the case of land that comprises: (i) a public road vested in, or under the control or management of, the local council, or (ii) the site of a work that is, or is proposed to be, constructed for the local council.	13 14 15 16
local	council means the council of a local government area.	18
	<i>newspaper</i> , in relation to any area or locality, means a paper circulating within that area or locality.	19 20
	transfer rules means rules referred to in section 18 (1) (d), her established by a management plan or Minister's plan.	21 22
local	water utility means:	23
(a)	a water supply authority established by this Act, or	24
(b)	a council or county council exercising water supply functions under Division 2 of Part 3 of Chapter 6 of the <i>Local Government Act 1993</i> .	25 26 27
major	utility means a person or body referred to in Schedule 2.	28
mana	gement committee means a committee referred to in section 9.	29
relation within	gement plan means a plan referred to in section 12 and, in on to any land that is not within a water management area, or is a water management area for which there is no relevant gement plan, includes a Minister's plan.	30 31 32 33
	<i>terial Corporation</i> means the Water Administration Ministerial oration constituted by section 382.	34 35

Mini	ster's plan means a plan referred to in section 42.	1
	er of land means the only person who, or each person who jointly verally, at law or in equity:	2 3
(a)	is entitled to the land for an estate of freehold in possession, or	4
(b)	is a person to whom the Crown has contracted to sell the land under the <i>Crown Lands Act 1989</i> , or	5 6
(c)	is entitled to receive, or receives, or if the land were let to a tenant would be entitled to receive, the rents and profits of the land, whether as beneficial owner, trustee, mortgagee in possession or otherwise,	7 8 9 10
,	in relation to land of the Crown other than land for which there owner under paragraph (b) or (c), means the Crown.	11 12
princ	<i>tiples of ecologically sustainable development</i> means the iples of ecologically sustainable development described in section of the <i>Protection of the Environment Administration Act 1991</i> .	13 14 15
	te drainage board means a private drainage board referred to in 3 of Chapter 4.	16 17
	<i>te irrigation board</i> means a private irrigation board referred to in 2 of Chapter 4.	18 19
	te water trust means a private water trust referred to in Part 4 of oter 4.	20 21
publi	ic authority means:	22
(a)	a Minister of the Crown, or	23
(b)	a Government Department or Administrative Office, or	24
(c)	a statutory body representing the Crown, or	25
(d)	a statutory State owned corporation (or any of its subsidiaries) within the meaning of the <i>State Owned Corporations Act 1989</i> , or	26 27 28
(e)	a council or county council within the meaning of the <i>Local Government Act 1993</i> ,	29 30
	oes not include any person or body declared by the regulations of be a public authority.	31 32
	<i>lated river</i> means a river that is declared by the Minister, by order shed in the Gazette, to be a regulated river.	33 34

river	includes:	1	
(a)	any watercourse, whether perennial or intermittent, flowing in a natural channel, or in a natural channel artificially improved, and	2 3 4	
(b)	any tributary, branch or other watercourse into or from which a watercourse referred to in paragraph (a) flows, and	5 6	
(c)	anything declared by the regulations to be a river,	7	
but do	bes not include anything declared by the regulations not to be a	8	
constr includassoci	ge work means a work (such as a pump, pipe or channel) that is ructed or used for the purpose of removing sewage from land, ling a reticulated system of such works, and includes all lated pipes, sluices, valves and equipment, but does not include work declared by the regulations not to be a sewage work.	10 11 12 13	
State?	s water rights means the rights referred to in section 393 (1).	15	
the tr	fer principles means principles established under section 60 for ansfer of access licences and the water allocations under an s licence.	16 17 18	
unreg	rulated river means a river that is not a regulated river.	19	
use a	water management work means:	20	
(a)	 in relation to a water supply work: (i) to operate the work for the purpose of taking water from a water source or supplying land with water taken (whether directly or indirectly) from a water source, or (ii) to allow the work to operate for that purpose, or 	21 22 23 24 25	
(b)	 in relation to a drainage work: (i) to operate the work for the purpose of draining water from land, or (ii) to allow the work to operate for that purpose, or 	26 27 28 29	
(c)	 in relation to a flood work: (i) to operate the work for the purpose of influencing the flow of floodwater in the vicinity of the land on which the work is situated, or (ii) to allow the work to operate for that purpose. 	30 31 32 33 34	
Water Advisory Council means the Water Advisory Council established by section 380.			

Dictionary

1 2
3 4
5
6
7 8
9 10
11
12 13
14 15
16 17
18 19
20 21
22 23
24 25 26
27 28
29
30 31 32
33 34 35 36

(c)	a work (such as a water pipe or irrigation channel) that is constructed or used for the purpose of conveying water to the point at which it is to be used,	1 2 3
	ding a reticulated system of such works, and includes all iated pipes, sluices, valves and equipment, but does not include:	4 5
(d)	any work that receives water from a water supply work under the control or management of the Sydney Water Board, the Hunter Water Board or a local water utility, or	6 7 8
(e)	any work declared by the regulations as not being a water supply work.	9 10
water section	supply work approval means an approval referred to in on 81.	11 12
water	ruse approval means an approval referred to in section 80.	13
water	front land means an area of land comprising:	14
(a)	a river, estuary or lake, or	15
(b)	if the regulations so provide, the coastal waters of the State, or	16
(c)	land within the prescribed distance inland of: (i) the high bank (or, if there is no high bank, the mean water level) of any river or lake, or (ii) the mean high water mark along the waterfront of any estuary or of the coastal waters of the State,	17 18 19 20 21
presc	e the prescribed distance is 40 metres or (if the regulations ribe a lesser distance, either generally or in relation to a particular on or class of locations) that lesser distance	22 23